# Chapter 5

# The phase field method for geometric moving interfaces and their numerical approximations

# Qiang Du<sup>a,\*</sup> and Xiaobing Feng<sup>b</sup>

<sup>a</sup>Department of Applied Physics and Applied Mathematics and the Data Science Institute, Columbia University, New York, NY, United States

<sup>b</sup>Department of Mathematics, The University of Tennessee, Knoxville, TN, United States

\*Corresponding author: e-mail: qd2125@columbia.edu

# **Chapter Outline**

Chapter Outline								
1	Introduction		426		3.1	Classical schemes	447	
2	Mat	thematical foundation of the			3.2	Convex splitting and		
	phase field method		431			stabilized schemes	449	
	2.1	Geometric surface			3.3	Schemes using Lagrangian		
		evolution	431			multipliers	451	
	2.2	Examples of geometric			3.4	Further considerations	453	
		surface evolution	431	4	Spa	tial discretization methods		
	2.3	Mathematical formulations		1	for	phase field models	454	
		and methodologies	435		4.1	Spatial finite difference		
	2.4	Level set and phase field				discretization	454	
		formulations of the MCF	436		4.2	Spatial Galerkin		
	2.5	Phase field formulations of				discretizations	455	
		other moving interface			4.3	Spatial mixed		
		problems	439			discretization	460	
	2.6	Relationships between phase			4.4	Implementations and		
		field and other				advantages of high order		
		formulations	442			methods	461	
	2.7	Phase function		5	Con	vergence theories of fully		
		representations of geometric			disc	rete numerical methods	462	
		quantities	444		5.1	Construction of fully discrete		
	2.8	Convergence of the phase				numerical schemes	462	
		field formulation	445		5.2	Types of convergence		
3	Time-stepping schemes					and a priori error		
	for	phase field models	446			estimates	464	

5.3 Coarse error estimates for a		7.1 Materials science	
fixed value $\varepsilon > 0$	466	applications	475
5.4 Fine error estimates and		7.2 Fluid and solid mechanics	
convergence of numerical		applications	477
interfaces as $\varepsilon$ , $h$ , $\tau \to 0$	468	7.3 Image and data processing	
6 A posteriori error estimates		applications	478
and adaptive methods	471	7.4 Biology applications	479
6.1 Spatial and temporal		7.5 Other variants of phase field	
adaptivity	471	models	481
6.2 Coarse and fine a posteriori		8 Conclusion	
error estimates for phase		Acknowledgements	484
field models	472	References	484
7 Applications and extensions	475	Further readings	

### **Abstract**

This chapter surveys recent numerical advances in the phase field method for geometric surface evolution and related geometric nonlinear partial differential equations (PDEs). Instead of describing technical details of various numerical methods and their analyses, the chapter presents a holistic overview about the main ideas of phase field modelling, its mathematical foundation, and relationships between the phase field formalism and other mathematical formalisms for geometric moving interface problems, as well as the current state of the art of numerical approximations of various phase field models with an emphasis on discussing the main ideas of numerical analysis techniques. The chapter also reviews recent development on adaptive grid methods and various applications of the phase field modelling and their numerical methods in materials science, fluid mechanics, biology and image science.

**Keywords:** Phase field method, Geometric law, Curvature-driven flow, Geometric nonlinear PDEs, Finite difference methods, Finite element methods, Spectral methods, Discontinuous Galerkin methods, Adaptivity, Coarse and fine error estimates, Convergence of numerical interfaces, Nonlocal and stochastic phase field models, Microstructure evolution, Biology and image science applications

**AMS Classification Codes:** 65M06, 65M60, 65M70, 65M20, 65M12, 65M15, 65Z05, 35B25, 35K20, 35K35, 53C44, 74A50, 74N20, 76T99, 80A22, 82C26, 92C17, 92C37, 92C55

## 1 Introduction

The idea of the phase field method could be traced back to Lord Rayleigh, Gibbs and Van der Waals and was used to describe material interfaces during phase transitions. It represents material interfaces as thin layers of finite thickness over which material properties vary smoothly. Such a thin layer of the width  $O(\varepsilon)$  is often referred as a diffuse interface, and by the design the exact interface is guaranteed to be within the thin layer. In other words, this is amount to smear the exact sharp interface into a thin diffuse interface layer. For this very reason, the phase field method is also known as the diffuse interface method in the literature.

The phase field method was first introduced to model solid-liquid phase transition in which surface tension and nonequilibrium thermodynamics behaviour become important at the interface. Computationally, the phase field method has features in common with the level set method (Chen et al., 1991; Evans and Spruck, 1991; Osher and Fedkiw, 2003; Osher and Sethian, 1988; Sethian, 1999) in that explicit tracking of the interface can be avoided in both mathematical formulation and in numerical computations, which has advantages in mesh generation and in capturing topological changes of the interface. The phase field method uses an auxiliary phase variable/function  $u^{\varepsilon}$ , also known as the order parameter, to indicate phases. The phase function assumes distinct values in the bulk phases away from the diffuse interface (or the interfacial region); the after-sought interface can be identified with an intermediate level set (e.g. the zero-level set) of the phase function  $u^{\varepsilon}$ . It should be noted that although the level set method and the phase field method are intimately related, they differ fundamentally because the former tracks the exact sharp interface without introducing any diffuse layer.

It is clear that the idea of the phase field method does not restrict to the material phase transition problems, it is applicable to any moving interface (or free boundary) problems. Indeed, in the past forty years the phase field (or the diffuse interface) method has been developed into a major and general methodology for moving interface problems arising from astrophysics, biology, differential geometry, image processing, multiphase fluid mechanics, chemical and petroleum engineering, materials phase transition and solidification. One common theme of these application problems is that interfacial energy plays an important role in each of these moving interface evolutions. Mathematically, interfacial energies such as the surface tension are characterized by the curvature(s) of the interface such as the mean and Gauss curvatures. It turns out that the mean and Gauss curvatures can be conveniently expressed in term of the phase function, which makes the phase field method very effective for modelling the interfacial energetics, particularly the surface tension effect.

Moving interface problems under the influence of the surface tension or other interfacial energy belong to a larger class of so-called geometric moving interface problems in which the motion of the interfaces is driven by some curvature-dependent geometric law. The phase field formulations of such problems often give rise interesting and difficult geometric partial differential equations (PDEs), which is a main subject of this chapter. Among many geometric moving interface problems, the best known one perhaps is the mean curvature flow (MCF) whose governing geometric law is:  $V_n(t) = H_{\Gamma(t)}$ , where  $V_n$  and  $H_{\Gamma(t)}$ , respectively, stand for the normal velocity and the mean curvature of the moving interface  $\Gamma(t)$  at time t. As the MCF is purely a geometric problem, it can be described by various formulations including parametric, level set and phase field formulations. It is well known that the best known phase field formulation for the MCF is the Allen-Cahn equation:  $u_t^{\varepsilon} - \Delta u^{\varepsilon} + \varepsilon^{-2}((u^{\varepsilon})^3 - u^{\varepsilon}) = 0$ , which is the simplest geometric PDE. Here the small positive parameter  $\varepsilon$  measures the width of the diffuse interface layer. One of the most important theoretical results for the phase field method is to prove that the zero-level set  $\Gamma^{\varepsilon}(t) := \{x \in \mathbf{R}^d; u^{\varepsilon}(x, t) = 0\}$  evolves according to the geometric law of the MCF and  $\Gamma^{\varepsilon}(t)$  converges (as a set) to the exact MCF interface  $\Gamma(t)$  as  $\varepsilon \to 0^+$  (Evans et al., 1992).

Looking at the PDE, it may not be clear why the Allen-Cahn equation is related to the MCF as its phase field model. This was indeed an open question for more than 10 years. It took a considerable amount of effort by many researchers before the above convergence result was proved (Bronsard and Kohn, 1991; Chen, 1992; de Mottoni and Schatzman, 1995; Evans et al., 1992; Kohn and Sternberg, 1989; Rubinstein et al., 1989). Although rigorous proofs of the convergence of the phase field models/formulations of moving interface problems is generally difficult, there is a formal procedure which has been widely used for constructing/deriving phase field model/formulation for a given moving interface problem. This procedure is based on an energetic approach in which one first construct/postulate a Helmholtz free energy functional associated with the underlying moving interface problem, a desired phase field model/formulation then is obtained as the gradient flow for the energy functional in an appropriately chosen topology. The Helmholtz free energy functional often can be written as the sum of two parts: the first part is called the bulk energy and the second part is the interfacial (or mixing) energy. As its name indicates, the interfacial/mixing energy measures the energy stored in the interfacial layer and can be defined and computed with the help of the phase function. As expected, it would depend on the width of the interfacial layer. On the other hand, the bulk energy is usually problem-dependent and may consist of several other types of energies such as kinetic and potential/gradient energy. It should be emphasized that this energetic approach makes the phase field modelling become systematic and methodical instead of ad hoc. It shifts the main task of modelling to the construction of the Helmholtz free energy functional and the driving forces that provide the mechanism of energy dissipation.

After the phase field model is obtained, several important questions naturally arise and must be addressed. Among them we mention the following three: (1) Is the model mathematically well-posed as a geometric PDE problem? (2) Does the phase field model converge to the original sharp interface model as the width of the interfacial layer goes to zero? If it does, how fast or what is the convergence rate? (3) How to efficiently solve the phase field model (i.e., the geometric PDE problem) numerically? It turns out that these questions are not all easy to answer. For question (2) the main difficulties are to derive required uniform in  $\varepsilon$  estimates for the phase function and to characterize the solution of the limiting model. Such a rigorous proof of convergence is still missing for many moving interface problems from materials science and multiphase fluid mechanics. For question (3) several important issues must be considered. First, in order to preserve the gradient flow structure at the discrete level, it is desirable to ensure the numerical method to be

energy stable besides to be accurate, which is a nontrivial task, especially if one aims to have high order (time) schemes (see Yang (2016), Yang et al. (2017), and Shen et al. (2018) for some recent advances in this direction). Second, the phase field method is used to formulate/approximate a moving interface problem (i.e., the underlying sharp interface problem) and numerical methods are constructed to solve the phase field model. If the goal is to solve the sharp interface problem, a critical question is that whether the numerical solution of the phase field model converges to the solution of the underlying sharp interface model as the width of the diffuse interface tends to zero (i.e.,  $\varepsilon \to 0$ ). One may further ask in what sense the above convergence should be interpreted. It turns out that these issues are quite difficult to address and remain open for many phase field models and their numerical approximations. The successful proofs all require to carry out some delicate and nonstandard error analyses (Feng and Li, 2015; Feng and Prohl, 2003a, 2004a,c, 2005; Feng et al., 2016). Finally, in order to resolve the thin interfacial layer, one must use spatial (and temporal) mesh sizes that are much smaller than the layer width  $\varepsilon$  for computer simulations (note that  $\varepsilon \to 0$ ). For such extremely small mesh sizes, the resulting large linear and nonlinear algebraic problems require huge computational effort to solve. This is especially true in the three-dimensional case with uniform spatial meshes. To overcome the difficulty, one may rely on highly scalable algorithms (Takashi et al., 2011; Zhang et al., 2016a) or use adaptive methods that only use fine meshes in the interfacial layer and much coarser meshes away from the layer. The use of adaptive mesh is a natural choice considering the fact that the phase function has a nice structured profile that assumes distinct (close-to-constant) values in bulk phases away from the diffuse interface and has a large gradient in the diffuse interface.

Although the phase field method was introduced much earlier as a modelling tool for material phase transition (in particular, solidification), it was proposed as a computational technique in the early 1980s by J. S. Langer, G. Fix, G. Caginalp and others. Fix (1983) was perhaps the first to use the phase field method to numerically solve moving interface (or free boundary) problems. Since then the phase field method has garnered a lot of interest and become more and more popular. It has been developed into a general numerical methodology for solving various moving interface (i.e., free boundary) problems from differential geometry, materials science, fluid mechanics and biology, just to name a few. Practically, the methodology is well understood, however, theoretically, many difficult questions remain to be addressed both at the PDE level and at the discrete level. Moreover, computer simulations of the phase field method, especially in high dimensions, remains to be challenging because of shear amount of computations involved. Indeed, phase field simulations on extreme scales have been used as a test of computing power on world's leading high performance computers as exemplified by the appearances on the Gordon Bell prize competition Takashi et al. (2011) (2011 Gordon Bell prize winner) and Zhang et al. (2016a) (2016 Gordon Bell

prize finalist, which reported the largest 3D phase field simulations to date that reached 40 Petaflops on the world's fastest supercomputer at the time, employing over 300 billion spatial grid points (6700<sup>3</sup>) and utilizing over 10 million cores). With the increasing popularity of phase field modelling in more and more application domains, it is imperative that efficient numerical algorithms and solvers such as adaptive grid methods and preconditioning techniques must be developed and utilized.

The phase field method provides a diffusive interface description to many physical processes and geometric models. Its capability goes beyond codimensional one surfaces. For example, using vector fields or tensor fields, they can be used to describe the characteristics of geometric information with higher codimension such as point defects in two dimensions and curves in three dimensions. Mathematical models of this kind include the celebrated Ginzburg-Landau models of superconductivity, Landau-deGennes model for liquid crystal flow, generalized Ohta-Kawasaki model of diblock and triblock copolymers, among others (Davis and Gartland, 1998; Du et al., 1992; Kim and Lowengrub, 2005; Ohta and Kawasaki, 1986; Priestly, 2012; Tinkham, 2004). While many numerical methods designed for scalar equations can be extended to systems, there are new challenges associated with invariance and symmetries and choices of gauge (Du, 2005) in addition to added computational complexity due to additional quantities of interests.

This chapter consists of additional seven sections. Section 2 devotes to the mathematical foundation of the phase field method. This includes its formulations for specific geometric moving interface problems, its intimate connection to the level set method, convergence of the phase field method as the defuse interface width tends to zero, fine properties of solutions of phase field models and phase function representations of geometric quantities. Section 3 focuses on time-stepping schemes for the phase field models, in addition to ensure their accuracy, the emphasis is on presenting various energy stable schemes of different orders which include the classical energy splitting schemes and stabilized schemes as well as the newly developed IEQ (invariant energy quadratization) and SAV (scalar auxiliary variable) schemes. Section 4 devotes to spatial discretization methods, all the classical methodologies are covered. Among them are finite difference methods, finite element methods, spectral methods, discontinuous Galerkin methods and isogeometric methods. Section 5 focuses on two convergence theories of fully discrete numerical methods for phase field models. The first theory analyzes rates of convergence (or a priori error estimates) for a fixed diffuse interface width  $\varepsilon$ , while the second theory considers the convergence to the solution of the underlying sharp interface problem when both the mesh parameters and the physical parameter  $\varepsilon$  all tend zero. Section 6 addresses a posteriori error estimates which are used to design adaptive methods. The emphasis will be on the residual-based a posteriori estimators for finite element discretizations. All three types of popular adaptive methodologies will be discussed, they include h-, hp- and r-adaptive

methods. Section 7 lists a host of applications of the phase field method ranging from materials science, fluid mechanics to biology. It also discusses some extensions of the classical phase field method such as nonlocal and stochastic phase field models. The chapter is ended with a brief summary and a few concluding remarks given in Section 8.

#### 2 Mathematical foundation of the phase field method

## Geometric surface evolution

Mathematically, a hypersurface or an interface in the Euclidean space  $\mathbf{R}^{d+1}$ refers to codimension one subset  $\Gamma$  of  $\mathbf{R}^{d+1}$ . A moving interface (or a free boundary) refers to one-parameter family of interfaces  $\{\Gamma_t\}_{t\geq 0}$  which evolves in the space  $\mathbf{R}^{d+1}$  and are governed by some explicit or implicit mechanism/ law, often called a geometric law. The parameter t represents the time. Such a geometric law determines how the interface moves. Most commonly seen geometric laws specify the velocity V or normal velocity  $V_n$  of the points (thinking them as particles) on the interface at each given time t. It may depend on intrinsic features (such as curvatures) of the interface and/or on external factors (such as flow velocity) of the environment where the interface exists.

Specifically, geometric surface evolution concerns with the following question: given an initial hypersurface  $\Gamma_0 \in \mathbf{R}^{d+1}$ , find  $\Gamma_t \in \mathbf{R}^{d+1}$  such that

$$V_n(t) = F_{\text{int}}(\lambda_1, \dots, \lambda_d) + F_{\text{ext}} \quad \text{on } \Gamma_t, \tag{1}$$

where  $V_n = V \cdot n$  denotes the (outward) normal velocity of  $\Gamma_t$  and  $n = n_{\Gamma_t}$ stands for the outward normal to  $\Gamma_t$ .  $\{\lambda_i(t)\}_{i=1}^d$  are the principle curvatures of  $\Gamma_t$  and  $F_{\text{int}}$  is a given function of  $\{\lambda_i\}_{i\geq 1}$ .  $F_{\text{ext}}$  denotes an external (source) function. The geometric law (1) says that the normal velocity  $V_n$  of the interface is determined/driven by the sum of an internal (Fint) and an external (Fext) "forces". We note that problem (1) is stated purely as a geometric problem, it may be embedded in a more complicated moving interface problem arisen from a particular application.

#### **Examples of geometric surface evolution** 2.2

The following is a list of some best known geometric moving interface problems in the literature. We rephrase them in the framework of problem (1) by explicitly describing the functions  $F_{int}$  and  $F_{ext}$  for each example.

Example 1 (Mean curvature flow).

The mean curvature flow (MCF) is defined by setting

$$F_{\text{int}} = H := \sum_{j=1}^{d} \lambda_j, \quad F_{\text{ext}} \equiv 0$$

in (1). Where H = H(t) denotes the mean curvature of  $\Gamma_t$ . So the MCF seeks a family of hypersurfaces whose normal velocity at every point on the hypersurface is equal to the mean curvature of the hypersurface at that point for all time t > 0. The MCF is a curve-shortening (in 2D) and an area minimization flow (in higher dimensions) in the sense that it can be interpreted as a gradient flow for the curve length functional (in 2D) and the surface area functional (in higher dimensions) (Ecker, 2004; Giga, 2006; Zhu, 2002). The MCF finds applications in many fields such as materials science, image processing, and multiphase fluids.

## **Example 2** (A generalized mean curvature flow).

The generalized mean curvature flow (gMCF) to be considered in this chapter refers to the case with

$$F_{\text{int}} = H$$
,  $F_{\text{ext}} \equiv \mathbf{v} \cdot n_{\Gamma_t} + g$ 

in (1). Clearly, the difference between and the MCF and the gMCF is that the latter has a nonzero external force  $F_{\text{ext}}$ . Here v represents the background fluid velocity, and g denotes the combined other external forces.

## **Example 3** (Inverse mean curvature flow).

The inverse mean curvature flow (IMCF) is defined by setting

$$F_{\text{int}} = \frac{1}{H}, \quad F_{\text{ext}} = 0$$

in (1), it describes the evolution of a family of hypersurfaces whose normal velocity at every point on the hypersurface is equal to the reciprocal of its mean curvature at that point for all time t > 0. The IMCF finds application in general relativity and was used as a main tool to prove the *Penrose Inequal*ity by Huisken and Ilmanen (2001).

## **Example 4** (Surface diffusion flow).

The surface diffusion flow refers to the case with

$$F_{\text{int}} = \Delta_{\Gamma} H$$
,  $F_{\text{ext}} = 0$ 

in (1). Where  $\Delta_{\Gamma_t}$  denotes the surface Laplace operator on  $\Gamma_t$ , so the flow requires that the normal velocity of the hypersurface equals the surface Laplace of its mean curvature at every point on the surface for all time t > 0. The surface diffusion arises from applications in materials science, image processing and cell biology (Chopp and Sethian, 1999; Elliott and Stinner, 2010a; Escher et al., 1998).

## **Example 5** (Willmore flow).

The Willmore flow is defined by setting

$$F_{\text{int}} = -\Delta_{\Gamma_i} H - 2H(H^2 - K), \quad K := \prod_{j=1}^{d} \lambda_j, \quad F_{\text{ext}} = 0$$

in (1). Note that K denotes the Gauss curvature of the surface. The Willmore flow can be interpreted as a gradient flow for the total quadratic mean curvature functional  $\int_{\Gamma} H^2$  and find applications in biology and materials science (Bretin et al., 2015a; Du et al., 2004; Elliott and Stinner, 2010a; Marques and Neves, 2014; Simonett, 2001; Wang and Du, 2008; Zhang et al., 2009).

## **Example 6** (Hele-Shaw flow).

To define the Hele-Shaw flow, we set

$$F_{\text{int}} = 0, \quad F_{\text{ext}} = \frac{1}{2} \left[ \frac{\partial w}{\partial n} \right]_{\Gamma}$$

in (1), where w is defined by

$$\Delta w = 0$$
 in  $\Omega \backslash \Gamma_t$ ,  $w = \sigma H$  on  $\Gamma_t$ ,

here  $\Omega \in \mathbf{R}^d$  is a bounded domain,  $\frac{\partial w}{\partial n}$  stands for the normal derivative of w and  $[w]_{\Gamma_t}$  denotes the jump of a function w across the interface  $\Gamma_t$ .  $\sigma$  is a positive constant and is called the surface tension in the literature. Note that we omit the outer boundary condition for w. The Hele-Shaw flow says that the normal velocity of the interface equals the jump of the normal derivative of the pressure field w across the interface. Since the pressure depends on the mean curvature H, then the normal velocity implicitly depends on the mean curvature H. The Hele-Shaw flow arises in the study of two-phase fluids. It should be noted that the identical model also arises in materials sciences and is known as the Mullins-Sekerka model, though function w stands for the temperature, not the pressure, in the Mullins-Sekerka model (Hele-Shaw, 1898b; Mullins and Sekerka, 1963).

## **Example 7** (Generalized Stefan problem).

The generalized Stefan problem can be defined by setting

$$F_{\text{int}} = 0, \quad F_{\text{ext}} = \frac{1}{2} \left[ \frac{\partial \phi}{\partial n} \right]_{\Gamma},$$

in (1), where  $\phi$  is defined by

$$\frac{\partial \phi}{\partial t} - \Delta \phi = 0 \quad \text{in } \Omega \backslash \Gamma_t,$$

$$\phi = \sigma(H - \alpha V_n) \quad \text{on } \Gamma_t,$$

here  $\alpha$  is another positive constant. Compared with the Hele-Shaw flow, the temperature  $\phi$  now satisfies the heat equation in  $\Omega \setminus \Gamma_t$  and the normal velocity  $V_n$  appears in the boundary condition for  $\phi$  (thus, it appears in both sides of (1)). Note that we omit the outer boundary condition and the initial condition

for  $\phi$ . The generalized Stefan problem describes the solidification process of fluids with consideration of the surface diffusion (characterized by  $\sigma$ ) and the super/under cooling effect (characterized by  $\alpha$ ) (Caginalp and Chen, 1998; Fix, 1983; Langer, 1986).

## **Example 8** (Two-phase immiscible fluids).

The sharp interface two-phase immiscible flow model is often written as

$$\rho\left(\frac{\partial \mathbf{u}}{\partial t} + (\mathbf{u} \cdot \nabla)\mathbf{u}\right) - \operatorname{div} T(\mathbf{u}, p) = \mathbf{f} \qquad \text{in } \Omega_T \backslash \Gamma_t,$$

$$\operatorname{div} \mathbf{u} = 0 \qquad \text{in } \Omega_T \backslash \Gamma_t,$$

$$[T(\mathbf{u}, p)n]_{\Gamma_t} = \sigma H \qquad \text{on } \Gamma_t,$$

$$[\mathbf{u}]_{\Gamma_t} = 0 \qquad \text{on } \Gamma_t,$$

$$T(\mathbf{u}, p) := \nu(\nabla \mathbf{u} + \nabla \mathbf{u}^T) - pI,$$

where  $\Omega_T := \Omega \times (0, T)$ , **u** and *p* denote the fluid velocity and pressure.  $\nu(>0)$  is the viscosity coefficient. Again,  $[\cdot]_{\Gamma_t}$  denotes the jump function. The first two equations in the PDE system are called the Navier–Stokes equations.  $T(\mathbf{u}, p)$  is called the deformation tensor which can be replaced by more complicated nonlinear version in the case when complex fluids are involved (Anderson et al., 1998; Feng et al., 2005a; Jacqmin, 1999; Liu and Shen, 2003a; Lowengrub and Truskinovsky, 1998; Yue et al., 2004). To fit the above model into the form (1), we simply set

$$F_{\text{int}} = 0$$
,  $F_{\text{ext}} = [T(\mathbf{u}, p)n]_{\Gamma_t}$ .

It follows from (1) and the first interface condition in the PDE system is  $V_n = \sigma H$ , hence the fluid interface  $\Gamma_t$  evolves as a scaled mean curvature flow.

## **Example 9** (Two-phase immiscible fluids).

An interesting special case of two-phase flows is so-called the two-phase Hele-Shaw flow which refers to the motion of (one or more) viscous fluids between two flat parallel plates separated by an infinitesimally small gap. Such a physical setup is often called a Hele-Shaw cell and was originally designed by Hele-Shaw to study two dimensional potential flows (Feng and Wise, 2012; Han and Wang, 2018; Hele-Shaw, 1898b; Kim and Lowengrub, 2005; Wang and Zhang, 2013). The sharp interface model of two-phase Hele-Shaw flows is given by

$$\begin{split} \mathbf{u} &= -\frac{1}{12\eta} (\nabla p - \rho \mathbf{g}) & \text{in } \Omega_T \backslash \Gamma_t, \\ \operatorname{div} \mathbf{u} &= 0 & \text{in } \Omega_T \backslash \Gamma_t, \\ [p] &= \gamma H & \text{on } \Gamma_t, \\ [\mathbf{u} \cdot n] &= 0 & \text{on } \Gamma_t, \end{split}$$

where the first interface condition is called the Laplace-Young condition in which  $\gamma$  is the dimensionless surface tension coefficient and H stands for the mean curvature of  $\Gamma_t$ . To fit the above model into the form (1), we set

$$F_{\text{int}} = 0$$
,  $F_{\text{ext}} = [p]_{\Gamma}$ ,

which says that the fluid interface  $\Gamma_t$  evolves as a scaled mean curvature flow.

#### Mathematical formulations and methodologies 2.3

As Eq. (1) is a geometric problem, in order to analyze and approximate its solutions, one needs to choose an appropriate formulation and to define the meaning of solutions, which in turn then lead to different mathematical and numerical methods for the problem.

Several mathematical formulations of (1) have been proposed and developed in the past 40 years. Among them the best known one is the parametric formulation (cf. Ecker (2004) and Zhu (2002) and the references therein), in which the coordinates of all points on the hypersurface are written as functions of chosen parameters. All best know geometric quantities can be computed in terms of the coordinate functions. The parametric formulation has been widely used to study smooth geometric surface evolutions. It has also been used to develop numerical methods for computing the solutions (i.e., the coordinate functions). In order to deal with nonsmooth geometric surface evolutions, other formulations and weak solution concepts have been introduced and investigated. One of such formulations is the varifold formulation developed by Brakke (1978), especially for the mean curvature flow. Its measurevalued weak solutions are called varifold solutions. Another weak formulation and weak solution notion, called the theory of minimal barriers, were introduced by De Giorgi (1994) and were further developed by Bellettini and Novaga (1997) and Bellettini (2013) including establishing the connection between barrier solutions and other type weak solutions. The fourth formulation, which has been very popular for both PDE analysis and numerical approximation, is the level set formulation (Osher and Fedkiw, 2003; Osher and Sethian, 1988; Sethian, 1999). The corresponding weak solutions are called level set solutions (Chen et al., 1991; Evans and Spruck, 1991). The level set formulation provides a convenient and effective formalism not only for analyzing curvature-driven flows such as the mean curvature flow but also for approximating their (level set) solutions numerically (see more details below). Finally, the fifth formulation of (1) is the phase field formulation (Anderson et al., 1998; Fife, 1988; Fix, 1983; Langer, 1986; McFadden, 2002), which is the main subject of this chapter. Like the level set formulation, the phase field formulation also provides a convenient and effective formalism for both mathematical analysis and numerical approximation of problem (1), especially, in the case when  $F_{\text{ext}} \not\equiv 0$ .

## Level set and phase field formulations of the MCF

We now use the mean curvature flow (MCF) as an example to demonstrate the derivations of both the level set and phase field formulations of the MCF.

The level set method was introduced by Osher and Sethian (1988) for general moving interface problems. As its name indicates, the main idea of the level set formulation/method is to represent the hypersurface  $\Gamma_t$  as a zero-level set of a function u in  $\mathbf{R}^{d+1}$ , that is

$$\Gamma_t := \{ x(t) \in \Omega; u(x(t), t) = 0 \},$$
 (2)

and then to evolve the level set function u, instead of the interface  $\Gamma_t$ . To transfer Eq. (1) for  $\Gamma_t$  to an equation for u, we formally differentiate equation u(x(t), t) = 0 with respect to t, while treating x = x(t) as an implicit function, and using the chain rule to get

$$\frac{\partial u}{\partial t} + \nabla u \cdot \frac{dx}{dt} = 0.$$

Since  $V = \frac{dx}{dt}$  is the velocity of the surface, then

$$\frac{\partial u}{\partial t} + \nabla u \cdot V = 0. \tag{3}$$

Eq. (3) is often called the level set equation, and it is determined by the velocity field V and the initial condition  $u_0$  such that  $\Gamma_0 = \{x \in \mathbb{R}^{d+1}; u_0(x) = 0\}$ . To illustrate, we consider the mean curvature flow described in Example 1 whose geometric law is

$$V_n(t, \cdot) = H(t, \cdot).$$

By the differential geometry facts we have

$$n = -\frac{\nabla u}{|\nabla u|}$$
 and  $H = -\operatorname{div}(n)$ ,

where n stands for the inward normal to  $\Gamma_t$ . Then the level set equation (3) becomes

$$0 = \frac{\partial u}{\partial t} + \nabla u \cdot V = \frac{\partial u}{\partial t} - |\nabla u|V_n = \frac{\partial u}{\partial t} - |\nabla u|H = \frac{\partial u}{\partial t} - |\nabla u|\operatorname{div}\left(\frac{\nabla u}{|\nabla u|}\right),$$

or

$$\frac{\partial u}{\partial t} - |\nabla u| \operatorname{div}\left(\frac{\nabla u}{|\nabla u|}\right) = 0. \tag{4}$$

Eq. (4) is the well-known level set formulation of the mean curvature flow (Chen et al., 1991; Evans and Spruck, 1991; Giga, 2006).

It should be pointed out that here we only showed that Eq. (4) holds on  $\Gamma_t$ , but it easy to verify that it actually holds on all level sets, hence, holds in  $\mathbf{R}^{d+1}$ . It is also important to note that the level set equation (3) remains the same for all moving interface problems although the velocity V may be different for different problems.

The phase field method for the moving interface problems can be traced back to Rayleigh (1899), Rowlinson (1979) and Edwards et al. (1991). It was originally developed as a model for solidification, but has been used for many other applications, such as crack propagation, electromigration, crystal and tumour growth. The main idea of phase field method is to introduce thickness to the interface, more precisely, it seeks a phase field function  $u^{\varepsilon}$  such that the interface lies in the narrow region (called the diffuse interface)

$$\Gamma_t \subset Q_t^{\varepsilon} := \left\{ x(t) \in \mathbf{R}^{d+1} : |u^{\varepsilon}(x(t), t)| \le 1 - \mathcal{O}(\varepsilon) \right\}. \tag{5}$$

Here  $\varepsilon$  is a small positive constant, which controls the width of  $Q_t^{\varepsilon}$ . The phase field function takes two distinct value +1 and -1, which represent two distinct phases, with a smooth change between -1 and +1 in  $Q_t^{\varepsilon}$ . The zero-level set  $\Gamma_t^{\varepsilon} := \{x(t) \in \mathbf{R}^{d+1}; u^{\varepsilon}(x(t), t) = 0\}$  of  $u^{\varepsilon}$ , which is contained in the diffuse interface  $Q_t^{\varepsilon}$ , is often chosen to represent  $\Gamma_t$  approximately. Like the level set method, this diffuse interface approach provides a convenient mathematical formalism for numerically approximating moving interface problems because explicitly tracking the interface is not needed in the formulation. Unlike the level set method, there is no master phase field equation which is valid for all moving interface problems, instead, the phase field formulation is problem-dependent and are often difficult to derive. The difficulty is due to the fact that the interface lies inside  $Q_t^{\varepsilon}$ , but the specified location of the interface is unknown, so the curvature at the interface cannot be calculated exactly as in the level set method. Below we again use the mean curvature flow (MCF) as an example to show a formal derivation of its phase field formulation when the profile of the interface is postulated. To this end, let d(x) denote the signed distance function between point x and the interface  $\Gamma_t$ , and consider the fact that the solution approximates the  $tanh(\cdot)$  function, we heuristically postulate that

$$u^{\varepsilon}(x,t) := \tanh\left(\frac{d(x)}{\sqrt{2}\varepsilon}\right),$$
 (6)

because it matches with the desired profile for the phase function  $u^{\varepsilon}$ . Then we have

$$\tanh'(s) = 1 - \tanh^2(s), \quad \tanh''(s) = -2\tanh(s)(1 - \tanh^2(s)),$$

and

$$\begin{split} \nabla u^{\varepsilon}(x) &= \frac{\tanh\left(\frac{d(x)}{\sqrt{2}\varepsilon}\right)}{\sqrt{2}\varepsilon} \nabla d(x), \\ D^2 d(x) &= \frac{\sqrt{2}\varepsilon}{1 - \left(u^{\varepsilon}(x)\right)^2} \left(D^2 u^{\varepsilon}(x) + \frac{2u^{\varepsilon}(x)}{1 - \left(u^{\varepsilon}(x)\right)^2} \nabla u^{\varepsilon}(x) \otimes \nabla u^{\varepsilon}(x)\right). \end{split}$$

By differential geometry facts we have

$$|\nabla d(x)| = 1$$
,  $|\nabla u^{\varepsilon}(x)|^2 = \frac{1}{2\varepsilon^2} \left(1 - \left(u^{\varepsilon}(x)\right)^2\right)^2$ ,

hence,

$$H = \operatorname{tr}(D^2 d(x)) = \frac{\sqrt{2}\varepsilon}{1 - (u^{\varepsilon}(x))^2} \left( \Delta u^{\varepsilon}(x) + \frac{1}{\varepsilon^2} (u^{\varepsilon}(x) - (u^{\varepsilon}(x))^3) \right). \tag{7}$$

Recall that the approximate moving interface  $\Gamma^{\varepsilon}_{t}$  is represented by the zero-level set of  $u_t^{\varepsilon}$ , that is,  $u_t^{\varepsilon}(x, t) = 0$  on  $\Gamma_t^{\varepsilon}$ . As in the derivation of the level set equation, we formally differentiate equation  $u^{\varepsilon}(x, t) = 0$  with respect to t, while treating x = x(t) as an implicit function, and using the chain rule to get

$$0 = \frac{\partial u^{\varepsilon}}{\partial t} + \nabla u^{\varepsilon} \cdot V = \frac{\partial u^{\varepsilon}}{\partial t} - |\nabla u^{\varepsilon}| V_n = \frac{\partial u^{\varepsilon}}{\partial t} - |\nabla u^{\varepsilon}| H, \tag{8}$$

here we have used the facts that  $V = \frac{dx(t)}{dt}$  and  $n = -\frac{\nabla u^{\epsilon}}{|\nabla u^{\epsilon}|}$ . Combining (7) and (8), we obtain the follow phase field equation for the MCF:

$$\frac{\partial u^{\varepsilon}}{\partial t} - \Delta u^{\varepsilon} + \frac{1}{\varepsilon^2} \left( (u^{\varepsilon})^3 - u^{\varepsilon} \right) = 0. \tag{9}$$

Eq. (9) is called the Allen–Cahn equation in the literature, it was introduced by Allen and Cahn (1979) as a model to describe the phase separation process of a binary alloy at a fixed temperature. In the original Allen–Cahn equation  $u^{\varepsilon}$ denotes the concentration of one of the two species of the alloy. We also remark that Eq. (9) differs from the original Allen–Cahn equation in the time scale, t here represents  $\frac{t}{\epsilon^2}$  in the original Allen–Cahn equation, hence, it is a fast time.

The Allen–Cahn equation can also be derived using the modern energetic approach. To this end, we introduce the following Cahn-Hilliard free energy functional

$$\mathcal{J}(u) := \int_{\Omega} \left( \frac{1}{2} |\nabla u|^2 + \frac{1}{\varepsilon^2} F(u) \right) dx, \quad F(u) := \frac{1}{4} (u^2 - 1)^2. \tag{10}$$

where the first term in  $\mathcal{J}$  is called the bulk energy and the second term is called the interfacial (or potential) energy. Then (9) can be interpreted as the  $L^2$ -gradient flow for  $\mathcal{J}$ , that is,

$$\frac{\partial u}{\partial t} = -\mathcal{J}'(u) \quad \text{in } L^2(\Omega), \tag{11}$$

where  $\mathcal{J}'(u)$  denotes the Gâteaux derivative of  $\mathcal{J}$  at u in the specified topological space. For the  $L^2(\Omega)$  space, it is easy to verify that

$$\mathcal{J}'(u) = -\Delta u + \frac{1}{\varepsilon^2} f(u), \tag{12}$$

where  $f(u) := F'(u) = u^3 - u$  for the double well potential given in (10). We thus get a concrete form of (11) given by

$$\frac{\partial u}{\partial t} = \Delta u - \frac{1}{\varepsilon^2} f(u). \tag{13}$$

Clearly, the above energetic approach is quite simple and systematic compared to the traditional derivation which could be involved for complex moving interface problems. The key ingredient/step is to design a "correct" energy functional  $\mathcal J$  and to choose the "right" topology (or inner product) for a specific problem, the resulting phase field equation is then given by the general gradient flow equation (11). For this reason we may regard Eq. (11) as the (master) phase field equation.

## 2.5 Phase field formulations of other moving interface problems

In this section we list the known phase field formulations for all moving interface problems introduced in Section 2.2.

For the generalized mean curvature flow defined in Example 2, its phase field formulation is given by the following *convective* Allen–Cahn equation (Caginalp and Chen, 1998; Chen et al., 1998):

$$\frac{\partial u^{\varepsilon}}{\partial t} - \Delta u^{\varepsilon} + \mathbf{v} \cdot \nabla u^{\varepsilon} + \frac{1}{\varepsilon^{2}} f(u^{\varepsilon}) = g. \tag{14}$$

For the surface diffusion flow defined in Example 4, its phase field formulation is the following *degenerate* Cahn–Hilliard equation (Cahn et al., 1996):

$$\frac{\partial u^{\varepsilon}}{\partial t} + \operatorname{div}\left(b(u^{\varepsilon})\nabla\left(\Delta u^{\varepsilon} - \frac{1}{\varepsilon^{2}}f(u^{\varepsilon})\right)\right) = 0, \quad b(z) := 1 - z^{2}$$
 (15)

where  $b(z) := 1 - z^2$ .

For the Willmore flow introduced in Example 5, its phase field formulation is given by the following fourth-order PDE (Bellettini, 2013; Bellettini and Novaga, 1997; Bretin et al., 2015b; Du et al., 2004):

$$\frac{\partial u^{\varepsilon}}{\partial t} + \Delta \left( \Delta u^{\varepsilon} - \frac{1}{\varepsilon^{2}} f(u^{\varepsilon}) \right) - \frac{1}{\varepsilon^{2}} f'(u) \left( \Delta u^{\varepsilon} - \frac{1}{\varepsilon^{2}} f(u^{\varepsilon}) \right) = 0, \tag{16}$$

corresponding to the following phase field relaxation of the Willmore energy functional:

$$W(u) := \frac{1}{2\varepsilon} \int_{\Omega} \left( \varepsilon \Delta u^{\varepsilon} - \frac{1}{\varepsilon} f(u^{\varepsilon}) \right)^{2} dx. \tag{17}$$

For the Hele-Shaw flow defined in Example 6, its phase field formulation is given by the following well-known Cahn-Hilliard equation (Alikakos et al., 1994; Cahn and Hilliard, 1958; Pego, 1989):

$$\frac{\partial u^{\varepsilon}}{\partial t} + \Delta \left( \varepsilon \Delta u^{\varepsilon} - \frac{1}{\varepsilon} f(u^{\varepsilon}) \right) = 0, \tag{18}$$

which can also be obtained from (15) after setting  $b(z) \equiv 1$ .

For the generalized Stefan problem introduced in Example 7 of Section 2.2, its phase field formulation is given by the following so-called classical phase field model (Caginalp and Chen, 1998):

$$\alpha(\varepsilon)\frac{\partial u^{\varepsilon}}{\partial t} = \varepsilon \Delta u^{\varepsilon} - \frac{1}{\varepsilon}f(u^{\varepsilon}) + s(\varepsilon)\phi^{\varepsilon}$$
(19)

$$c(\varepsilon)\frac{\partial\phi^{\varepsilon}}{\partial t} = \Delta\phi^{\varepsilon} - \frac{\partial u^{\varepsilon}}{\partial t}.$$
 (20)

Note that  $u^{\varepsilon}$  denotes the phase function in the model.

Remark 1. There is also a phase field model for generalized anisotropic Stefan problem which is widely used in the materials science community (Boettinger et al., 2002; McFadden, 2002; Plapp and Karma, 2000; Provatas and Elder, 2011).

For the two-phase-fluid diffuse interface model introduced in Example 8 of Section 2.2, when  $\rho = 1$  and  $\nu = 1$ , its phase field formulation is given by the following coupled Navier-Stokes and Cahn-Hilliard system (Anderson et al., 1998; Jacqmin, 1999; Liu and Shen, 2003a):

$$\frac{\partial \mathbf{u}^{\varepsilon}}{\partial t} + (\mathbf{u}^{\varepsilon} \cdot \nabla)\mathbf{u}^{\varepsilon} - \Delta \mathbf{u}^{\varepsilon} + \nabla p^{\varepsilon} = \mu^{\varepsilon} \nabla \varphi^{\varepsilon} + \mathbf{f} \qquad \text{in } \Omega_{T}, \quad (21)$$

$$\operatorname{div} \mathbf{u}^{\varepsilon} = 0 \qquad \qquad \operatorname{in} \Omega_{T}, \quad (22)$$

$$\frac{\partial \varphi^{\varepsilon}}{\partial t} - \Delta \mu^{\varepsilon} + \mathbf{u}^{\varepsilon} \cdot \nabla \varphi^{\varepsilon} = 0 \qquad \text{in } \Omega_{T}, \quad (23)$$

$$\mu^{\varepsilon} = -\varepsilon \Delta \varphi^{\varepsilon} + \frac{1}{\varepsilon} f(\varphi^{\varepsilon}) \quad \text{in } \Omega_{T}.$$
 (24)

Note that  $\varphi^{\varepsilon}$  is used to denote the phase function in the model.

Finally, the phase field formulation for the two-phase Hele-Shaw flow model described in Example 9 of Section 2.2 is given by the following coupled Darcy and Cahn-Hilliard system (Chen et al., 2017; Diegel et al., 2015; Feng and Wise, 2012; Han and Wang, 2018; Wang and Zhang, 2013):

$$\mathbf{u}^{\varepsilon} = -\nabla p^{\varepsilon} - \gamma \varphi^{\varepsilon} \nabla \mu^{\varepsilon} \quad \text{in } \Omega_{T}, \tag{25}$$

$$\operatorname{div} \mathbf{u}^{\varepsilon} = 0 \qquad \qquad \operatorname{in} \Omega_{T}, \qquad (26)$$

$$\frac{\partial \varphi^{\varepsilon}}{\partial t} - \Delta \mu^{\varepsilon} + \mathbf{u}^{\varepsilon} \cdot \nabla \varphi^{\varepsilon} = 0 \qquad \text{in } \Omega_{T}, \tag{27}$$

$$\mu^{\varepsilon} = -\varepsilon \Delta \varphi^{\varepsilon} + \frac{1}{\varepsilon} f(\varphi^{\varepsilon}) \quad \text{in } \Omega_{T}.$$
 (28)

Again,  $\varphi^{\varepsilon}$  is used to denote the phase function.

Remark 2. Huisken and Ilmanen (2001) proposed a sublevel set formulation for the inverse mean curvature flow (IMCF), as expected, this formulation does not have the form of the general level set equation (3), which would give the following unusual PDE:

$$\operatorname{div}\left(\frac{\nabla u}{|\nabla u|}\right) \cdot \frac{\partial u}{\partial t} = |\nabla u|. \tag{29}$$

To the best of our knowledge, no phase field formulation for the IMCF was proposed in the literature, however, for a phase function  $u^e$  there holds the approximate mean curvature formula (Bellettini, 2013; Bretin et al., 2014; Du et al., 2005):

$$H^{\varepsilon} = \frac{1}{2} \left( \Delta u^{\varepsilon} - \frac{1}{\varepsilon^2} f(u^{\varepsilon}) \right). \tag{30}$$

Using this formula, the chain rule and the geometric law for the IMCF, we then get the following phase field formulation for the IMCF:

$$\frac{1}{2} \left( \Delta u^{\varepsilon} - \frac{1}{\varepsilon^{2}} f(u^{\varepsilon}) \right) \cdot \frac{\partial u^{\varepsilon}}{\partial t} = |\nabla u^{\varepsilon}|, \tag{31}$$

which is also an unusual PDE.

Finally, we like to point out that similar to the formulation of the Allen–Cahn equation as an  $L^2$ -gradient flow discussed earlier in (11) and (12), many of the phase field models presented here can also be obtained by the energetic approach. For example it is well known (Alikakos et al., 1994; Chen, 1996; Fife, 1988) that the Cahn–Hilliard equations given in Example 6 is the  $H^{-1}$ -gradient flow of the Cahn–Hilliard free energy  $\widehat{\mathcal{J}} := \varepsilon \mathcal{J}$ . That is,

$$\frac{\partial u}{\partial t} = -\widehat{\mathcal{J}}'(u) \quad \text{in } H^{-1}(\Omega), \tag{32}$$

where the  $H^{-1}$  inner product is defined by

$$(u,v)_{H^{-1}} = ((-\Delta)^{-1}u,v)_{L^2},$$

where  $(\cdot,\cdot)_{L^2}$  denotes the standard  $L^2$ -inner product, and  $(-\Delta)^{-1}$  denotes the inverse of the negative Laplacian (subject to appropriate boundary and normalization conditions).

The gradient flow structure implies in particular the dynamic energy law

$$\frac{\partial \widehat{\mathcal{J}}(u)}{\partial t} = -\left\| \frac{\partial u}{\partial t} \right\|^2,\tag{33}$$

where the  $\|\cdot\|$  denotes the norm induced by the inner product that defines the gradient flow.

In the same fashion, the degenerate Cahn–Hilliard equation given in (33) can also be interpreted as a weighted  $H^{-1}$ -gradient flow of the free energy  $\widehat{\mathcal{J}}$  with the energy law

$$\frac{\partial \widehat{\mathcal{J}}(u)}{\partial t} = - \| b(u) \nabla \mu \|_{L^{2}}^{2}, \tag{34}$$

where  $\mu := -\varepsilon \Delta u + \frac{1}{\varepsilon} f(u)$  is often called the *chemical potential*. On the other hand, Eq. (16) in Example 5 is the  $L^2$ -gradient flow of the Willmore energy functional (17). The energy law for the coupled Navier-Stokes and Cahn-Hilliard system in Example 8 can also be formulated using the idea of flow map for the total energy that is the sum of the phase field interfacial energy and the fluid kinetic energy.

#### Relationships between phase field and other formulations 2.6

Among five mathematical formulations/methodologies for moving interface problems, the first four, namely, the parametric formulation, the (Brakke's) varifold formulation, the (De Georgi's) barrier formulation and the level set formulation, all intend to represent and to seek the sharp interface exactly. For this reason we call them sharp interface formulations/methodologies. In contrast, the phase field formulation and methodology does not represent the interface exactly but only seeks an approximate interface, it introduces a width to the interface so the interface is diffused. For this reason, it is often called a diffuse interface formulation/methodology. So the phase field methodology is fundamentally different from other methodologies. On the other hand, if we use how the interface is represented and sought as a yardstick, we also can divide the five formulations/methodologies differently as direct and *indirect* formulations/methodologies. The parametric, the varifold and the barrier's formulations/methodologies belong to the direct approach camp because they represent and seek the interface directly, while the level set and the phase field methodologies belong to the indirect approach camp because they both represent and seek the interface indirectly as a level set of an auxiliary function (i.e., the level set function and phase field function/ variable). Comparative studies of phase field methods and other methods are of much interests, see for example, a comparison of phase field and parametric front tracking reported in Barrett et al. (2014a).

Indeed, the level set method and the phase field method are intimately connected. Their connection is revealed in the limiting process as the parameter  $\varepsilon \to 0^+$  in the phase field formulations/equations. It can be shown (rigorously or formally) that the diffuse interface converges (in some sense) to the sharp interface defined by the level set formulation (Abels, 2015; Abels and Lengeler, 2014; Abels and Liu, 2018; Alikakos et al., 1994; Chen, 1992, 1996; de Mottoni and Schatzman, 1995; Evans et al., 1992; Pego, 1989). It should be noted that the convergence may hold beyond onset of singularities. This is quite important because it says that theoretically the level set formulation provides a nice mathematical framework and a good solution notion for the underlying moving interface problem which the phase field method aims to approximate and to solve, although numerically both methods provide different formalisms and platforms for developing efficient numerical methods for moving interface problems.

On the other hand, the level set method and the phase field method are fundamentally different methodologies because they are based on different principles/ideas (i.e., sharp interface vs diffuse interface). The difference also reflects on their respective PDE models. The level set PDEs often have stronger nonlinearities than their phase field counterparts. For example, the level set PDE for the mean curvature flow, see (4), is a nondivergence form quasilinear PDE whose solutions are often defined in the viscosity sense (Chen et al., 1991; Evans and Spruck, 1991), while the phase field PDE for the mean curvature flow is a semilinear PDE with constant coefficient whose solutions are defined simply by using integration by parts. Compared to the phase field method, the level set method has advantages of being simple to obtain the level set equation and to provide sharp representation of the interface. On the other hand, it often does not satisfy mass conservation, which may lead to the nonphysical motions of the interface, and strong nonlinearity of the level set PDE makes it harder to solve numerically. Since the phase field function usually has physical meaning, which makes the phase field method more convenient to handle physically or biologically charged interface motion. However, although the phase field equation is easier to discretize, to resolve the thin diffuse interface one is forced to use adaptive mesh techniques (Feng and Wu, 2005, 2008; Kessler et al., 2004; Provatas and Elder, 2011; Provatas et al., 1999) because the computation becomes intractable, especially in three-dimensional cases if uniform meshes are used. Both the level set method and the phase field method share an important advantage over the direct methods, that is, they can handle with ease singularities (or topological changes) of the interfaces.

We note that the phase field formulation also relates to other sharp interface formulations. Ilmanen (1993) proved the convergence of the solution of a phase field equation to the Brakke's varifold solution for the mean curvature flow, and in Bellettini (2013), Bellettini and Novaga (1997), etc., established the convergence of the solution of another phase field equation to the De Georgi's barrier solution for the Willmore flow.

## Phase function representations of geometric quantities

To obtain a phase field formulation for a given geometric law, it is very important to be able to represent well-known geometric quantities such as the mean curvature and Gauss curvature in terms of the phase function, this then requires to develop so-called the phase field calculus (cf. Du, 2011).

First, it is well known that the volume formula of a geometric domain in terms of a phase function  $u^{\varepsilon}$  can be derived from the integral of the phase field  $u^{\varepsilon}$ , or more generally, the integral of  $g^{\varepsilon}(u^{\varepsilon})$  where  $g^{\varepsilon}$  could take on an approximation of indicator function such that  $g^{\varepsilon}(u^{\varepsilon})$  is approximately one inside the geometric domain and zero outside (as determined by the values of  $u^{\varepsilon}$ ).

Second, the interface area formula can be expressed, as discussed before by the Cahn–Hilliard energy functional given in (10).

Other frequently used geometric quantities include the interface normal, expressed by  $\nabla u^{\varepsilon}/|\nabla u^{\varepsilon}|$ , and as already mentioned above, the formula for the mean curvature of the interface represented by the zero-level set of the phase field function  $u^{\varepsilon}$  as given in (30) (Du et al., 2004, 2006). In addition, the phase field version of Gauss curvature can also be computed (Bellettini, 2013; Bellettini and Mugnai, 2010; Bretin et al., 2014; Du et al., 2005). A form similar to (30) is given in Bellettini and Mugnai (2010) as

$$K^{\varepsilon} = \frac{1}{2\varepsilon} \left[ \left( \varepsilon \Delta u^{\varepsilon} - \frac{1}{\varepsilon} f(u^{\varepsilon}) \right)^{2} - \left| \varepsilon \nabla^{2} u^{\varepsilon} - \frac{1}{\varepsilon} f(u^{\varepsilon}) \frac{\nabla u}{|\nabla u|} \otimes \frac{\nabla u}{|\nabla u|} \right|^{2} \right],$$

where | denotes the standard Euclidean norm. Consequently, the phase field Euler-Poincaré index can be computed by integrating the Gauss curvature to retrieve topological information of the implicitly defined zero-level set surface of the phase field function (Du et al., 2005). A simple form in 2D can be found in Du et al. (2005) and is given by

$$\chi^{\varepsilon} = \frac{C_2}{\varepsilon} \int_{\Omega} \left( \Delta u^{\varepsilon} - \frac{1}{\varepsilon^2} f(u^{\varepsilon}) \right) dx, \tag{35}$$

with the 3D version given in Du et al. (2007) by

$$\chi^{\varepsilon} = \frac{C_2}{\varepsilon} \int_{\Omega} \left( \Delta u^{\varepsilon} - \frac{1}{\varepsilon^2} f(u^{\varepsilon}) \right) p(u^{\varepsilon}) dx$$

for a suitably chosen function p such as  $p(t) = -2n(1-t^2)^{n-1}t$ ,  $n \ge 1$  and normalization constants  $C_2$  and  $C_3$ . Other studies concerning topological constraints in phase field models can be found in Dondl et al. (2017)

#### 2.8 Convergence of the phase field formulation

Since the phase field method is based on a diffuse interface idea/approach, a fundamental mathematical question is whether the diffuse interface converges to a sharp interface when the width of the diffuse interface tends to zero (i.e.,  $\varepsilon \to 0$ ). It turns out that this is very difficult question to rigorously answer for all moving interface problems. It remains an open problem for many phase field models. To address this question, one first needs to know or to guess what is the limiting sharp interface formulation.

Many people contributed to the convergence proof for the Allen-Cahn problem to the MCF (cf. Bronsard and Kohn, 1991; Chen, 1992; de Mottoni and Schatzman, 1995; Kohn and Sternberg, 1989; Rubinstein et al., 1989), it was finally proved by Evans et al. (1992) that

$$\operatorname{dist}_{H}\left(\Gamma_{t}^{\varepsilon},\Gamma_{t}\right)=O(\varepsilon),\quad \text{ as }\varepsilon\to0,$$

where  $dist_H$  denotes the Hausdorff distance between two sets.

The connection between the Cahn-Hilliard equation and the Hele-Shaw problem was first formally established by Pego (1989). It was later proved by Alikakos et al. (1994) that if  $u^{\varepsilon}$  and  $\mu^{\varepsilon}$  satisfy the Cahn–Hilliard system

$$\frac{\partial u^{\varepsilon}}{\partial t} = \Delta \mu^{\varepsilon}, \quad \mu^{\varepsilon} = -\varepsilon \Delta u^{\varepsilon} + \frac{1}{\varepsilon} f(u^{\varepsilon}),$$

then

$$\mu^{\varepsilon} \to w$$
,  $\operatorname{dist}_{H}(\Gamma_{t}^{\varepsilon}, \Gamma_{t}) \to 0$  as  $\varepsilon \to 0$ 

before onset of singularities (or topological changes). Where w denotes the solution of the Hele-Shaw problem. Chen also proved in Chen (1996) using the energy method the convergence of the solution of the Cahn-Hilliard problem to a (very) weak solution of the Hele-Shaw problem.

For the surface diffusion flow  $V_n = -\alpha \Delta_{\Gamma} H$ , the convergence was formally proved by Cahn et al. (1996) if  $u^{\varepsilon}$  and  $\mu^{\varepsilon}$  satisfy the following degenerate Cahn-Hilliard equation/system:

$$\begin{split} &\frac{\partial u^{\varepsilon}}{\partial t} = \operatorname{div}\left(b(u^{\varepsilon})\nabla\mu^{\varepsilon}\right), \\ &\mu^{\varepsilon} = -\Delta u^{\varepsilon} + \frac{1}{\varepsilon^{2}}f(u^{\varepsilon}), \quad b(z) := 1 - z^{2}. \end{split}$$

For the generalized Stefan problem, Caginalp and Chen (1998) proved that if  $\varphi^{\varepsilon}$  and  $u^{\varepsilon}$  satisfies the following parabolic system (which is often called the classical phase field model)

$$\begin{split} &\alpha(\varepsilon)\frac{\partial\varphi^{\varepsilon}}{\partial t} = \varepsilon\Delta\varphi^{\varepsilon} - \frac{1}{\varepsilon}f(\varphi^{\varepsilon}) + s(\varepsilon)u^{\varepsilon},\\ &c(\varepsilon)\frac{\partial u^{\varepsilon}}{\partial t} = \Delta u^{\varepsilon} - \frac{\partial\varphi^{\varepsilon}}{\partial t}, \end{split}$$

then

$$u^{\varepsilon} \to \phi$$
,  $\operatorname{dist}_{H}(\Gamma_{t}^{\varepsilon}, \Gamma_{t}) \to 0$  as  $\varepsilon \to 0$ 

before onset of singularities (or topological changes) for various combinations of  $\alpha(\varepsilon)$  and  $s(\varepsilon)$ . Where  $\phi$  denotes the solution (temperature) of the generalized Stefan problem.

Other works concerning the sharp interface limit of Cahn-Hilliard equations and its variants can be found in Antonopoulou et al. (2018a), Dai and Du (2012, 2014), Chen et al. (2014), Dai et al. (2018), Dai and Promislow (2013), Garcke and Novick-Cohen (2000), Lee et al. (2016), and Novick-Cohen (2000). Recently a convergence proof of the Navier-Stokes-Cahn-Hilliard phase field model to the sharp interface model for two-phase fluids was carried out in Abels and Lengeler (2014), Abels et al. (2017), and Abels and Liu (2018) and for a Cahn–Larché phase field model approximating an elasticity sharp interface problem in Abels (2015).

We conclude this section by summing up the main points of the above convergence results and also make a few comments. First, the level set method is a sharp interface method, while phase field method is a diffuse interface method. The latter converges to the former as  $\varepsilon \to 0$ . Second, the level set function of each moving interface problem must satisfy a master equation, which is the following Hamilton-Jacobi equation:

$$\varphi_t + V_n |\nabla \varphi| = 0$$
 on  $\Gamma_t$ .

However, there is no such a general master equation for phase field models. Phase field equations are usually not unique and different for different moving interface problems. Third, level set formulations often give rise to quasilinear or fully nonlinear PDEs while phase field formulations result in semilinear or quasilinear PDEs which are of singularly perturbed type and involve a small parameter  $\varepsilon$ . Lastly, level set functions are purely mathematical objects and may not have physical meaning while phase field functions often represent physical quantities such as densities and concentrations.

#### 3 Time-stepping schemes for phase field models

Simulating the time evolution of phase field models has many important applications such as phase transition, microstructure coarsening and cell motion. Typically the dynamic process governed by phase field models involve a number of stages that exhibit features on multiple time and spatial scales. The main focuses on effective time discretization have largely been devoted to the construction of methods which can offer both stable long time simulations and adequate resolutions of transient phenomena. Preserving, on the discrete level, some mathematical features associated with the continuum phase field dynamics (such as the energy law and uniform pointwise bounds on the phase field variable) has taken a centre stage in much of the numerical analysis literature. It is also desirable that these feature preserving schemes can provide high level resolution (which is not limited to just a formal high order truncation error) and can be efficiently implemented (from effective solver to scalable algorithms). While a number of classical time integrators have been studied, various splitting and stabilization techniques have been proposed to address these concerns. We refer to, for example, Barrett et al. (1999), Chen and Shen (1998), Chen et al. (2000), Du and Nicolaides (1991), Du and Yang (2017), Zhang and Du (2009), Elliott and French (1987, 1989), Eyre (1998), Feng and Prohl (2003a, 2004a,c, 2005), Feng and Li (2015), Feng et al. (2016), Wu and Li (2018), Gomez and van der Zee (2018), Li et al. (2016a), Shen and Yang (2010, 2019), Shen et al. (2018), Wise et al. (2009), Yang et al. (2018) and the references cited therein.

We note in particular that past works have utilized the gradient flow structure of many phase field equations which we focus on here. As an illustration, following (32), we formulate a generic phase field equation as  $u_t = -\mathcal{J}'(u)$ with  $u_t$  being the time derivative and  $\mathcal{J}'$  representing the variation of the phase field energy with respect to a generic inner product (topology) that could be either  $L^2$  (for Allen-Cahn dynamics) or  $H^{-1}$  (for Cahn-Hilliard dynamics). We use  $\langle \cdot, \cdot \rangle$  to represent the associated inner product for convenience.

To present time discretization schemes, we let  $\{t_n\}_{n=1}^N$  denote the discrete time steps with  $t_0$  be the initial time and  $t_N = T$  be the terminal time. We use  $\tau$  to denote the discrete time step size, generically  $\tau$  can be dependent on  $t_n$ , that is,  $\tau = \tau_n$ , which gives nonuniform time steps or adaptive time steps. For the sake of simple notation, we drop the explicit dependence on n for the discussion in this section. We use  $u^n$  to denote the numerical approximation of  $u(t_n)$ , the function value of the phase field variable u at time  $t_n$ .

#### Classical schemes 3.1

By viewing phase field equations as abstract dynamic systems, many classical time discretization schemes can be used. With the gradient flow formulation, one may offer various interpretations to these classical schemes.

**Example 10** (Fully explicit/forward Euler Scheme).

Standard fully explicit Euler schemes for the gradient system  $u_t = -\mathcal{J}'(u)$  can be written as

$$u^{n+1} = u^n - \tau \mathcal{J}'(u^n). \tag{36}$$

It can be viewed as a steepest descent iteration of the phase field energy, in the corresponding topology, with a step size  $\tau$ . While popular among domain scientists due to the simplicity in the numerical implementation, the limitation on the time step size, due to stability considerations, severely affects its performance for long time integration.

## **Example 11** (Fully implicit/backward Euler Scheme).

To deal with the stiff nature of phase field models, implicit schemes that often better stability properties are better choices for long time integration than the conventional explicit Euler scheme.

The standard backward/implicit Euler scheme can be written as

$$u^{n+1} = u^n - \tau \mathcal{J}'(u^{n+1}), \tag{37}$$

which may be reformulated in a variational form as

$$\min\left\{\mathcal{J}(u) + \frac{1}{2\tau} \|u - u^n\|^2\right\},\,$$

where  $\|\cdot\|$  is the norm corresponding to the same inner product  $\langle \cdot, \cdot \rangle$  that defines the variational derivative of J.

Thus, by interpreting  $1/\tau$  as a penalty constant, one can view the backward/ implicit Euler scheme as to look for the minimum of J near  $u^n$ . A consequence is that

$$\mathcal{J}(u^{n+1}) + \frac{1}{2\tau} \|u^{n+1} - u^n\|^2 = \min \left\{ \mathcal{J}(u) + \frac{1}{2\tau} \|u - u^n\|^2 \right\} \le \mathcal{J}(u^n).$$

In particular, this leads to the decrease of energy for any step size. Nevertheless, to compute  $u^{n+1}$ , a nonconvex problem needs to be solved and condition of  $\tau$  may need to be imposed to ensure the uniqueness of the minimizer.

## Example 12 (Crank-Nicolson and its variant).

For second-order Crank-Nicolson type schemes, we may consider

$$u^{n+1} = u^n - \tau \mathcal{J}'\left(\frac{u^{n+1} + u^n}{2}\right),$$

which may be obtained via the extrapolation  $2u_*^{n+1} - u^n$  where  $u_*^{n+1}$  is the backward/implicit Euler solution. Other variants include

$$u^{n+1} = u^n - \frac{\tau}{2} \mathcal{J}'(u^{n+1}) - \frac{\tau}{2} \mathcal{J}'(u^n),$$

which solves

$$\min \left\{ \mathcal{J}(u) + \frac{1}{\tau} \|u - u^n\|^2 + \langle u - u^n, \mathcal{J}'(u^n) \rangle \right\},\,$$

instead, and a modified scheme

$$u^{n+1} = u^n - \tau \phi(u^n, u^{n+1}), \tag{38}$$

$$\langle \phi(v, w), v - w \rangle = \mathcal{J}(v) - \mathcal{J}(w) \quad \forall v, w.$$
 (39)

For the latter, we have

$$\frac{1}{\tau} (\mathcal{J}(u^{n+1}) - \mathcal{J}(u^n)) + \frac{1}{\tau} ||u^{n+1} - u^n||^2 = 0.$$

While for other schemes one can often only derive an inequality version the energy law, the above can be seen as an exact time discrete analogue of the energy law associated with the continuum model.

The use of the discrete variation  $\phi(v, w)$  has been considered in other works such as Du et al. (2011) and Furihata (2001). Different variants have also been studied based on Taylor expansions (Gomez and Hughes, 2011).

Besides classical schemes considered above, there are also works on numerical discretization of phase field models based on traditional linear multistep methods (Akrivis et al., 1998) and Runge–Kutta (RK) methods (Shin et al., 2017; Song and Shu, 2017) which also can be studied for general gradient system (Humphries and Stuart, 1994).

# 3.2 Convex splitting and stabilized schemes

It is desirable to have time discretization of the gradient flows satisfying the energy stability which refers to

$$\mathcal{J}(u^{n+1}) \leq \mathcal{J}(u^n).$$

In particular, it is often viewed that the decay of energy not only preserves the physical feature but also provides stability to numerical discretization. For some phase models, preserving the pointwise bound enjoyed by the continuum model is also an important feature of numerical schemes. While a number of classical schemes mentioned in the above share such properties, there are more variants that offer more efficient implementation and better performance.

## Example 13 (Convex splitting).

The fully implicit schemes enjoy energy stability but at the expense of solving nonlinear nonconvex problems in general. An alternative is the convex splitting for gradient flows (Elliott and Stuart, 1993; Eyre, 1998). Here, based on a decomposition  $J = J_1 + J_2$  where  $J_1$  is convex and  $J_2$  is concave, the convex splitting scheme is given by

$$u^{n+1} = u^n - \tau \mathcal{J}_1'(u^{n+1}) - \tau \mathcal{J}_2'(u^n), \tag{40}$$

which also has an equivalent variations formulation

$$\min \left\{ \mathcal{J}_1(u) + \frac{1}{2\tau} \|u - u^n\|^2 - \left\langle u - u_n, \mathcal{J}'_2(u^n) \right\rangle \right\}.$$

The convexity of  $\mathcal{J}_1$  assures the unique solution for any  $\tau > 0$ . The energy stability also follows, that is,

$$\mathcal{J}(u^{n+1}) + \frac{1}{2\tau} \|u^{n+1} - u^n\|^2 \le \mathcal{J}(u^n).$$

There are a number of higher order extensions, for example, by combining the convex splitting with Runge–Kutta methods.

## **Example 14** (Linearly-implicit stabilized schemes).

To make the computational task simpler for each time step than even that associated with the convex splitting schemes, a popular technique for getting energy stable scheme is to use a semiimplicit or coupled implicit-explicit scheme that lead to only linear systems of equation.

Among the first-order scheme, this generically amounts to a linear splitting of the  $\mathcal{J}'(u)$  into the sum  $\mathcal{J}'(u) - Au = N(u)$  and Au for some linear operator A where the first term is treated explicitly while the second is treated implicitly. That is,

$$(I + \tau A)u^{n+1} = u^n - \tau N(u^n), \tag{41}$$

Typically Au is taken to be a scalar linear combination of the variation of the quadratic energy involving  $|\nabla u|$  and u itself. The scalars in the combination are choosing to make A sufficiently coercive so that (41) becomes dissipative and energy stability can be assured.

The energy stability and the need for only linear solvers per time step have made the first-order stabilized scheme popular in practice. Similar work on second-order BDF scheme can be found in (Yan et al., 2018). For studies on other higher order stabilization, we refer to Li and Qiao (2017) and Song and Shu (2017).

## **Example 15** (Exponential integrator).

Another way of utilizing the linear splitting of the  $\mathcal{J}'(u) = Au + N(u)$  is to develop exponential integrators (Cox and Matthews, 2002; Hochbruck and Ostermann, 2010; Kassam and Trefethen, 2005). The central idea is to use an equivalent formulation of the phase field equation

$$u(t+\tau) = e^{\tau A}u(t) + \int_0^{\tau} e^{(\tau-s)A}N(u(t+s)) ds,$$
 (42)

or

$$e^{-\tau A}u(t+\tau) = u(t) + \int_0^{\tau} e^{sA}N(u(t+s)) ds.$$
 (43)

By applying different quadratures to the integral over time, various discrete integration schemes can be derived. For (42), if the quadratures are based polynomial interpolations of N(u(t + s)) using the discrete values of u from previous time steps, we end up with the ETD (exponential time difference) linear multistep schemes. Meanwhile, the ETD-RK type scheme is to approximate N(u(t + s)) via polynomial interpolations of  $\{N(u(t + s_i))\}$  where  $\{s_i\}$ are the quadrature nodes in  $(0, \tau)$  and the nodal values are constructed stage-wise through the lower order ETD-RK schemes. On the other hand, if we consider (43), a polynomial interpolation can be done for the entire integrand  $e^{sA}N(u(t+s))$  instead, which would lead to the IF-RK schemes if a multistage interpolation is used.

As an illustration, the ETD-RK1 scheme is simply

$$u^{n+1} = e^{\tau A} u^n + (e^{\tau A} - I)A^{-1}N(u^n). \tag{44}$$

and the ETD-RK2 scheme is given by

$$u^{n+1} = \tilde{u}^{n+1} + \tau^{-1}(e^{-\tau A} - 1 + \tau A)A^{-2}[N(\tilde{u}^{n+1}) - N(u^n)], \tag{45}$$

where  $\tilde{u}^{n+1}$  is obtained from (44).

For both schemes, stability can be established for various phase field models (Du and Zhu, 2005; Du et al., 2018a; Kassam and Trefethen, 2005). Long time integrations and large scale simulations have been carried out (Ju et al., 2015a; Wang et al., 2016; Zhang et al., 2016a).

We note that if a first-order Taylor expansion of the exponential is used, then we recover the first-order explicit Euler scheme if  $e^{\tau A} \sim I + \tau A$ , or the first-order semiimplicit Euler  $(I - \tau A)u^{n+1} = u^n + \tau N(u^n)$  if  $e^{-\tau A} \sim I - \tau A$ .

#### 3.3 **Schemes using Lagrangian multipliers**

Recently an interesting approach was proposed in Guillén-González and Tierra (2013) for constructing unconditional stable linear schemes, it is based on a Lagrange multiplier approach introduced in Badia et al. (2011). This approach has lately been extensively developed and expanded, two new families of time-stepping schemes have been obtained, namely, invariant energy quadratization (IEQ) (Yang, 2016; Zhao et al., 2016b) and scalar auxiliary variable schemes (SAV) schemes (Shen et al., 2018), which are applicable to a large class of free energies. Below we explain the ideas of each family using the constant mobility Cahn-Hilliard equation (18) as an example.

## **Example 16** (Invariant energy quadratization schemes).

Consider the Cahn–Hilliard equation (18) and suppose that  $F(u) \ge -C_0$  for some positive constant  $C_0$  (note that  $F'(u) = f(u) = u^3 - u$  and  $C_0 = 0$  for the Cahn-Hilliard equation). Let  $q := \sqrt{F(u) + C_0}$  and rewrite (18) as the system

$$\begin{split} u_t &= \Delta \mu, \\ \mu &= -\varepsilon \Delta u + \frac{1}{\varepsilon} G(u) q, \\ q_t &= \frac{1}{2} G(u) u_t, \quad G(u) := \frac{f(u)}{\sqrt{F(u) + C_0}}. \end{split}$$

Based on the above reformulation, the following linear implicit-explicit (IMEX) scheme, which is referred as an IEQ scheme, can be easily obtained

$$u^{n+1} - u^n = \tau \Delta \mu^{n+1}, (46a)$$

$$\mu^{n+1} = -\varepsilon \Delta u^{n+1} + \frac{1}{\varepsilon} G(u^n) q^{n+1}, \tag{46b}$$

$$q^{n+1} - q^n = \frac{1}{2}G(u^n)(u^{n+1} - u^n). \tag{46c}$$

It is easy to check that the above IEQ scheme has  $O(\tau)$  truncation error and it is unconditional energy stable for the discrete energy functional

$$E^n := \frac{\varepsilon}{2} \| \nabla u^n \|_{L^2}^2 + \varepsilon \| q^n \|_{L^2}^2.$$

It needs to be noted that the above discrete energy functional is a modification of the original energy functional and both are the same at the PDE level. For its detailed analysis and higher order extensions as well as applications to other problems, we refer to Yang and Ju (2017b) and Yang (2019) and the references therein.

## **Example 17** (Scalar auxiliary variable schemes).

Notice that although the above IEQ only solves a linear system (after a spatial discretization is applied) at each time step, its coefficient matrix varies at each time step. To remedy this drawback, the SAV approach starts with the induction of an auxiliary scalar function

$$r := \sqrt{E_1(u) + C_1}$$
, where  $E_1(u) := \int_{\Omega} F(u) dx$ ,

assuming that  $E_1(u) \ge -C_1$ . Then the original Cahn–Hilliard equation can be rewritten as the following system:

$$\begin{aligned} u_t &= \Delta \mu, \\ \mu &= -\varepsilon \Delta u + \frac{1}{\varepsilon} H(u) r, \\ r_t &= \frac{1}{2} (H(u), u_t), \quad H(u) := \frac{f(u)}{\sqrt{E_1(u) + C_1}}. \end{aligned}$$

Based on the above reformulation, the following linear IMEX scheme, which is referred as a SAV scheme, can be easily obtained:

$$u^{n+1} - u^n = \tau \Delta \mu^{n+1},\tag{47a}$$

$$\mu^{n+1} = -\varepsilon \Delta u^{n+1} + \frac{1}{\varepsilon} H(u^n) r^{n+1}, \tag{47b}$$

$$r^{n+1} - r^n = \frac{1}{2} (H(u^n), u^{n+1} - u^n), \tag{47c}$$

It should be noted that constant coefficient equations only need to be solved at each time step because auxiliary variable  $r^{n+1}$  can be eliminated. It is easy to check that the above SAV scheme has  $O(\tau)$  truncation error and it is unconditional energy stable for the discrete energy functional

$$G^n := \frac{\varepsilon}{2} \| \nabla u^n \|_{L^2}^2 + \varepsilon (r^n)^2.$$

It also needs to be pointed out that the above discrete energy functional is a modification of the original energy functional and both are the same at the PDE level. For its detailed analysis and higher order extensions as well as applications to other problems, we refer to Cheng and Shen (2018), Shen and Xu (2018) and Shen et al. (2018) and the references therein.

#### **Further considerations** 3.4

To design effective time integrator, there are other important aspects besides the construction of the time discretization. For example, spatial approximations require particular attention due to the appearance of high order spatial derivatives in many phase field models such as the Cahn-Hilliard equation, see discussions in Section 4. A related matter is the design of efficient linear and nonlinear solvers to find the discrete solutions at each time step (from Newton's methods, Quasi-Newton, optimization-based methods, preconditioner techniques, multilevel methods, and domain decomposition and scalable algorithms). While some time discretizations lead to difficult nonlinear systems, others (such as ETD schemes) avoids the linear solver completely. The latter, when combined with domain decomposition techniques, can offer high scalability as demonstrated recently (Ju et al., 2015b; Zhang et al., 2016a).

Concerning long time integration, there is much potential to get more effective simulations via variable time steps. This is discussion in the context of adaptive methods in Section 6. Another aspect is the study of time integrators for coupling phase field and other physical models such as Cahn–Hilliard–Navier–Stokes equations and Cahn–Hilliard-Microelasticity models. This will be discussed later in Section 7. We also note that the time integration of stochastic phase field models such as stochastic Allen–Cahn and Cahn–Hilliard equations are also topics studied in the literature, we refer the reader to Otto et al. (2014), Röger and Weber (2013), Furihata et al. (2018), Kovács et al. (2014, 2015) and the references therein for detailed discussions.

Related to the subject, there are algorithmic issues in search of transition states and critical nuclei of the phase field energy, which can help shed light on processes of phase transformation. Instead of gradient dynamics that lead to equilibria, gentlest ascend dynamics and shrinking dimer dynamics have been developed for converging to saddle points (Zhang and Du, 2012; Zhou et al., 2010). Various optimization algorithms and time-stepping schemes have also been developed (Zhang et al., 2016b).

## 4 Spatial discretization methods for phase field models

Spatial discretization of phase field models has been widely studied, ranging from conventional finite difference, finite element and spectral methods, to specialized techniques like radial basis functions and isogeometric analysis. In this section, we present the basic formulations of a number of spatial discretization schemes.

# 4.1 Spatial finite difference discretization

For numerical solutions of PDE models, finite difference methods are often the ones studied first before other discretization techniques. Along with the discussion of phase field modelling and simulations of free boundary problems in the 1980s, there have been works on their spatial approximations by finite difference methods (Caginalp and Lin, 1987; Lin, 1988). Much of the difference approximations are standard, for example, second-order centre differences are used to approximate the second-order spatial derivatives. Higher order spatial derivatives are then approximated by the compositions of differences.

For example, for the Allen–Cahn equation (13) and the Cahn–Hilliard equation (18) with a constant mobility, the semidiscrete in space approximation can be given, respectively, by

$$\frac{\partial u_j^h}{\partial t} = (\Delta_h u^h)_j - \frac{1}{\varepsilon^2} f(u_j^h).$$

and

$$\frac{\partial u_j^h}{\partial t} + \Delta_h \left( \varepsilon \Delta_h u_j^h - \frac{1}{\varepsilon} f(u_j^h) \right) = 0,$$

Here  $u_i^h$  denotes the numerical solution at spatial grid point (labelled by the subscript j) with a typical uniform grid spacing h.  $\Delta_h$  denotes the standard discrete Laplacian based on the second-order centre difference. Coupled with time integration schemes, one can get fully discrete approximation, see for example the work in Du and Nicolaides (1991) on the Cahn-Hilliard equation that can preserve the energy law at the discrete level. Further numerical analysis work on the difference approximations can also be found in Sun (1995) and Furihata (2001). Analysis of difference approximations in space under periodic conditions coupled with some BDF energy stable time discretization was presented in Cheng et al. (2019). For the Allen-Cahn equation, the analysis of difference approximations can be found in Chen et al. (1998).

Starting from the earlier attempts, finite difference approximations have been adopted in many subsequent computational investigations of spinodal decomposition, dendritic growth, solidification and so on, see for example, Rogers et al. (1988). In particular, there are discussions, largely based on numerical experiments, on the choices of time steps and spatial grid size with respect to the diffuse interfacial width parameter  $\varepsilon$ , see for example, Wheeler et al., 1993. In the application domain, difference approximations remain popular for many practical applications of phase field methods in recent years as exemplified by the Gordon Bell prize winning work (Takashi et al., 2011).

On the algorithm development and numerical analysis side, more recent focus has been on effective linear and nonlinear solvers, including the development of various adaptive and multigrid techniques, see Kim et al. (2004), Rosam et al. (2008) and Wise et al. (2007) and discussion in Sections 6 and 7. High order compact difference spatial approximations have also been studied (Li et al., 2016c) which, when coupled with high order time integrators, have led to record-breaking extreme scale 3D simulations of coarsening dynamics based on the Cahn-Hilliard model.

#### 4.2 **Spatial Galerkin discretizations**

Unlike the finite difference method that is based on the idea of approximating derivatives in a PDE by discrete finite difference operators, another larger class of numerical methods, called the Galerkin method, for PDEs are based on a very different idea, namely, to employ a variational principle (or a weak formulation) and to approximate an infinite-dimensional Banach (or Hilbert) space by a sequence of finite-dimensional spaces which may or may not be subspaces of the infinite-dimensional Banach space.

Again, we use the Allen–Cahn equation (13) and the Cahn–Hilliard equation (18) as two examples to demonstrate the formulation of their Galerkin approximations and the widely used four types of Galerkin methods, that is, traditional finite element methods, spectral methods, discontinuous Galerkin methods and isogeometric analysis.

A simple integration by parts immediately leads to the following weak formulation for the Allen-Cahn equation (13) complemented by the Neumann boundary condition: Find  $u:(0,T)\to V:=H^1(\Omega)$  such that

$$(u_t, v) + a(u, v) + \frac{1}{\epsilon^2} (f(u), v) = 0 \quad \forall v \in V,$$
 (48)

where and throughout this section  $(\cdot, \cdot) = (\cdot, \cdot)_{\Omega}$  denotes the  $L^2(\Omega)$ -inner product and

$$a(u, v) := (\nabla u, \nabla v).$$

For the Cahn–Hilliard equation (18) with no-flux boundary conditions, its weak formulation is defined as finding  $u:(0,T)\to \widetilde{V}=\left\{v\in H^2(\Omega);\ \frac{\partial v}{\partial n}=0\ \text{on}\ \Omega\right\}$ such that

$$(u_t, v) + \varepsilon \widetilde{a}(u, v) + \frac{1}{\varepsilon} c(u, v) = 0 \quad \forall v \in \widetilde{V},$$
(49)

where

$$\widetilde{a}(w, v) := (\Delta w, \Delta v), \quad c(w, v) := a(f(w), v).$$

Let  $N(\gg 1)$  be a positive integer and  $V_N$  denote an N-dimensional approximation of V. In addition, let  $a_N(\cdot, \cdot)$ , which is defined on  $V \cup V_N \times V \cup V_N$ , denote an approximation of  $a_N(\cdot, \cdot)$ , then the (semidiscrete in space) Galerkin method for (48) is defined as seeking  $u_N: (0, T) \to V_N$  such that

$$((u_N)_t, v_N) + a_N(u_N, v_N) + \frac{1}{\epsilon^2} (f(u_N), v_N) = 0 \quad \forall v_N \in V_N.$$
 (50)

Similarly, the (semidiscrete in space) Galerkin method for (49) is defined as seeking  $u_N:(0,T)\to \widetilde{V}_N$  such that

$$((u_N)_t, v_N) + \varepsilon \widetilde{a}_N(u_N, v_N) + \frac{1}{\varepsilon} c_N(u_N, v_N) = 0 \quad \forall v_N \in \widetilde{V}_N.$$
 (51)

To define specific Galerkin methods, one needs to choose some N-dimensional spaces  $V_N$  and  $\widetilde{V}_N$ , specific bilinear forms  $a_N(\cdot,\cdot)$ ,  $\widetilde{a}_N(\cdot,\cdot)$ and a nonlinear form  $c_N(\cdot,\cdot)$ . It turns out that the choice of the latter often depends on the choice of the former and different combinations of  $V_N$ (resp.,  $\widetilde{V}_N$ ) and  $a_N(\cdot, \cdot)$  (resp.,  $\widetilde{V}_N(\cdot, \cdot)$ ) lead to different Galerkin methods for problem (48) (resp., (49)). Below we briefly discuss four types of Galerkin methods which are widely used in the literature to approximate phase field models.

#### 4.2.1 Galerkin discretization via finite element methods

To define the finite element spatial discretization, let  $\mathcal{T}_h$  denote a quasi-uniform (conforming) triangular or rectangular mesh for the physical domain  $\Omega$  and h(>0) denote the mesh size. The finite element method can be viewed as a Galerkin method in which the finite-dimensional space  $V_N$  (resp.,  $\widetilde{V}_N$ ), denoted by  $V_h$  (resp.,  $\widetilde{V}_h$ ), consists of piecewise polynomial functions (of a fixed degree) over  $\mathcal{T}_h$ . Here one may think that  $h = O(N^{-\frac{1}{d}})$ . There are two scenarios to be considered separately. First,  $V_h \subset V$  (resp.,  $\widetilde{V}_h \subset \widetilde{V}$ ). In this case, we define  $a_N(\cdot,\cdot):=a(\cdot,\cdot)$  (resp.,  $\widetilde{a}_N(\cdot,\cdot):=\widetilde{a}(\cdot,\cdot)$ ) and the method is called a *conform*ing finite element method. Second,  $V_h \not\subset V$  (resp.,  $\widetilde{V}_h \not\subset \widetilde{V}$ ), the resulting method is called a nonconforming finite element method. In this case,  $a_N(\cdot, \cdot) = a(\cdot, \cdot)$ (resp.,  $\widetilde{a}_N(\cdot, \cdot) := \widetilde{a}(\cdot, \cdot)$ ) does not work anymore because it may not even be defined on  $V_h \times V_h$  (resp.,  $\widetilde{V}_h \times \widetilde{V}_h$ ). The remedy for coping with this difficulty depends on the structure of  $V_h$  (resp.,  $\widetilde{V}_h$ ), however, the choice  $a_N(\cdot, \cdot) = a_h(\cdot, \cdot)$ (resp.,  $\widetilde{a}_N(\cdot, \cdot) = \widetilde{a}_h(\cdot, \cdot)$  and  $c_N(\cdot, \cdot) = c_h(\cdot, \cdot)$ ) works most of the time, where

$$a_h(w,v) := \sum_{K \in \mathcal{T}_h} a_K(w,v) \quad \forall v, w \in V_h \cup V,$$
 (52)

$$\widetilde{a}_h(w,v) := \sum_{K \in \mathcal{T}_h} \widetilde{a}_K(w,v) \quad \forall v, w \in \widetilde{V}_h \cup \widetilde{V},$$
(53)

$$c_h(w,v) := \sum_{K \in \mathcal{T}_h} c_K(w,v) \quad \forall v, w \in \widetilde{V} \cup \widetilde{V},$$
(54)

and  $a_K(\cdot,\cdot):=a(\cdot,\cdot)|_K$ ,  $\widetilde{a}_K(\cdot,\cdot):=\widetilde{a}(\cdot,\cdot)|_K$  and  $c_K(\cdot,\cdot):=c(\cdot,\cdot)|_K$ , they are, respectively, the restrictions of  $a(\cdot,\cdot)$ ,  $\widetilde{a}(\cdot,\cdot)$  and  $c(\cdot,\cdot)$  on the element  $K \in \mathcal{T}_h$ . It should be noted that for the Cahn–Hilliard equation, the construction of both conforming and nonconforming finite element spaces  $\widetilde{V}_h$  has been a nontrivial problem, especially in the 3D case (Ciarlet, 1978; Hu and Zhang, 2017).

Conforming and nonconforming finite element methods for the Allen–Cahn equation were studied in Chen et al. (1998), Bates et al. (2009) and Feng and Prohl (2003a). Conforming and nonconforming finite element methods for the Cahn-Hilliard equation were studied in Barrett and Blowey (1999), Barrett et al. (1999), Elliott and French (1987, 1989), Du and Nicolaides (1991), Feng and Prohl (2004c, 2005), Du et al. (2011), Wu and Li (2018) and Zhang and Wang (2010).

## Galerkin discretization via spectral methods

The spectral element method can be viewed as a Galerkin method in which the finite dimensional space  $V_N$  (resp.,  $\widetilde{V}_N$ ) is chosen as the space of (global algebraic or trigonometric) polynomials whose degree does not exceed  $m(\gg 1)$  in each coordinate direction. Hence,  $N=(m+1)^d$ . In practice,  $V_N$  (resp.,  $\widetilde{V}_N$ ) is often written as a tensor product space and is expanded by an orthogonal polynomial basis such as Legendre polynomials and Chebyshev polynomials. Since functions in  $V_N$  (resp.,  $\widetilde{V}_N$ ) are smooth, there always hold  $V_N \subset V$  and  $\widetilde{V}_N \subset \widetilde{V}$ . Hence, all spectral methods are confirming methods and one naturally chooses  $a_N(\cdot,\cdot) = a(\cdot,\cdot)$  and  $\widetilde{a}_N(\cdot,\cdot) := \widetilde{a}(\cdot,\cdot)$  in the general Galerkin framework.

For problems defined on regular computational domains but not with periodic boundary conditions, other spectral implementations are possible such as Chebyshev spectral methods for Dirichlet boundary conditions, see discussions in the following books and recent surveys (Bernardi and Maday, 1997; Boyd, 2001; Canuto et al., 2007; Gottlieb and Orszag, 1983; Guo, 1998; Luo et al., 2016; Shen et al., 2011).

Optimal error estimates can be found in Shen and Yang (2010) for the above approximation and a number of other methods involving spectral approximations in space and semiimplicit or fully implicit in time schemes. These estimates show that for a fixed  $\varepsilon$ , the convergence rate is exponential in terms of the number of the Fourier modes or the dimension of the associated finite dimensional spaces.

Spectral methods can be extended to more complex phase field models such as those involving elastic interactions via microelasticity theory (Hu and Chen, 2001). Adaptive spectral approximations for phase field models have been implemented in the context of moving mesh methods (Feng et al., 2006).

## Galerkin discretization via DG methods

Although conceptually discontinuous Galerkin (DG) methods can be viewed as nonconforming finite element methods, it is often presented as a distinct class of Galerkin methods because they use totally discontinuous piecewise polynomials, while classical nonconforming finite element methods use piecewise polynomial functions which are not totally discontinuous across element edges/faces. So all DG methods (for second and higher order PDEs) are nonconforming Galerkin methods. In addition, as expected, the convenience of using totally discontinuous piecewise polynomial functions should add some complications for designing bilinear forms  $a_N(\cdot, \cdot)$  and  $\widetilde{a}_N(\cdot, \cdot)$  as well as the nonlinear form  $c_N(\cdot, \cdot)$ .

Again, let  $\mathcal{T}_h$  denote a (conforming or nonconforming) triangular or rectangular mesh for the physical domain  $\Omega$  and h(>0) denote the mesh size. Also let  $\mathcal{E}^{I}$  and  $\mathcal{E}^{B}$ , respectively, denote the interior and boundary edge/face sets and define  $\mathcal{E} := \mathcal{E}^I \cup \mathcal{E}^B$ , The finite dimensional space  $V_N$  (resp.,  $\widetilde{V}_N$ ), denoted by  $V_h^{\rm DG}$  (resp.,  $\widetilde{V}_h^{\rm DG}=V_h^{\rm DG}$ ) and called it the DG space, consists of piecewise polynomial functions (of a fixed degree) over  $\mathcal{T}_h$  which are totally discontinuous across element edges/faces. Here one may think that  $h = O(N^{-\frac{1}{d}})$ .

The choices for  $a_N(\cdot, \cdot)$  and  $\widetilde{a}_N(\cdot, \cdot)$  are more complicated, the following are two widely used DG discretizations of  $a(\cdot, \cdot)$  and  $\widetilde{a}(\cdot, \cdot)$  (Feng and Karakashian, 2007; Feng and Li, 2015):

$$\begin{split} a_h(w,v) &:= \sum_{K \in \mathcal{T}_h} a_K(w,v) - \sum_{e \in \mathcal{E}^l} \bigg( \langle \{\partial_n w\}, [v] \rangle_e + \lambda \langle [w], \{\partial_n v\} \rangle_e + \frac{\sigma_e}{h_e} \langle [w], [v] \rangle_e \bigg), \\ \widetilde{a}(w,v) &:= \sum_{K \in \mathcal{T}_h} \widetilde{a}_K(w,v) + \sum_{e \in \mathcal{E}^l} \bigg( \langle \{\partial_n \Delta w\}, [v] \rangle_e + \lambda \langle \{\partial_n \Delta v\}, [w] \rangle_e \\ &+ \gamma_e h_e^{-3} \langle [w], [v] \rangle_e \bigg) - \sum_{e \in \mathcal{E}} \bigg( \langle \{\Delta w\}, [\partial_n v] \rangle_e + \lambda \langle \{\Delta v\}, [\partial_n w] \rangle_e \\ &- \beta_e h_e^{-1} \langle [\partial_n w], [\partial_n v] \rangle_e \bigg) \end{split}$$

for all  $v, w \in V_h^{DG}$ . Where  $\langle \cdot, \cdot \rangle_e$  stands for the  $L^2$ -inner product on  $e, \partial_n v$  denotes the normal derivative (on e),  $[\cdot]$  and  $\{\cdot\}$  denote, respectively, the jump and average operators (on e). Moreover,  $\sigma_e, \beta_e$  and  $\gamma_e$  are under-determined positive constants, and  $\lambda = \{-1, 0, 1\}$  with  $\lambda = 1$  giving two symmetric DG bilinear forms.

The choices for  $c_N(\cdot, \cdot)$  are various. The simplest one is

$$c_h(w,v) := \sum_{K \in \mathcal{T}_h} c_K(w,v) \quad \ \forall v,w \in V_h^{\mathrm{DG}}.$$

Another more involved choice (Feng and Karakashian, 2007) is

$$c_h(w, v) := -\sum_{K \in \mathcal{T}_h} (f(u), \Delta v)_K + \sum_{e \in \mathcal{E}} \langle f(\{u\}), [\partial_n v] \rangle_e - \sum_{e \in \mathcal{E}^l} \langle f'(\{u\}) \{\partial_n u\}, [v] \rangle_e.$$

DG methods for the Allen-Cahn equation were studied in Feng and Li (2015), Guo and Xu (2018), Guo et al. (2016a) and Xia et al. (2009) and for the Cahn-Hilliard equation in Feng and Karakashian (2007), Xia et al. (2007), Kay et al. (2009), Aristotelous et al. (2015), Feng et al. (2016) and Song and Shu (2017). We also note that a weak Galerkin method was recently proposed and analyzed for the Cahn-Hilliard equation in Wang et al. (2019).

# 4.2.4 Isogeometric analysis

In recent years, the technique of isogeometric analysis has been developed with the aim at integrating traditional numerical discretization like the finite element method with computer-aided geometric design tools. Isogeometric analysis has been studied in many application domains including the discretization of phase field models. The first study in the latter direction is given in Gómez et al. (2008) for Cahn–Hilliard equations. Subsequent studies include applications to phase field models for topology optimization (Dedè et al., 2012), phase field models of brittle fracture (Borden et al., 2014), high order

equations on surfaces (Bartezzaghi et al., 2015), Darcy flows (Dedè and Quarteroni, 2018), chemotaxis (Moure and Gomez, 2018) and electrochemical reactions (Zhao et al., 2016c). Convergence studies have also been carried out, see for example (Kästner et al., 2016). While many works on isogeometric analysis are based on the Galerkin approach, there are also works that adopted other formulations like collocation method (Schillinger et al., 2015) and discontinuous Galerkin method (Zhang et al., 2015).

Isogeometric analysis adopts C<sup>1</sup> NURBS-based finite dimensional spaces so that conforming finite element framework can be utilized for spatially high order differential equations. For phase field models with regularized solutions, the high order approximations can be effective using NURBS basis as in the case of spectral method and high order DG methods. For a recent review and additional references, we refer to Gomez and van der Zee (2018).

#### Spatial mixed discretization 4.3

To overcome the difficulty caused by discretization of the biharmonic operator in the Cahn-Hilliard equation, especially in 3D case, one popular approach is to discretize the Cahn-Hilliard by mixed finite element or mixed DG methods, which are based on rewriting (18) as a system of two second-order PDEs given by

$$u_t + \Delta \mu = 0, \tag{55}$$

$$\mu - \varepsilon \Delta u + \frac{1}{\varepsilon} f(u) = 0, \tag{56}$$

and discretizing the above system either by finite element methods or by DG methods.

A semidiscrete approximation can be formulated as: finding  $(u^h, \mu^h)$  such that for any  $(v^h, q^h)$ , it holds

$$(u_t^h, v^h) - a(\mu^h, v^h) = 0,$$
 (57)

$$(\mu^h, q^h) + \varepsilon a(u^h, q^h) + \frac{1}{\varepsilon} (f(u^h), q^h) = 0.$$
 (58)

One of the early works on the fully discrete conforming finite element approximations to the mixed weak formulation of (55) and (56) was given in Copetti and Elliott (1992) which has also noted an even earlier work by Du in an unpublished preprint. The latter adopted the same modified Crank-Nicolson in time discretization as in Du and Nicolaides (1991) (also see Du et al. (2011) for similar discussions on solving the Cahn-Hilliard equation on a sphere), while the former took the backward Euler in time discretization. Mixed finite element methods for the Cahn-Hilliard equation were also studied in Feng and Prohl (2004c) and Feng and Prohl (2005) with the aim of deriving refined estimates in the sharp interface limit. Mixed least square finite element was studied in Dean et al. (1996). Mixed DG methods were developed in Kay et al. (2009), Aristotelous et al. (2015) and Feng et al. (2016). For Cahn-Hilliard equations on evolving surfaces, mixed finite element methods were formulated and analyzed in Elliott and Ranner (2015). Local DG methods were also studied for Allen-Cahn and Cahn-Hilliard equations (Guo et al., 2016b; Xia et al., 2007, 2009).

#### 4.4 Implementations and advantages of high order methods

Given the smooth phase field functions, it is expected that high order methods such as high order DG and spectral methods can yield more competitive numerical algorithms. For instance, errors of spectral approximations can be reduced exponentially as the number of spectral basis functions grow, which is known in the literature as spectral accuracy. Based on such observations, Fourier spectral methods, for example, have become particularly effective for phase field simulations in a periodic cell.

Since spatial differential operators subject to periodic conditions are diagonalizable using Fourier modes, the coupling of spectral spatial discretization with semiimplicit in time schemes, as well as ETD and IF exponential type time integration schemes, becomes particularly popular. The treatment of nonlinearity can also be greatly simplified with a collocation implementation or pseudo-spectral approximation (Chen and Shen, 1998; Cheng et al., 2016).

Taking, for example, the Fourier spectral method for the Allen-Cahn model over a periodic domain  $[-\pi, \pi]^d$ , we let k denote a d-dimensional index with |k| denoting the  $\ell^{\infty}$  norm and consider a Fourier approximation given by

$$u_k^n(x) = \sum_{|k| \le K} \hat{u}_k^n e^{ikx} \in \widehat{B}_K = \operatorname{span}\{e^{ikx}\}_{|k| \le K}.$$

We can use the first-order stabilized ETD-RK1 scheme (44) to get (Du and Zhu, 2004, 2005)

$$\hat{u}_k^{n+1} = e^{-\tau k \cdot k \hat{u}_k^n - \tau \alpha \varepsilon^{-2}} + \frac{1 - e^{-\tau k \cdot k - \tau \alpha \varepsilon^{-2}}}{\tau k \cdot k \varepsilon^2 + \tau \alpha} (P_K[\widehat{f(u_k^n)}]_k) - \alpha \hat{u}_k^n),$$

where  $\tau$  is a step size,  $\alpha > 0$  is a stabilizing constant that is often taken to be larger than the one half of the Lipschitz constant of the nonlinear term f, and  $P_K$  denotes the projection to  $\widehat{B}_K$ .

Replacing  $e^{\tau k \cdot k + \tau \alpha \varepsilon^{-2}}$  by  $1 + \tau k \cdot k + \tau \alpha \varepsilon^{-2}$ , we recover

$$(1+\tau k\cdot k+\tau\alpha\varepsilon^{-2})\hat{u}_k^{n+1}=(1+\tau\alpha\varepsilon^{-2})\hat{u}_k^n-\tau\varepsilon^{-2}P_K[\widehat{f(u_k^n)}]_k.$$

Computationally, Fourier spectral methods can be implemented via FFT so that the complexity per step is  $O(K \log K)$ . Pseudo-spectral or spectral collocation methods are often adopted to treat the nonlinear term instead of the spectral projection, that is,  $f(u_k^n)$  is sampled at the discrete spatial grid points, then the FFT is applied to obtained a representation in the Fourier space.

The low computational cost and high spectral accuracy make spectral methods, when applicable, highly attractive for phase field simulations. An interesting question pertains to their performance with  $\varepsilon \ll 1$ , that is, the sharp interface limit. In Chen and Shen (1998), it has been shown for a typical benchmark setting concerning a single spherical droplet; Fourier approximations can be much more effective than low order finite difference even as  $\varepsilon$ gets significantly reduced. In other words, high order methods are competitive even in the sharp interface limit. The latter is intriguing as the sharp interface limit of phase field variables are generically discontinuous functions that often do not share high order approximations due to the lack of regularity. A justification was given in Zhang and Du (2009) when mesh and time step refinement are considered as  $\varepsilon$  is getting smaller. If the sharp interface limit is viewed as the quantity of interests to be sought after, then the typical error expected is of the form

$$O(\varepsilon^\beta) + O(\tau^\gamma,\, \varepsilon^{-\gamma}) + O(h^\delta,\, \varepsilon^{-\delta})$$

for a finite difference or finite element methods and

$$O(\varepsilon^{\beta}) + O(\tau^{\gamma}, \varepsilon^{-\gamma}) + O(e^{-cK/\varepsilon})$$

for spectral methods where  $O(\varepsilon^{\beta})$  accounts for the diffuse interface approximation error while the other two terms are due to time and spatial discretization. Thus as  $\varepsilon \to 0$ , to have a balanced total error, asymptotically the need for refinement of spectral approximation is insignificant in comparison with a low order spatial approximation.

### 5 Convergence theories of fully discrete numerical methods

In this section we shall first present a few exemplary fully discrete numerical schemes for phase field models by combining the time-stepping schemes of Section 3 and the spatial discretization methods of Section 4, we then discuss the convergence results of these fully discrete numerical methods. Below we divide those results into two groups: the first group addresses convergence and error estimates for a fixed diffusive interfacial width  $\varepsilon$ , the second group concerns the convergence and error estimates for the numerical solutions as well as the numerical interfaces as  $\varepsilon$ ,  $\tau$ ,  $h \to 0$ .

# Construction of fully discrete numerical schemes

As for most evolution equations, a fully discrete numerical method for phase field models can be easily constructed by any combination of a time-stepping scheme of Section 4 and a spatial discretization method of Section 5. Such a construction is also known as the *method of lines* in the literature (cf. Reddy and Trefethen, 1992; Schiesser, 1991 and the references therein). Below we shall use the Allen–Cahn equation (13) and the Cahn–Hilliard equation (18) to demonstrate the procedure.

We start with the spatial semidiscrete finite element method for the Allen–Cahn equation (with the Neumann boundary condition), which is given by (50) with (52), since  $V_h$  is finite dimensional, it is easy to see that (50) with (52) is a system of ODEs, which must be complemented by the initial condition

$$(u_h(0), v_h) = (u_0, v_h) \quad \forall v_h \in V_h.$$

Applying the convex splitting scheme to (50) then leads to the following fully discrete convex splitting finite element method: find  $u_h^{n+1} \in V_h$  such that

$$(d_t u_h^{n+1}, v_h) + a_h(u_h^{n+1}, v_h) + \frac{1}{\varepsilon^2} ((u_h^{n+1})^3, v_h) = \frac{1}{\varepsilon^2} (u_h^n, v_h)$$
 (59)

for all  $v_h \in V_h$ . Where  $d_t u_h^{n+1} := (u_h^{n+1} - u_h^n)/\tau$ . As expected, it is easy to check that the above method is unconditionally (in h and  $\tau$ ) and uniformly (in  $\epsilon$ ) energy stable (cf. Feng and Li, 2015; Shen and Yang, 2010).

Similarly, applying the implicit-explicit (IMEX) scheme to (50) gives the following fully discrete IMEX finite element method: find  $u_h^{n+1} \in V_h$  such that

$$(d_t u_h^{n+1}, v_h) + a_h(u_h^{n+1}, v_h) = \frac{1}{\varepsilon^2} (u_h^n - (u_h^n)^3, v_h), \quad \forall v_h \in V_h.$$
 (60)

As expected, this IMEX finite element method is not unconditional stable. To strengthen its stability, the following popular stabilized method was proposed in the literature (Feng et al., 2015; Li et al., 2016a,b; Shen and Yang, 2010; Xu and Tang, 2006): find  $u_h^{n+1} \in V_h$  such that

$$(d_{t}u_{h}^{n+1}, v_{h}) + a_{h}(u_{h}^{n+1}, v_{h}) + S(u_{h}^{n+1} - u_{h}^{n}, v_{h}) = \frac{1}{\varepsilon^{2}} (u_{h}^{n} - (u_{h}^{n})^{3}, v_{h})$$
(61)

for all  $v_h \in V_h$ . Where S is an under-determined constant or linear operator.

Now adopting the finite element method for the spatial discretization in the IEQ scheme (46) then leads to the following fully discrete IEQ finite element method for the Cahn–Hilliard equation (18): find  $(u_h^{n+1}, \mu_h^{n+1}, q_h^{n+1}) \in [V_h]^3$  such that

$$\begin{split} \left(u_h^{n+1}-u_h^n,\ v_h\right) &= -\tau a_h \left(\mu_h^{n+1},v_h\right) \quad \forall v_h \in V_h, \\ \left(\mu_h^{n+1},\ \lambda_h\right) &= \varepsilon a_h \left(u_h^{n+1},\lambda_h\right) + \frac{1}{\varepsilon} \left(G(u_h^n)q_h^{n+1},\lambda_h\right) \quad \forall \lambda_h \in V_h, \\ \left(q_h^{n+1}-q_h^n,\ p_h\right) &= \frac{1}{2} \left(G(u_h^n) \left(u_h^{n+1}-u_h^n\right),\ p_h\right) \quad \forall p_h \in V_h, \end{split}$$

where  $a_h(\cdot, \cdot)$  denotes the discrete bilinear form for  $-\Delta$  as defined in (52).

Finally, using the finite element method for the spatial discretization in the SAV scheme (47) one immediately obtains the following fully discrete SAV finite element method for the Cahn–Hilliard equation (18): find  $(u_h^{n+1}, \mu_h^{n+1}, r_h^{n+1}) \in [V_h]^2 \times \mathbf{R}$  such that

$$\begin{split} \left(u_{h}^{n+1} - u_{h}^{n}, \ v_{h}\right) &= -\tau a_{h}\left(\mu_{h}^{n+1}, \ v_{h}\right) \quad \forall v_{h} \in V_{h}, \\ \left(\mu_{h}^{n+1}, \ \lambda_{h}\right) &= \varepsilon a_{h}\left(u_{h}^{n+1}, \ \lambda_{h}\right) + \frac{1}{\varepsilon}\left(H(u_{h}^{n}), \ \lambda_{h}\right)r_{h}^{n+1} \quad \forall \lambda_{h} \in V_{h}, \\ r_{h}^{n+1} - r_{h}^{n} &= \frac{1}{2}\left(H(u_{h}^{n}), \ u_{h}^{n+1} - u_{h}^{n}\right), \end{split}$$

where  $a_h(\cdot, \cdot)$  is the same as above.

# 5.2 Types of convergence and a priori error estimates

As for many numerical methods, three important theoretical issues to be addressed for above fully discrete numerical methods are *stability*, *convergence* and *rates of convergence*. As the stability issue has already been considered when constructing the time-stepping schemes, and it is not difficult to show that the similar stability properties also hold for the corresponding fully discrete schemes, especially when Galerkin spatial discretization methods are used, in this section we only focus on the issues related to convergence and rates of convergence (i.e., error estimates) of fully discrete numerical methods for phase field models in two cases: (i)  $\varepsilon > 0$  is fixed and h,  $\tau \to 0$ ; (ii)  $\varepsilon$ , h,  $\tau \to 0$ . It should be emphasized that the stability is only a necessary criterion for screening "good" numerical methods, however, it does not guarantee the convergence, as a result, it is important to examine the convergence (and rates of convergence) for stable methods, see Xu et al. (2019) for further discussions in this direction.

Let  $u^{\varepsilon}$  denote the PDE solution of a underlying phase field model and  $u_{h,\tau}^{\varepsilon}$  denote a fully discrete numerical solution. The goal of error estimates for a fixed  $\varepsilon > 0$  is to derive the following type of error bounds:

$$\|u^{\varepsilon} - u_{h\tau}^{\varepsilon}\|_{*} \le C_{1}(\varepsilon, u^{\varepsilon})\tau^{\ell_{1}} + C_{2}(\varepsilon, u^{\varepsilon})h^{\ell_{2}}$$

$$\tag{62}$$

for some  $\varepsilon$ - and  $u^{\varepsilon}$ -dependent positive constants  $C_1$  and  $C_2$  and positive numbers (mostly integers)  $\ell_1$  and  $\ell_2$ . Where  $\|\cdot\|_*$  denotes a (function space) norm which is problem-dependent.

The precise dependence of  $C_1$  and  $C_2$  on  $\varepsilon$  are usually complicated and difficult to obtain, but they are expected to grow in  $\frac{1}{\varepsilon}$  as  $\varepsilon \to 0$ , this is because  $C_1$  and  $C_2$  depend on high order ( $\geq 2$ ) space and time derivatives of  $u^{\varepsilon}$  and those derivatives can be proved to grow *polynomially* in  $\frac{1}{\varepsilon}$  as  $\varepsilon \to 0$  (cf. Abels, 2015; Abels and Lengeler, 2014; Abels and Liu, 2018; Abels et al., 2017; Feng, 2006; Feng and Prohl, 2003a, 2004a, 2005; Feng et al., 2007a).

Here we distinguish two cases based on how  $C_1$  and  $C_2$  depend on  $\frac{1}{\epsilon}$ . If at least one of  $C_1$  and  $C_2$  grow exponentially in  $\frac{1}{\epsilon}$ , we then call (62) a coarse error estimate. On the other hand, if both  $C_1$  and  $C_2$  grow polynomially in  $\frac{1}{c}$ , the estimate is called a *fine error estimate*. Obviously, in both cases, for a fixed  $\varepsilon > 0$ , (62) implies the convergence

$$\lim_{h,\tau\to 0}\|u^{\varepsilon}-u_{h,\tau}^{\varepsilon}\|_*=0,$$

that is,  $u_{h,\tau}^{\varepsilon}$  converges to  $u^{\varepsilon}$  in the  $\|\cdot\|_*$ -norm. Hence, when  $\varepsilon > 0$  is fixed, both types of coarse and fine error estimates imply the convergence of the numerical solution as  $h, \tau \to 0$ .

On the other hand, the situation is very different if the convergence of the numerical interface to the sharp interface of the underlying geometric moving interface problem is a main concern. In that case, one must study the limiting behaviours of the numerical solution  $u_{h,\tau}^{\varepsilon}$  and the numerical interface

$$\Gamma_t^{\varepsilon,h,\tau} := \left\{ x \in \Omega; u_{h,\tau}^{\varepsilon}(x,t) = 0 \right\}$$

as  $\varepsilon \to 0$ . First, it is easy to see that the coarse error estimates will become useless because they fail to provide any useful information about the convergence. To see the point, we notice that  $\tau^{\ell_1}$  and  $h^{\ell_1}$  only decrease in polynomial orders but  $C_1$  and/or  $C_2$  grow in exponential orders, hence, the limit of the right-hand side of (62) is  $+\infty$  as  $\varepsilon$ , h,  $\tau \to 0$ . Second, on contrary the fine error estimates are still valuable, this is because  $C_1$  and  $C_2$  grow in polynomially orders, h and  $\tau$  can be chosen as powers of  $\varepsilon$  so that the right-hand side of (62) is guaranteed to converge to 0 as  $\varepsilon \to 0$ . Let  $u^0$  denote the solution of the underlying moving sharp interface problem (which is defined by its level set formulation), by the triangle inequality we have

$$||u^0 - u_{h,\tau}^{\varepsilon}||_{\dagger} \le ||u^0 - u^{\varepsilon}||_{\dagger} + ||u^{\varepsilon} - u_{h,\tau}^{\varepsilon}||_{\dagger},$$

where  $\|\cdot\|_{\dagger}$  stands for another (function space) norm which is a topology used to measure the convergence of  $u^{\varepsilon}$  to  $u^{0}$  as described in Section 2. We note that  $\|\cdot\|_{\dagger}$  is usually a weaker norm compared to the  $\|\cdot\|_{\ast}$ -norm. Thus, combining the PDE convergence result and the above numerical convergence result we get

$$\lim_{\varepsilon \to 0} \|u^0 - u_{h,\tau}^{\varepsilon}\|_{\dagger} = 0,$$

provided that h and  $\tau$  are chosen as appropriate powers of  $\varepsilon$ . Moreover, it turns out that (Feng and Li, 2015; Feng and Prohl, 2003a, 2004a, 2005; Feng et al., 2016) such a convergence result is good enough to infer the following convergence of the numerical interface:

$$\lim_{t\to 0} \operatorname{dist}_{H}\left(\Gamma_{t}, \; \Gamma_{t}^{\varepsilon, h, \tau}\right) = 0,$$

as well as its rates of convergence in powers of  $\varepsilon$  if  $\Gamma_t$  is sufficiently smooth. Where  $dist_H$  denotes the Hausdorff distance between two sets.

We shall focus on coarse error estimates in the next section and discuss fine error estimates and the convergence of numerical interfaces in the subsequent section.

#### Coarse error estimates for a fixed value $\varepsilon > 0$ 5.3

For a fixed  $\varepsilon > 0$ , all phase field equations are either semilinear or quasilinear parabolic PDEs, the desired error estimates for their fully discrete (and spatial semidiscrete) methods can be obtained by following the standard perturbation procedure which consists of three main steps: (i) decomposing the global error with help of an elliptic projection of the exact PDE solution; (ii) using the energy method to obtain an error equation/inequality for the error between the numerical solution and the elliptic projection; (iii) completing the error estimate using a discrete Gronwall's inequality and the triangle inequality. Below we shall demonstrate this perturbation procedure using scheme (59).

Let

$$R^{n+1} := u_t(t_{n+1}) - d_t u^{n+1}$$

denote the truncation error of the backward difference operator  $d_t$  which is well known to be of  $O(\tau)$  order provided that  $u_{tt} \in L^2((0, T); H^{-1}(\Omega))$ . Subtracting (59) from the weak formulation of (13) immediately yields the following error equation:

$$(d_{t}e_{h}^{n+1}, v_{h}) + a_{h}(e_{h}^{n+1}, v_{h}) + \frac{1}{\varepsilon^{2}} ((u(t_{n+1}))^{3} - (u_{h}^{n+1})^{3}, v_{h})$$

$$= \frac{1}{\varepsilon^{2}} (e_{h}^{n+1} + \tau d_{t}u_{h}^{n+1}, v_{h}) + R^{n+1} \quad \forall v_{h} \in V_{h},$$
(63)

where  $e_h^n := u(t_n) - u_h^n$  denotes the global error at  $t_n$ . Note that the super-index  $\varepsilon$  is suppressed on all functions.

Now, let  $P_h: H^1(\Omega) \to V_h$  denote the standard elliptic projection operator associated with the discrete bilinear form  $a_h(\cdot, \cdot)$  and set

$$e_h^n := \eta_h^n + \xi_h^n; \ \eta_h^n := u(t_n) - P_h u(t_n), \ \xi_h^n := P_h u(t_n) - u_h^n.$$

Then (63) implies that

$$(d_{t}\xi_{h}^{n+1}, v_{h}) + a_{h}(\xi_{h}^{n+1}, v_{h}) + \frac{1}{\varepsilon^{2}} ((P_{h}u(t_{n+1}))^{3} - (u_{h}^{n+1})^{3} - \xi_{h}^{n+1}, v_{h})$$

$$= \frac{1}{\varepsilon^{2}} (\eta_{h}^{n+1} + \tau d_{t}u_{h}^{n+1}, v_{h}) + R^{n+1} - (d_{t}\eta_{h}^{n+1}, v_{h})$$

$$- \frac{1}{\varepsilon^{2}} ((u(t_{n+1}))^{3} - (P_{h}u_{h}^{n+1})^{3}, v_{h}) \quad \forall v_{h} \in V_{h}.$$

$$(64)$$

Setting  $v_h = \xi_h^{n+1}$  and using the monotonicity of cubic power function and the stability estimates of the numerical solution  $u_h^{\varepsilon}$  (not shown here) as well as the approximation properties of  $P_h$  we get

$$\frac{1}{2}d_{t} \|\xi_{h}^{n+1}\|_{L^{2}}^{2} + a_{h}(\xi_{h}^{n+1}, \xi_{h}^{n+1}) \leq \frac{1}{\varepsilon^{2}} \|\xi_{h}^{n+1}\|_{L^{2}}^{2} + c_{1}\tau^{2} + c_{2}h^{r+1}$$
 (65)

for some positive constants  $c_i = c_i(\varepsilon)(i = 1, 2)$ . Where r, a positive integer, denotes the order of the underlying finite element method (i.e., the degree of polynomial shape functions). Here we have also used the fact that

$$\|\eta_h^{n+1}\|_{L^2} \leq Ch^{r+1}$$
.

Finally, applying a discrete Gronwall's inequality to (65) yields

$$\|\xi_h^m\|_{L^2}^2 + \tau \sum_{n=0}^{m-1} a_h(\xi_h^{n+1}, \xi_h^{n+1}) \le C \exp\left(\varepsilon^{-2}\right) \left(c_1 \tau^2 + c_2 h^{r+1}\right)$$
 (66)

for some positive constant C = C(T), which and the triangle inequality infer

$$\|e_h^m\|_{L^2}^2 \le C \exp\left(\varepsilon^{-2}\right) \left(c_1 \tau^2 + c_2 h^{r+1}\right),$$
 (67)

$$\tau \sum_{n=0}^{m-1} a_h(e_h^{n+1}, e_h^{n+1}) \le C \exp(\varepsilon^{-2}) (c_1 \tau^2 + c_2 h^r)$$
 (68)

for all 1 < m < M.

We remark that error estimates (67) and (68) are two coarse error estimates for the fully discrete implicit Euler finite element method for the Allen-Cahn equation. It turns out that they are typical error estimates for all fully discrete numerical methods for various phase field models, which can be abstractly written into the form (62). Although for different models, the norms used to measure errors and the orders of errors (indicated by  $\ell_1$ and  $\ell_2$  in (62) ) may be different, the nature of the exponential dependence on  $\frac{1}{s}$  of the constants  $C_1$  and  $C_2$  remains the same for all these methods and models when the above standard perturbation procedure is used to derive the error estimates, see Elliott and French (1987, 1989), Du and Nicolaides (1991), Barrett and Blowey (1999), Barrett et al. (1999), Xu and Tang (2006), Feng et al. (2007a, 2015), Liu and Shen (2003a), Du et al. (2004), Feng (2006), Feng and Karakashian (2007), Wise et al. (2009), Li et al. (2009), Kay et al. (2009), Shen and Yang (2010), Aristotelous et al. (2015), Li et al. (2016a), Diegel et al. (2017a), Yang (2016), Song and Shu (2017) and Shen et al. (2018).

# 5.4 Fine error estimates and convergence of numerical interfaces as $\varepsilon$ , h, au o 0

As explained earlier, the coarse error estimates obtained in the previous section are not useful to study the convergence as  $\varepsilon \to 0$ . A couple natural questions arise: (i) Are the error estimates sharp in terms of  $\varepsilon$ ? (ii) If not, what are sharp error estimates and how to establish such estimates? As all numerical experiments (cf. Provatas et al., 1999; Zhang and Du, 2009 and the references therein) indicate that the above coarse error estimates are not sharp in terms of  $\varepsilon$  (otherwise, the phase field methodology would be an impractical method to compute moving interface problems!), then the focus is on addressing the second question. The good news is that there are a couple successful numerical analysis techniques which have been developed in the last 30 years, the bad news is that they are the only known techniques for deriving so-called fine error estimates, and the applicability of both techniques is restrictive although one technique fares better than the other. Below we shall again use scheme (59) for the Allen–Cahn equation as an example to explain the ideas of both techniques.

The first technique, which was developed in Nochetto and Verdi (1997) and Nochetto et al. (1994), is to use the discrete maximum principle to derive the  $L^{\infty}$ -norm fine error estimate for  $u^{\varepsilon}-u^{\varepsilon}_{h,\tau}$  for the implicit Euler  $P_1$ -conforming finite element (i.e., r = 1), such an estimate in turn leads to the convergence proof of the numerical interface  $\Gamma_t^{\epsilon,h,\tau}$  to its underlying sharp interface limit  $\Gamma_t$ . However, this maximum principle technique only applies to the Allen-Cahn equation which does possesses a maximum principle and only to  $P_1$ -conforming finite element method in order to ensure a discrete maximum principle. The technique does not work for high order phase field model such as the Cahn-Hilliard equation nor for phase field systems which do not have a maximum principle. Moreover, it does not apply to higher order finite element methods either because higher order finite element methods do not have a discrete maximum principle in general even the underlying PDE (such as the Allen-Cahn equation) does.

Before introducing the second technique, we first have a closer look at the derivation of the coarse error estimate given in the previous section to find out the guilty part/step which leads to the exponential dependence of the error constants on  $\frac{1}{c}$ . First, since the error equation (64) is an identity, no error is introduced there. Second, after the inequality (65) is reached, noticing that the coefficient  $\frac{1}{\epsilon^2}$  in the first term on the right-hand side, then it is too late to improve the estimate because (65) and the Gronwall's inequality certainly lead to a coarse error estimate! This simple observation suggests that the guilty step is from the error equation (64) to the error inequality (65), in other words, the transition from equality to inequality is too loose which makes the right-hand side of (65) becomes a too large bound for its left-hand side. To improve the coarse error estimate, we must refine this step and to obtain a tight upper bound for the left-hand side of (65). It turns out that this is exactly what needs to be done to derive a fine error estimate for the Allen–Cahn equation, and a similar situation occurs for several other phase field models.

Moreover, since the right-hand side of (64) only involve the truncation error and the projection error, they contribute  $c_1\tau^2$  and  $c_2h^{r+1}$  terms in (65), which is good. Hence, the desired improvement (if possible) must comes from bounding the last term on the left-hand side of (64). To obtain (65) we bounded that term as two separate terms that inevitably produces the first term on the right-hand side of (65). A key idea of the second technique of the fine error analysis, which was developed in Feng and Prohl (2003a, 2004a, 2005), Feng and Li (2015) and Feng et al. (2016), is not to separate that term, instead, to estimate it together with the proceeding term. We demonstrate the steps of this techniques below.

Let  $f(u) := u^3 - u$ , then  $f'(u) = 3u^2 - 1$ . The last term on the left-hand side of (64) with  $v_h = \xi_h^{n+1}$  can be written as

$$\begin{split} \frac{1}{\varepsilon^{2}} & \left( f'(P_{h}u(t_{n+1}))\xi_{h}^{n+1}, \; \xi_{h}^{n+1} \right) + \frac{1}{\varepsilon^{2}} \left\| \xi_{h}^{n+1} \right\|_{L^{4}}^{4} - \frac{3}{\varepsilon^{2}} \left( P_{h}u(t_{n+1}), (\xi_{h}^{n+1})^{3} \right) \\ & \geq \frac{1}{\varepsilon^{2}} \left( f'(P_{h}u(t_{n+1}))\xi_{h}^{n+1}, \xi_{h}^{n+1} \right) + \frac{1}{\varepsilon^{2}} \left\| \xi_{h}^{n+1} \right\|_{L^{4}}^{4} \\ & - \frac{3}{\varepsilon^{2}} \left\| P_{h}u(t_{n+1}) \right\|_{L^{\infty}} \left\| \xi_{h}^{n+1} \right\|_{L^{3}}^{3}. \end{split}$$

Substituting the above inequality into (64) yields

$$\frac{1}{2}d_{t} \|\xi_{h}^{n+1}\|_{L^{2}}^{2} + \varepsilon^{2} \left[ a_{h} (\xi_{h}^{n+1}, \xi_{h}^{n+1}) + \frac{1}{\varepsilon^{2}} (f'(P_{h}u(t_{n+1}))\xi_{h}^{n+1}, \xi_{h}^{n+1}) \right] 
+ (1 - \varepsilon^{2}) \left[ a_{h} (\xi_{h}^{n+1}, \xi_{h}^{n+1}) + \frac{1}{\varepsilon^{2}} (f'(P_{h}u(t_{n+1}))\xi_{h}^{n+1}, \xi_{h}^{n+1}) \right] + \frac{1}{\varepsilon^{2}} \|\xi_{h}^{n+1}\|_{L^{4}}^{4} 
\leq \frac{C}{\varepsilon^{2}} \|\xi_{h}^{n+1}\|_{L^{3}}^{3} + c_{1}\tau^{2} + c_{2}h^{r+1},$$
(69)

which replaces (65). One key step of the technique is to show that there exists an  $\varepsilon$ -independent positive constant  $\widehat{C}_0$  such that

$$a_h(\xi_h^{n+1}, \xi_h^{n+1}) + \frac{1}{\varepsilon^2} (f'(P_h u(t_{n+1})) \xi_h^{n+1}, \xi_h^{n+1}) \ge -\widehat{C}_0 \|\xi_h^{n+1}\|_{L^2}^2, \tag{70}$$

which is a corollary of the following PDE spectral estimate result (Chen, 1994; de Mottoni and Schatzman, 1995):

$$\lambda_{\min} := \inf_{\varphi \in H^1} \frac{(\nabla \varphi, \nabla \varphi) + \frac{1}{\varepsilon^2} (f'(u^{\varepsilon}) \varphi, \varphi)}{\|\varphi\|_{L^2}} \ge -C_0$$

for some positive constant  $C_0$  and  $0 < \varepsilon \ll 1$ . Notice that  $\lambda_{\min}$  is the principal eigenvalue of the linearized Allen-Cahn operator

$$L_{\mathrm{AC}}(\varphi) := -\Delta \varphi + \frac{1}{\varepsilon^2} f'(u^{\varepsilon}) \varphi.$$

Another key step of the technique is to bound the first term on the righthand side of (69) as follows:

$$\frac{C}{\varepsilon^2} \|\xi_h^{n+1}\|_{L^3}^3 \leq \frac{\varepsilon^2}{2} \|\nabla \xi_h^{n+1}\|_{L^2}^2 + \widehat{C}_0 \varepsilon^2 \|\xi_h^{n+1}\|_{L^2}^2 + C \varepsilon^{\frac{-2(4+d)}{4-d}} \|\xi_h^{n+1}\|_{L^2}^{\frac{2(6-d)}{4-d}}.$$

Combining the above with (69) and (70) yields

$$d_{t} \|\xi_{h}^{n+1}\|_{L^{2}}^{2} + \varepsilon^{2} a_{h} (\xi_{h}^{n+1}, \xi_{h}^{n+1}) + \frac{2}{\varepsilon^{2}} \|\xi_{h}^{n+1}\|_{L^{4}}^{4}$$

$$\leq C \varepsilon^{\frac{-2(4+d)}{4-d}} \|\xi_{h}^{n+1}\|_{L^{2}}^{\frac{2(6-d)}{4-d}} + 2 (1+\widehat{C}_{0}) \|\xi_{h}^{n+1}\|_{L^{2}}^{2} + c_{1}\tau^{2} + c_{2}h^{r+1}.$$

$$(71)$$

Notice that the exponent  $\frac{2(6-d)}{4-d} > 2$  for all  $d \ge 1$ , the final key step of the technique is to obtain the desired fine error bound for  $\xi_h^n$  (and hence for  $e_h^n$ ) from (71) by using a discrete generalized Gronwall's inequality. See Feng and Prohl (2003a) for finite element approximations and Feng and Li (2015) for discontinuous Galerkin approximations.

From the above derivation one can see that the most important step of the second technique for fine error estimates is to establish a discrete spectral estimate based on the PDE spectral estimate for the linearized operator of the underlying PDE operator. It was proved that such a spectral estimate also holds for the Cahn-Hilliard equation/operator and for the general phase field model for the generalized Stefan problem (Alikakos and Fusco, 1993; Bates and Fife, 1990; Chen, 1994), and similar fine error estimates to above estimate were also established in Feng and Prohl (2004c, 2005) (mixed finite element approximations), Feng et al. (2016) (mixed discontinuous Galerkin approximations), Wu and Li (2018) and Li (2019) (Morley nonconforming finite element approximation) for the Cahn-Hilliard equation and in Feng and Prohl (2004a) for the general phase field model based on those spectral estimates. It should be emphasized that this technique is applicable to any phase field model as long as its linearized operator satisfies a desired spectral estimate. Hence, this technique essentially reduces a numerical problem of deriving fine error estimates for a phase field model into a problem of proving a PDE spectral estimate for the linearized phase field operator. Finally, we also note that a duality argument was recently proposed in Chrysafinos (2019) to derive unconditional stability for a space-time finite element method, it may have a potential to offer an alternative technique for fine error estimates.

# 6 A posteriori error estimates and adaptive methods

# 6.1 Spatial and temporal adaptivity

Since phase field models are singularly perturbed equations which involve a small scale parameter  $\varepsilon$  and the solutions of phase field models have distinctive profiles in the sense that they take a close-to-constant value in each bulk region, which indicates a phase of the material or fluid mixture, and vary smoothly but sharply in a thin layer (called diffuse interface) of width  $O(\varepsilon)$  between bulk regions, to resolve this kind of (solution) functions, it is necessary to use spatial mesh size smaller than  $\varepsilon$  (sometimes much smaller) in the thin layer and it is preferable to use much coarser meshes in the bulk regions. This seems to provide an ideal situation for using adaptive grid methods for efficient numerical simulations. Moreover, should a uniform mesh be used for simulations, the mesh must be very fine over the whole domain, this then results in huge nonlinear algebraic systems to solve, which may not be feasible for large scale and long time simulations, especially in high dimensions. Furthermore, the time scale in most phase field models represents so-called fast time scale, to capture the dynamics of underlying physical or biological phase transition phenomena, it is necessary to use very fine time step size. If the physical or biological processes are slowly varying, it often takes a long time for them to reach equilibrium states, when that happens, long time numerical simulations are required and adaptive time-stepping schemes must be used to make such numerical simulations feasible (Chen and Shen, 2016; Li et al., 2017; Wodo and Ganapathysubramanian, 2011; Zhang and Qiao, 2012).

Since the method of lines is the preferable approach in practice, as explained in Sections 3–5, after a spatial discretization is done, each phase field model then reduces into a system of nonlinear ordinary differential equations (ODEs), then the existing well-developed adaptive time-stepping schemes for ODEs can be utilized for temporal adaptivity. For this reason we shall not further discuss temporal adaptivity in this section and refer the reader to Iserles (2009) and Lambert (1991), instead, below we shall focus on discussing the ideas of various approaches of spatial adaptivity for phase field models.

To design spatial adaptive methods/algorithms for PDE problem, three major approaches have been extensively developed in the past 30 years, they are *h*-, *hp*- and *r*-adaptivity (see Eriksson et al. (1996), Ainsworth and Oden (2000) and Huang and Russell (2011) for detailed expositions). The *h*-adaptivity starts with a relatively coarse initial mesh and then refines or coarsens the mesh successively based on a chosen error estimator, the sought-after adaptive method, if done correctly, should be able to automatically refine or coarsen the mesh where a refinement or coarsening should be done, so the method successively searches an "optimal" mesh and computes a satisfactory numerical solution on the mesh. The *h*-adaptivity often

goes together with finite element and finite difference methods, and the same finite element or finite difference method is often employed on each mesh successively generated by the adaptive algorithm. That is, the degree of the finite element space does not change in the algorithm. A typical h-adaptive algorithm consists of successive loops of the following sequence:

$$|SOLVE| \rightarrow |ESTIMATE| \rightarrow |MARK| \rightarrow |REFINE/COARSEN|$$

The hp-adaptivity is often associated with the hp-finite element method (Schwab, 1998), it only differs from the h-adaptivity in one aspect, that is, the refinement or coarsening is not only done for the mesh but also for the degree of finite element spaces. Hence, the finite element spaces may be different on the successive meshes generated by the adaptive method. The r-adaptivity, which is significantly different from the h- and hp-adaptivity, uses a fixed number of mesh points in the adaptive algorithm, however, the distribution of these mesh points is allowed to move, hence, a new mesh is generated by a new distribution of the mesh points, in particular, a locally fine mesh can be obtained by clustering a large (enough) number of the mesh points to where a fine mesh is desired. Clearly, a central issue for r-adaptive methods is how to develop an automatic strategy to move the mesh points. In Feng et al. (2006, 2009), Yu et al. (2008), and Shen and Yang (2009a), r-adaptivity was developed for spectral approximations of several phase field models. For more strategies on r-adaptivity, we refer to Huang and Russell (2011), Tan et al. (2006), Zhang and Tang (2007), Shen et al. (2011), Hu et al. (2009) and Di et al. (2008).

In the rest of this section, we shall only discuss h-adaptive finite element methods for phase field models. Indeed, there has been a lot of interest in spatial adaptive simulations for various phase field models (Braun and Murray, 1997; Ceniceros and Roma, 2007; Ceniceros et al., 2010; Du and Zhang, 2008; Joshi and Jaiman, 2018; Provatas et al., 2005; Rosam et al., 2008; Stogner et al., 2008).

We refer the reader to Ainsworth and Oden (2000), Shen et al. (2011) and Huang and Russell (2011) for detailed expositions for the other two spatial adaptive methodologies.

# 6.2 Coarse and fine a posteriori error estimates for phase field models

As mentioned above the key step for developing an h-adaptive method is to design a "good" error indicator which can successively predict where to refine and where to coarsen the mesh. As expected, such an indicator should be computable and must relate to the error of numerical solutions, so the mesh is refined where the error is large and coarsened where the error is too small. Error indicators of this kind is known as a posteriori error estimates in the literature.

The general form of a posteriori error estimates is given by

$$\|u^{\varepsilon} - u_{h,\tau}^{\varepsilon}\|_{\#} \le \mathcal{E}(\varepsilon, h, \tau, u_{h}^{\varepsilon})$$
 (72)

for some positive functional  $\mathcal{E}$ . Where  $\|\cdot\|_{\#}$  denotes a (function space) norm which is problem-dependent. Compare to the error estimate (62), there is one important difference, that is,  $\mathcal{E}(\varepsilon,h,\tau,u_h^{\varepsilon})$  is computable because it depends on the numerical solution  $u_h^{\varepsilon}$ , not the PDE solution  $u^{\varepsilon}$ ; on the other hand, the error bound in (62) is not computable because it depends on the unknown PDE solution  $u^{\varepsilon}$ . For this very reason, (62) and (72) are called, respectively, a priori and a posteriori error estimates.

The primary objective of a posteriori error estimates is to derive a tight error bound  $\mathcal{E}(\varepsilon, h, \tau, u_h^{\varepsilon})$  which is relatively easy to compute. Similar to the classification of a priori error estimates, we also divide a posteriori error estimates into two groups for phase field models. The first group of estimates, called coarse estimates, depends on  $\varepsilon^{-1}$  exponentially and the second group, called fine estimates, depends on  $\varepsilon^{-1}$  polynomially. Both groups of a posteriori error estimates have been developed for phase field models although the majority of them belong to the first group. Moreover, two main techniques have been used to derive the a posteriori error bound/functional  $\mathcal{E}(\varepsilon, h, \tau, u_h^{\varepsilon})$ . The first technique gives rise so-called residual-based error bounds (Binev et al., 2004; Eriksson et al., 1996; Morin et al., 2002; Nochetto et al., 2009), and the second one is a duality-based technique, called the Dual Weighted Residual method, which leads to goal-oriented error estimates (Bangerth and Rannacher, 2003). Below we shall only focus on discussing the residual-based a posteriori error estimation because of its popularity and superiority for phase field models.

The residual-based a posteriori error estimates give an error bound/functional  $\mathcal{E}(\varepsilon, h, \tau, u_h^{\varepsilon})$  which typically depends on local residuals of  $u_h^{\varepsilon}$  and the jumps of the flux  $\nabla u_h^{\varepsilon} \cdot n$  along each interior element edge/face. It turns out that residual-based error estimators are quite effective for phase field models, this is because the distinctive profiles of all phase field solutions, which are almost constant-valued in each bulk region and vary smoothly but sharply in the thin diffuse interface. We refer to Provatas and Elder (2011), Provatas et al. (1999), Du and Zhang (2008) and Elliott and Stinner (2010a) for detailed discussions about adaptive simulations of phase field models from materials science and biology applications. It should be noted that although coarse a posteriori error estimates were not rigorously derived in those cited works, they could be obtained by following the standard residual-based a posteriori estimate techniques (Ainsworth and Oden, 2000; Eriksson et al., 1996; Nochetto et al., 2009). Moreover, we note that an alternative technique based elliptic reconstruction was also developed for deriving a posteriori error estimates for parabolic PDEs including the Allen-Cahn equation (Georgoulis et al., 2011; Lakkis and Makridakis, 2006; Lakkis et al., 2015).

Like in the case of a priori error estimates for phase field models, the coarse a posteriori error bounds is not useful anymore if one is interested in knowing the limiting behaviour of the solution error and the error for numerical interfaces when  $\varepsilon \to 0$ . To this end, one must obtain fine a posteriori error estimates. As expected, the only known technique for deriving such error estimates is to utilize the PDE spectral estimate for the linearized operator of the underlying phase field PDE operator. Indeed, the desired fine a posteriori error estimates were obtained in Kessler et al. (2004) and Feng and Wu (2005) for the Allen–Cahn equation and in Feng and Wu (2008) for the Cahn-Hilliard equation. We note that the arguments and the usages of the PDE spectral estimate are quite different in Kessler et al. (2004) and in Feng and Wu (2005) and Feng and Wu (2008). A topological argument was used in Kessler et al. (2004), in which the PDE spectral estimate is embedded, to obtain a fine a posteriori error bound; while a sharp continuous dependence estimate for the PDE solution was used, in which the PDE spectral estimate was crucially utilized, to derive fine a posteriori error estimates in Feng and Wu (2005, 2008) (also see Cockburn, 2003). Because this continuous dependence argument is very simple and may be applicable to other evolution PDEs, we briefly explain it below.

Let V be a Hilbert space and  $\mathcal{L}$  be an operator from  $D(\mathcal{L})$  ( $\subset V$ ), the domain of  $\mathcal{L}$ , to  $V^*$ , the dual space of V. Consider the abstract evolution problem

$$\frac{\partial u}{\partial t} + \mathcal{L}(u) = g \quad \text{in } \Omega_T := \Omega \times (0, T), \tag{73}$$

$$u(0) = u_0 \quad \text{in } \Omega. \tag{74}$$

Assume that the above PDE problem satisfies the continuous dependence estimate in the sense that if  $u^{(j)}$  is the (unique) solution of (73) and (74) with respect to the data  $(g^{(j)}, u_0^{(j)})$  for j = 1, 2, then there holds

$$\|u^{(1)} - u^{(2)}\|_{L^p(0,T;V)} \le \Phi\left(g^{(1)} - g^{(2)}\right) + \Psi\left(u_0^{(1)} - u_0^{(2)}\right) \tag{75}$$

for some (monotone increasing) nonnegative functionals  $\Phi(\cdot)$  and  $\Psi(\cdot)$ . and  $1 \le p \le \infty$ . Let  $u^A$  be an approximation of u with the initial value  $u_0^A$ , it is easy to show that (Feng and Wu, 2005)

$$||u - u^{A}||_{L^{p}(0,T;V)} \le \Phi(R(u^{A})) + \Psi(u_{0} - u_{0}^{A}),$$
 (76)

where

$$R(u^A) := g - \frac{\partial u^A}{\partial t} - \mathcal{L}(u^A).$$

Clearly,  $R(u^A)$  denotes the residual of the approximation  $u^A$ . To utilize the above continuous dependence result to derive a posteriori error estimates for a spatial numerical discretization method, one only needs to set  $u^A = u_h$ , the space-semidiscrete numerical solution and to get an upper bound for the residual  $R(u_h)$  usually in terms of the element-wise local residuals and the jumps of the flux across the interior element edges/faces (cf. Feng and Wu, 2005, 2008). The biggest advantage of this approach for a posteriori error estimates is that it converts a numerical error estimate problem into a PDE continuous dependence (or stability) problem, which may be easier to cope with, especially, for deriving fine a posteriori error estimates for phase field models. In that case, the functionals  $\Phi(\cdot)$  and  $\Psi(\cdot)$  are expected to grow in  $\frac{1}{\epsilon}$  polynomially as shown in Feng and Wu (2005) and Feng and Wu (2008).

Finally, we point out that the spectral estimate idea has also been successfully used as an a posteriori indicator for detecting singularities (such as topological changes) of the underlying sharp interface limit of a phase field model in Antil and Bartels (2017), Bartels (2015), Bartels and Müller (2010, 2011a,b) and Bartels et al. (2011). The rationale for this approach is that if no singularity occurs, the discrete principal eigenvalue of the linearized operator, which is computable, should have a finite negative low bound. Hence, when such a low bound ceases to exist (i.e., it tends to  $-\infty$  faster than  $O(|\ln \varepsilon|)$  order as  $\varepsilon \to 0$ ), then it flags a possible singularity of the underlying moving interface problem, and suggests that a fine mesh should be used to resolve the singularity.

# **Applications and extensions**

The phase field (diffuse interface) method has many applications ranging from mathematical subjects like differential geometry, to image processing and geometric modelling, and to physical sciences like astrophysics, cell biology, multiphase fluid mechanics and (of course) materials science. We provide a sampler of some of the applications here.

### 7.1 Materials science applications

Applications to problems in materials science are among the early and most widely recognized successes of phase field models. From the works of Lord Rayleigh, Gibbs and Van de Walls to the seminal contributions of John W. Cahn, ideas of using diffuse interface and phase order parameters have drawn much interests in the materials science community in the mesoscopic modelling of materials structure. Generically, a set of conserved field variables  $c_1, c_2, ..., c_m$  and nonconserved field variables  $\eta_1, \eta_2, ..., \eta_n$  are often used to describe the compositional/structural domains and the interfaces, and the total free energy of an inhomogeneous microstructure system is formulated as

$$E_{total} = \int \left[ \sum_{i=1}^{m} \alpha_{i} (\nabla c_{i})^{2} + \sum_{i=1}^{3} \sum_{j=1}^{3} \sum_{k=1}^{n} \beta_{ij} \nabla_{i} \eta_{k} \nabla_{j} \eta_{k} \right.$$

$$\left. + f(c_{1}, c_{2}, ..., c_{m}, \eta_{1}, \eta_{2}, ..., \eta_{n}) \right] dx + \int \int G(x - x', \vec{c}, \vec{\eta}) dx dx',$$
(77)

where the gradient coefficient  $\alpha_i$  and  $\beta_{ij}$  can be used to reflect the interfacial energy anisotropy and the function f corresponds to the local free energy density. The last integral in the above equation represents a nonlocal term that includes a general long-range interaction such as elastic interactions in solids. The time evolution is governed by either the nonconserved (Allen–Cahn type for  $c_i$ 's) or conserved dynamics (Cahn–Hilliard type for  $\eta_i$ 's).

Phase field modelling is arguably one of the most popular methods for modelling and simulation of microstructure evolution under different driving forces such as compositional gradients, temperature, stress/strain and electric and magnetic fields (Biner, 2017). Phase field modelling has now been used to study not only binary phases and single components but also materials with multicomponents multiphases. It also can account for mechanical, thermodynamic, electric and magnetic interactions. It can deal with spinodal decomposition, dendritic growth, Ostwald ripening and coalescence, adhesion and dewetting, sintering, directional solidification, nucleation and coarsening, grain boundary motion, pattern formation in thin films, epitaxial growth, electromigration and many other processes. For additional references and more comprehensive reviews, we refer to Barrett et al. (2004, 2006, 2007, 2013, 2014b), Biner (2017), Baňas and Nürnberg (2008, 2009), Boettinger et al. (2002), Chen (2002), Dai and Du (2016), Provatas and Elder (2011), Shu et al. (2007), Garcke et al. (2004), Singer-Loginova and Singer (2008), Steinbach (2009) and Wang and Li (2010).

As an example, we mention one application on the study of nucleation based on the phase field approach (Zhang et al., 2016b). The search for index-1 saddle point of the phase field free energy (Heo et al., 2010; Zhang et al., 2007) can help characterize the transition states that lead to critical nuclei. For solid state phase transformation, anisotropic elastic energy plays a critical role, which leads to formation of nonspherical nuclei. A particular form of the total energy in the phase field set up is given by

$$E_{total} = \int (|\nabla \eta|^2 + f(\eta)) dx + \frac{1}{2} \int C_{ijkl} C_{ijkl} \varepsilon_{ij}^{el} \varepsilon_{kl}^{el} dx,$$

where the elastic strain  $\varepsilon^{el}$  is the difference between the total strain and stress-free strain since stress-free strain does not contribute to the total elastic energy. In Zhang et al. (2007, 2008), different critical nuclei have been computed as the driving forces change and the elastic effects vary. Numerical algorithms for index-1 saddle points are constructed by essentially reformulating the problem into a problem of optimization in an extended configuration space which includes not only the state variable (point on the energy landscape) but also the most unstable direction. In other words, one is effectively doing minimization along the stable directions (descending) while maximization along the (most) unstable direction (ascending) directions. This is the essential behind various saddle-point algorithms such as the Shrinking Dimer Methods (Zhang and Du, 2012) and other methods

reviewed in Zhang et al. (2016b). The predicted phase field nuclei can also be utilized in dynamic simulations and extended to investigate complex nucleation phenomena (Heo et al., 2010).

Let us also mention other variations of phase fields model related to materials science applications. For example, phase field modelling of polymeric materials has received much attention in theoretical chemistry and materials science. Phase field models of diblock copolymers, such as the model due to Ohta and Kawasaki (1986), have had a long history and is an active area of research (Ceniceros and Fredrickson, 2004; Cheng et al., 2017; Choksi and Ren, 2003; Li et al., 2013; Ren and Wei, 2007; Shi, 2019: Wanner, 2017).

In addition, as noted previously, there are many other extended phase field models but with vector or tensor valued fields as phase field variables. These models can describe physical phenomena and geometric objects with higher codimension such as point defects in two dimensions and curves in three dimensions.

### 7.2 Fluid and solid mechanics applications

Applications of phase field models and diffuse interface approaches to mechanical problems is also a major area of research.

Concerning fluid mechanics, a review was given in Anderson et al. (1998) on phase field/diffuse interface models of hydrodynamics and their application to a wide variety of interfacial phenomena involving fluid flows. Phase field models have been developed in various cases of fluid flows with a length scale commensurate with the diffuse interfacial width. Examples include small-scale flows and multiphase flows. The latter subject, typically involving breakup and coalescence such as fluid jets and droplets as well as fluid mixing and other interfacial phenomena, has become particularly important in microfluidics and micro-process engineering, see Prosperetti and Tryggvason (2009) and Wörner (2012) for reviews on the various numerical approaches. Phase field modelling of multiphase flows has been developed by a number of groups along with the development and numerical analysis, see, for instance (Abels et al., 2012; Badalassi et al., 2003; Baňas and Nürnberg, 2017; Diegel et al., 2017a; Feng et al., 2005a; Garcke et al., 2000; Grun, 2013; Liu and Shen, 2003b; Lowengrub and Truskinovsky, 1998) and the references cited therein. Various phase field based simulations of drop formation in microfluidic channels have been carried out, see for example, (Yue et al., 2004). Related optimal control problems and their simulations have been considered in Hintermüller and Wegner (2012) and Hintermüller et al. (2017). Three-dimensional phase field simulations of interfacial dynamics in Newtonian and viscoelastic fluids can be found in Zhou et al. (2010). Algorithm development and numerical analysis of three phase flows can be found in Yang et al. (2017).

Phase field modelling and simulation of multiphase flows with additional complications have also been made, studies include, for example, the wetting phenomena (Jacqmin, 2004), two-phase flow in porous media (Cogswell and Szulczewski, 2017). A problem of considerable interests is the moving contact line problem for the interaction of fluid-fluid interface with a solid wall is a widely studied subject, see Pomeau (2002) for a review. Ideas of diffuse interface and phase field models have also been considered (Jacqmin, 2000; Pismen, 2001). A phase field model with generalized Navier boundary condition (GNBC) is proposed in Qian et al. (2003). Further computational studies can be found in He et al. (2011), Bao et al. (2012), Shi et al. (2013), Gao and Wang (2014) and Zhang and Wang (2016).

Phase field modelling has also received much attention in solid mechanics. Representative works in this direction includes the modelling of brittle fracture (Borden et al., 2012; Conti et al., 2016; Miehe et al., 2010b). For example, in addition to the fracture energy of the type given by the Cahn-Hilliard functional as proposed in Bourdin et al. (2000), one can define the elastic energy density as

$$\psi_e(\varepsilon, u) = [(1-k)u^2 + k]\psi_e^+(\varepsilon)k\psi_e^-(\varepsilon),$$

where u is the phase field variable,  $\psi_e^\pm$  are the strain energies computed from the positive and negative components of the strain tensor  $\varepsilon^{\pm}$ , respectively, defined through a spectral decomposition of strain (Miehe et al., 2010a), that is,

$$\psi_e^+(\varepsilon) = \frac{1}{2}\lambda[(\mathrm{Tr}\varepsilon)^+]^2 + \mu\mathrm{Tr}[(\varepsilon^+)^2],$$

$$\psi_e^-(\varepsilon) = \frac{1}{2} \lambda [\mathrm{Tr} \varepsilon (\mathrm{Tr} \varepsilon)^+]^2 + \mu \mathrm{Tr} [(\varepsilon - \varepsilon^+)^2].$$

The phase field variable is only applied to the tensile part of the elastic energy density, so that crack propagation under compression can be prevented (Miehe et al., 2010a,b). One may further incorporate other features to model ductile fracture (Ambati et al., 2016; Miehe et al., 2016) and fluid-filled fracture (Wheeler et al., 2014), and utilize various discretization techniques, see recent discussions in Heister et al. (2015) and Zhang et al. (2018).

### Image and data processing applications 7.3

Ideas of phase field modelling are used not only for physical sciences but also for other domain sciences. Imaging science is a good example where variational methods have been widely used for imaging and data analysis (Aubert and Kornprobst, 2006; Chan and Shen, 2005). Phase field models and diffuse interface descriptions of geometric features form one major type of variational formulations for various image processing tasks. Taking, for example, the task

of image restoration and image deblurring, a diffuse interface relaxation to Mumford-Shah model can be formulated by Brett et al. (2014)

$$\mathcal{J}(u) := \int_{\Omega} \left( \frac{1}{2} |\nabla u|^2 + \frac{1}{\varepsilon^2} (u^2 - 1)^2 \right) dx + ||Su - f||^2, \tag{78}$$

where S can be a deblurring operator. Other examples include image classification and restoration (Samson et al., 2000), edge detection (Ambrosio and Tortorelli, 1990), motion estimation (Preusser et al., 2007), image segmentation (Burke et al., 2013; Jung et al., 2007; Li and Kim, 2011), curve smoothing (Zhu et al., 2010), active contour (Rochery et al., 2005), clustering of vector fields (Garcke et al., 2001) and so on.

For image segmentation, a phase field relaxation to Mumford-Shah can be formulated as (March and Dozio, 1997)

$$\mathcal{J}(v,u) := E(u) + \lambda \int_{\Omega} u^2 |\nabla v|^2 dx + ||v - f||^2, \tag{79}$$

where f is the image under consideration, v represents a piecewise smooth approximation to f, and E(u) is a diffuse interfacial energy of the phase field variable u.

While a standard Cahn-Hilliard energy is the most popular choice, a more general form has also been used as well:

$$E(u) := \alpha \int_{\Omega} \left( \frac{1}{2} |\nabla u|^2 + \frac{1}{\varepsilon^2} (u^2 - 1)^2 \right) dx + \beta \int_{\Omega} \left| \Delta u + \frac{1}{\varepsilon^2} (u - u^3) \right|^2 dx. \tag{80}$$

The first term in the above is the standard phase field form of the surface tension, the second term corresponds to a diffuse interface relaxation of the Euler-elastica (integral of the square of curvature along a planar curve) as discussed in De Giorgi (1991).

Discussions on issues related to the application of similar phase field energy to image inpainting can be found in Esedoglu and Shen (2002). More discussions on the three-dimensional analogue of the phase field Eulerelastica, a special form of the Helfrich bending energy, are given later in this chapter for phase field modelling of vesicle membrane.

Phase field approaches have also been used in data analysis and discrete graph modelling, see, for example, Bertozzi and Flenner (2012), Van Gennip et al. (2014) and Li and Kim (2011).

### **Biology applications** 7.4

In more recent years, phase field modelling and simulations have also become very successful tools to study problems arising in life sciences, including problems in biophysics, bioengineering, ecology, biology and so on. Some examples include the development of phase field models and computational

methods for the deformation and dynamics of bio-membranes and cell vesicles, cellular activities such as motility and cell division, fluid-structure interactions in blood flows and other biological fluids, and tumour adhesion and growth. These problems span across scales from cellular and molecular levels to macroscopic systems.

On the cellular level, activities such as cell crawling, cell motion on patterned substrate and other migration processes are also important biological questions. They often involve deformable geometry and associated mechanical interactions. These questions have been studied using various phase field approaches in recent years, see, for example, (Camley et al., 2013; Löber et al., 2014; Marth and Voigt, 2014; Moure and Gomez, 2017; Najem and Grant, 2016; Shao et al., 2012; Ziebert and Aranson, 2016). Phase field modelling has also found many applications in biological molecular processes such as conformational change, molecular recognition and molecular assembly. For example, Sun et al. (2015) presented phase field models for the implicit solvation of charged molecules with a coupling to Poisson-Boltzmann electrostatics. Various refined phase field models, numerical methods on both spatial discretization and time integration have also been studied (Dai et al., 2018; Li and Zhao, 2013; Zhao et al., 2018).

Bilayer vesicles are models systems for cell membranes. Their deformation has often been modelled by the Helfrich bending elasticity model for fluid membranes with contributions from the surface integrals of the mean curvature squared and the Gaussian curvature, thus leading to natural connections with the Willmore energy and Willmore flow. The diffuse interface formulation has been introduced in Du et al. (2004, 2006). Other variants have also been studied (Aland et al., 2014; Barrett et al., 2017; Biben et al., 2005; Bretin et al., 2015a; Campelo and Hernandez-Machado, 2007; Du et al., 2009; Esedoglu et al., 2014; Jamet and Misbah, 2007), together with analytical studies (Bellettini and Mugnai, 2010; Bretin et al., 2015b; Mugnai, 2013) and numerical approximations (Chen et al., 2015; Colli and Laurencot, 2011; Du and Zhang, 2008; Du and Zhu, 2006; Yang and Ju, 2017a) for the deformation and dynamics of vesicles. The effective phase field modelling of Gaussian curvature energy also leads to an interesting development of the diffuse interface Euler-Poincaré characteristics (Bellettini and Mugnai, 2010; Du et al., 2005) that can be used to detect topological change of the implicitly defined interface within the phase field framework beyond the biophysical applications. Phase field formulation has also been developed for vesicle-substrate and vesicle-vesicle interactions (Gu et al., 2016; Zhang et al., 2009). For multicomponent membranes, Elliott and Stinner (2010a) and Lowengrub et al. (2009b) studied phase field models defined on an evolving surface. Using two-phase field variables, Wang and Du (2008) developed phase field models for two-component membranes and obtained interesting patterns mimicking to experimental observations. Given the complexities involved in

these systems, effective numerical algorithms such as those utilizing high order spectral methods and adaptive discretizations (Du and Zhang, 2008; Du et al., 2004) and energy law preserving schemes (Hua et al., 2011) can be very useful for large scale and long time simulations.

Another example of broad applications of phase field modelling and simulations in biomedical field is on the cancer and tumour growth models. This is a subject that have been studied by a number of authors (Garcke et al., 2018; Lima et al., 2014; Vilanova et al., 2017), see Lowengrub et al. (2009a) for a review on the models and numerical methods.

### 7.5 Other variants of phase field models

There are many other variants of phase field models. We provide a couple of examples here that focus on the modelling of the nonlocal and stochastic aspects of the underlying physical processes.

#### 7.5.1 Nonlocal and factional order phase field models

Discussion on nonlocal interactions in the form of integral operators may be tracked back to the work of Van der Waals (1894), see discussions made in Pismen (2001). The usual differential equation form of the local phase field energy can be derived from the nonlocal version via the so-called Landau expansion (Landau and Lifshitz, 2013), assuming a smooth and spatially slowly varying field. A number of studies on nonlocal Allen-Cahn and nonlocal Cahn-Hilliard can be found in Bates et al. (2006), Bates (2006), Benesová et al. (2014), Choksi et al. (2009, 2011), Du et al. (2019), Fife (2003) and Jeong and Kim (2015). More studies on nonlocal modelling, analysis and computation can be found in Du (2019). A special class of nonlocal models are fractional phase field models where fractional derivatives are used to replace the integer order derivatives in the conventional local phase field models. Such models have attracted the attention of community in recent years (Akagi et al., 2016; Gui and Zhao, 2015; Milovanov and Rasmussen, 2005; Tarasov and Zaslavsky, 2005; Valdinoci, 2013). Algorithmic development and numerical analysis concerning these nonlocal models (including the fractional ones in either space or time or both ) can be found in Ainsworth and Mao (2017), Bates et al. (2009), Du and Yang (2016), Du et al. (2018b), Guan et al. (2014), Hou et al. (2017), Song et al. (2016) and Zhai et al. (2016). Related algorithmic studies with respect to different applications were presented in Antil and Bartels (2017).

For spatially nonlocal phase field models, due to the spread of nonlocal interactions, the phase field variables may no longer be as smooth as their local counterpart. They may also lead to narrower interfacial region and permit singularities across the interface or at the defects (Du and Yang, 2016; Gui and Zhao, 2015; Song et al., 2016).

### 7.5.2 Stochastic phase field models

Uncertainty may arise from various sources such as thermal fluctuation, impurities of the materials and the intrinsic instabilities of the deterministic evolutions. Therefore, the evolution of interfaces under influence of noise is of great importance in applications such as lattice models, the scaling limit of lattice models and derivations of continuum equations, it is necessary and interesting to consider stochastic effects, and to study the impact of noise on phase field modelling and on their solutions, especially on their long time behaviours. This then leads to considering the stochastic phase field models. However, how to incorporate noises correctly into those models is a nontrivial matter, which turns out to be both a science and an art.

A few approaches have been known in the literature. The first one, which is the simplest, is to add (small) noise terms to the existing deterministic models and to study and simulate those stochastically perturbed models. Some recent works on modelling and PDE analyses in this direction can be found in Weinan and Liu (2002), Weber (2010a,b), Hairer et al. (2012), Otto et al. (2014), Antonopoulou et al. (2018a,c), Debussche and Zambotti (2007), Debussche and Goudenège (2011), Da Prato and Debussche (1996) and Röckner et al. (2018) and the references therein. Finite element approximations have been studied in Kovács et al. (2011), Kovács et al. (2014) and Furihata et al. (2018) for the Cahn-Hilliard-Cook equation (which is a stochastic Cahn-Hilliard equation with additive noise) and in Kovács et al. (2015) and Feng et al. (2018) for stochastic Allen-Cahn models. Relevant time discretizations have been discussed in Printems (2001). Parallel algorithms and numerical simulations of the Cahn-Hilliard-Cook equation have also been reported in Zheng et al. (2015). Stochastic Cahn-Hilliard dynamics have also been used to study nucleation in microstructure evolution in Heo et al. (2010) and Li et al. (2012, 2013). It should be noted that the added noises may not always have physical meaning and those stochastically perturbed models may not be associate with sharp interface models.

An alternative approach is to directly consider stochastic sharp interface problems such as stochastic mean curvature flows and to derive (formally) corresponding phase field models, subsequently, to study and simulate the resulting stochastic models. Recent PDE studies in this direction can be found in Röger and Weber (2013), Dirr et al. (2001), Souganidis and Yip (2004) and Yip (1998, 2002), those stochastic PDEs involve gradient-type multiplicative noise and thus have stronger nonlinearity. Finite element approximations have been carried out in Feng et al. (2017) for a stochastic Allen-Cahn equation with gradient-type multiplicative noise and in Feng et al. (2019) for a related stochastic Cahn-Hilliard equation with gradient-type multiplicative noise. Not surprisingly, their sharp interface models are expected to be a stochastic mean curvature flow (Röger and Weber, 2013) and a stochastic Hele-Shaw model (Antonopoulou et al., 2018b; Feng et al., 2019), respectively. However, their rigorous convergence proofs are still missing although some partial results

were reported in Röger and Weber (2013) and Antonopoulou et al. (2018b) and positive numerical results were given in Feng et al. (2014, 2017, 2019).

An additional approach to model stochasticity is to study stochastic variants of coupled phase field equations with other physical models. Stochasticity can enter through other physical laws such as equations for fluid flows in multiphase flows and flow structure interactions (Chaudhri et al., 2014; Du and Li, 2011).

#### 8 Conclusion

In this chapter, we presented a holistic review about some basic elements of phase field modelling, particularly related to curvature-driven geometric interfacial motion. We discussed the relevant mathematical theory, numerical approximations and selected applications, as well as the relationship between the phase field methodology and other methodologies (such as the level set methodology) for geometric moving interface problems. Instead of presenting much involved technical details, we focused on discussing the main elements and ideas of the phase field modelling, analysis and their numerical approximations, with extensive references provided on each of these aspects. They should be helpful for the interested reader to do further reading for specific details.

Over last several decades, the phase field method has been developed into a powerful and versatile general methodology for interface problems arising from various scientific and engineering applications including biology, differential geometry, fluid and solid mechanics, image processing and materials science. It has garnered tremendous attention and popularity among researchers and practitioners, the trend will likely continue and make even broader impact in more and more fields. Yet, there are still many challenging issues in the phase field methodology, from modelling, analysis, approximation and application, that should be further investigated. On the modelling side, addressing different geometric features and connecting to microscopic physics are important future research topics. Effective recovery of geometric and topological features and statistical information from the phase field approach are of both theoretical and practical interests. On the computational side, there are much need and urgent demand for developing effective and robust adaptive algorithms, particularly those involving anisotropic spatial adaptivity and locally adapted time steps. Developing efficient and fast linear and nonlinear solvers, preconditioning techniques, high order and stable time-stepping schemes and scalable algorithms remains a focus of future research. Effective methods for computing saddle points or transition states, rather than equilibrium, are also very limited currently and are in high demand. Methods for characterizing the captured interfacial geometry and quantifying statistical features based on the phase field models are also helpful for many practical applications. On the analysis front, a lot of open questions remain to be answered. For the PDE analysis, convergence studies of many phase field models to their respective sharp interface limits are still missing. The lack of analysis techniques and machineries along with singularities of solutions of the sharp interface limits seem to be one of the primary hurdles to overcome. For the numerical analysis, analyzing the approximation errors and their dependencies on the diffuse interface parameter  $\varepsilon$ remains an interesting and challenging issue. Finally, while extreme scale simulations have been achieved for some model phase field equations, simulating more realistic and more complex phase field systems still require much future effort.

# Acknowledgements

The work of the first author was partially supported by the NSF grant DMS-1719699, and US Army Research Office MURI grant W911NF-15-1-0562. The work of the second author was partially supported by the NSF grants DMS-0410266 and DMS-1620168.

### References

- Abels, H., 2015. Sharp interface limit for the Cahn-Larché system. Asymptot. Anal. 91 (3-4), 283-340. ISSN 0921-7134.
- Abels, H., Lengeler, D., 2014. On sharp interface limits for diffuse interface models for twophase flows. Interfaces Free Bound. 16 (3), 395-418. ISSN 1463-9963. https://doi.org/ 10.4171/IFB/324.
- Abels, H., Liu, Y., 2018. Sharp interface limit for a Stokes/Allen-Cahn system. Arch. Ration. Mech. Anal. 229 (1), 417–502. ISSN 0003-9527. https://doi.org/10.1007/s00205-018-1220-x.
- Abels, H., Garcke, H., Grün, G., 2012. Thermodynamically consistent, frame indifferent diffuse interface models for incompressible two-phase flows with different densities. Math. Models Methods Appl. Sci. 22 (03), 1150013.
- Abels, H., Liu, Y., Schöttl, A., 2017. Sharp interface limits for diffuse interface models for twophase flows of viscous incompressible fluids. In: Transport processes at fluidic interfaces. Adv. Math. Fluid Mech, Birkhäuser/Springer, Cham, pp. 231–253.
- Ainsworth, M., Mao, Z., 2017. Analysis and approximation of a fractional Cahn-Hilliard equation. SIAM J. Numer. Anal. 55 (4), 1689-1718.
- Ainsworth, M., Oden, J.T., 2000. A Posteriori Error Estimation in Finite Element Analysis, Pure and Applied Mathematics (New York). Wiley-Interscience [John Wiley & Sons], New York, p. xx+240. ISBN 0-471-29411-X. https://doi.org/10.1002/9781118032824.
- Akagi, G., Schimperna, G., Segatti, A., 2016. Fractional Cahn-Hilliard, Allen-Cahn and porous medium equations. J. Differ. Equ. 261 (6), 2935-2985.
- Akrivis, G., Crouzeix, M., Makridakis, C., 1998. Implicit-explicit multistep finite element methods for nonlinear parabolic problems. Math. Comput. Am. Math. Soc. 67 (222), 457-477.
- Aland, S., Egerer, S., Lowengrub, J., Voigt, A., 2014. Diffuse interface models of locally inextensible vesicles in a viscous fluid. J. Comput. Phys. 277, 32-47.
- Alikakos, N.D., Fusco, G., 1993. The spectrum of the Cahn-Hilliard operator for generic interface in higher space dimensions. Indiana Univ. Math. J. 42 (2), 637-674. ISSN 0022-2518.
- Alikakos, N.D., Bates, P.W., Chen, X., 1994. Convergence of the Cahn-Hilliard equation to the Hele-Shaw model. Arch. Rational Mech. Anal. 128 (2), 165-205. ISSN 0945-8396.

- Allen, S., Cahn, J.W., 1979. A microscopic theory for antiphase boundary motion and its application to antiphase domain coarsening. Acta Metall. 27, 1084-1095.
- Ambati, M., Kruse, R., De Lorenzis, L., 2016. A phase-field model for ductile fracture at finite strains and its experimental verification. Comput. Mech. 57 (1), 149-167.
- Ambrosio, L., Tortorelli, V.M., 1990. Approximation of functionals depending on jumps by elliptic functionals via Γ-convergence. Comm. Pure Appl. Math. 43 (8), 999–1036. ISSN 0010-3640. https://doi.org/10.1002/cpa.3160430805.
- Anderson, D.M., McFadden, G.B., Wheeler, A.A., 1998. Diffuse-interface methods in fluid mechanics, In: Annual review of fluid mechanics, Annual Review of Fluid Mechanics, vol. 30. Annual Reviews, Palo Alto, CA, pp. 139-165.
- Antil, H., Bartels, S., 2017. Spectral approximation of fractional PDEs in image processing and phase field modeling. Comput. Methods Appl. Math. 17 (4), 661-678.
- Antonopoulou, D.C., Blömker, D., Karali, G.D., 2018a. The sharp interface limit for the stochastic Cahn-Hilliard equation. Ann. Inst. Henri Poincaré Probab. Stat. 54 (1), 280-298. ISSN 0246-0203. https://doi.org/10.1214/16-AIHP804.
- Antonopoulou, D.C., Blömker, D., Karali, G.D., 2018b. The sharp interface limit for the stochastic Cahn-Hilliard equation. Ann. Inst. Henri Poincaré Probab. Stat. 54 (1), 280-298. ISSN 0246-0203. https://doi.org/10.1214/16-AIHP804.
- Antonopoulou, D.C., Farazakis, D., Karali, G., 2018c. Malliavin calculus for the stochastic Cahn-Hilliard/Allen-Cahn equation with unbounded noise diffusion. J. Differ. Equ. 265 (7), 3168-3211. ISSN 0022-0396. https://doi.org/10.1016/j.jde.2018.05.004.
- Aristotelous, A.C., Karakashian, O.A., Wise, S.M., 2015. Adaptive, second-order in time, primitive-variable discontinuous Galerkin schemes for a Cahn-Hilliard equation with a mass source. IMA J. Numer. Anal. 35 (3), 1167-1198. ISSN 0272-4979. https://doi.org/10.1093/ imanum/dru035.
- Aubert, G., Kornprobst, P., 2006. Mathematical Problems in Image processing: Partial Differential Equations and the Calculus of Variations. vol. 147. Springer Science & Business Media.
- Badalassi, V.E., Ceniceros, H.D., Banerjee, S., 2003. Computation of multiphase systems with phase field models. J. Comput. Phys. 190 (2), 371-397.
- Badia, S., Guillén-González, F., Gutiérrez-Santacreu, J.V., 2011. Finite element approximation of nematic liquid crystal flows using a saddle-point structure. J. Comput. Phys. 230 (4), 1686–1706. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2010.11.033.
- Baňas, L., Nürnberg, R., 2008. Finite element approximation of a three dimensional phase field model for void electromigration. J. Sci. Comp. 37 (2), 202–232. https://doi.org/10.1007/ s10915-008-9203-y.
- Baňas, L., Nürnberg, R., 2009. Phase field computations for surface diffusion and void electromigration in  $\mathbb{R}^3$ . Comput. Vis. Sci. 12 (7), 319–327. ISSN 1432-9360. https://doi.org/10.1007/ s00791-008-0114-0.
- Baňas, L., Nürnberg, R., 2017. Numerical approximation of a non-smooth phase-field model for multicomponent incompressible flow. ESAIM: Math. Model. Numer. Anal. 51 (3), 1089–1117.
- Bangerth, W., Rannacher, R., 2003. Adaptive Finite Element Methods for Differential Equations, Lectures in Mathematics ETH Zürich. Birkhäuser Verlag, Basel, Basel. ISBN: 3-7643-7009-2, p. viii+207. https://doi.org/10.1007/978-3-0348-7605-6.
- Bao, K., Shi, Y., Sun, S., Wang, X.-P., 2012. A finite element method for the numerical solution of the coupled Cahn-Hilliard and Navier-Stokes system for moving contact line problems. J. Comput. Phys. 231 (24), 8083-8099. ISSN 0021-9991. https://doi.org/ 10.1016/j.jcp.2012.07.027.

- Barrett, J.W., Blowey, J.F., 1999. Finite element approximation of the Cahn-Hilliard equation with concentration dependent mobility. Math. Comp. 68 (226), 487-517. ISSN 0025-5718.
- Barrett, J.W., Blowey, J.F., Garcke, H., 1999. Finite element approximation of the Cahn-Hilliard equation with degenerate mobility. SIAM J. Numer. Anal. 37 (1), 286-318. ISSN 1095-7170
- Barrett, J.W., Nürnberg, R., Styles, V., 2004. Finite element approximation of a phase field model for void electromigration. SIAM J. Numer. Anal. 42 (2), 738-772. ISSN 0036-1429. https:// doi.org/10.1137/S0036142902413421.
- Barrett, J.W., Garcke, H., Nürnberg, R., 2006. Finite element approximation of a phase field model for surface diffusion of voids in a stressed solid. Math. Comp. 75 (253), 7-41. ISSN 0025-5718. https://doi.org/10.1090/S0025-5718-05-01802-8.
- Barrett, J.W., Garcke, H., Nürnberg, R., 2007. A phase field model for the electromigration of intergranular voids. Interfaces Free Bound. 9 (2), 171-210. ISSN 1463-9963. https://doi. org/10.4171/IFB/161.
- Barrett, J.W., Garcke, H., Nürnberg, R., 2013. On the stable discretization of strongly anisotropic phase field models with applications to crystal growth. ZAMM Z. Angew. Math. Mech. 93 (10–11), 719–732. https://doi.org/10.1002/zamm.201200147.
- Barrett, J.W., Garcke, H., Nürnberg, R., 2014a. Phase field models versus parametric front tracking methods: are they accurate and computationally efficient? Commun. Comput. Phys. 15 (2), 506-555. https://doi.org/10.4208/cicp.190313.010813a.
- Barrett, J.W., Garcke, H., Nürnberg, R., 2014b. Stable phase field approximations of anisotropic solidification. IMA J. Numer. Anal. 34 (4), 1289-1327. https://doi.org/10.1093/imanum/ drt044.
- Barrett, J.W., Garcke, H., Nürnberg, R., 2017. Finite element approximation for the dynamics of fluidic two-phase biomembranes. M2AN Math. Model. Numer. Anal. 51 (6), 2319-2366. https://doi.org/10.1051/m2an/2017037.
- Bartels, S., 2015. Robustness of error estimates for phase field models at a class of topological changes. Comput. Methods Appl. Mech. Eng. 288, 75-82. ISSN 0045-7825. https://doi.org/ 10.1016/j.cma.2014.11.005.
- Bartels, S., Müller, R., 2010. A posteriori error controlled local resolution of evolving interfaces for generalized Cahn-Hilliard equations, Interfaces Free Bound, 12 (1), 45-73, ISSN 1463-9963. https://doi.org/10.4171/IFB/226.
- Bartels, S., Müller, R., 2011a. Error control for the approximation of Allen-Cahn and Cahn-Hilliard equations with a logarithmic potential. Numer. Math. 119 (3), 409–435. ISSN 0029-599X. https://doi.org/10.1007/s00211-011-0389-9.
- Bartels, S., Müller, R., 2011b. Quasi-optimal and robust a posteriori error estimates in  $L^{\infty}(L^2)$  for the approximation of Allen-Cahn equations past singularities. Math. Comp. 80 (274), 761-780. ISSN 0025-5718. https://doi.org/10.1090/S0025-5718-2010-02444-5.
- Bartels, S., Müller, R., Ortner, C., 2011. Robust a priori and a posteriori error analysis for the approximation of Allen-Cahn and Ginzburg-Landau equations past topological changes. SIAM J. Numer. Anal. 49 (1), 110-134. ISSN 0036-1429. https://doi.org/10.1137/ 090751530.
- Bartezzaghi, A., Dedè, L., Quarteroni, A., 2015. Isogeometric analysis of high order partial differential equations on surfaces, Comput. Methods Appl. Mech. Eng. 295, 446-469. ISSN 0045-7825. https://doi.org/10.1016/j.cma.2015.07.018.
- Bates, P.W., 2006. On some nonlocal evolution equations arising in materials science. Nonlinear Dyn. Evol. Equ. 48, 13-52.

- Bates, P.W., Fife, P.C., 1990. Spectral comparison principles for the Cahn-Hilliard and phasefield equations, and time scales for coarsening. Physica D 43 (2-3), 335-348. ISSN 0167-2789.
- Bates, P.W., Han, J., Zhao, G., 2006. On a nonlocal phase-field system. Nonlinear Anal. Theory Methods Appl. 64 (10), 2251–2278.
- Bates, P.W., Brown, S., Han, J., 2009. Numerical analysis for a nonlocal Allen-Cahn equation. Int. J. Numer. Anal. Model 6 (1), 33-49.
- Bellettini, G., 2013. Lecture Notes on Mean Curvature Flow, Barriers and Singular Perturbations, Appunti. In: Scuola Normale Superiore di Pisa (Nuova Serie) [Lecture Notes. Scuola Normale Superiore di Pisa (New Series)], vol. 12. Edizioni della Normale, Pisa, p. xviii+325. https://doi.org/10.1007/978-88-7642-429-8. ISBN 978-88-7642-428-1; 978-88-7642-429-8.
- Bellettini, G., Mugnai, L., 2010. Approximation of Helfrich's functional via diffuse interfaces. SIAM J. Math. Anal. 42 (6), 2402-2433. ISSN 0036-1410. https://doi.org/10.1137/ 09077549X.
- Bellettini, G., Novaga, M., 1997. Minimal barriers for geometric evolutions. J. Differ. Equ. 139 (1), 76-103. ISSN 0022-0396.
- Benesová, B., Melcher, C., Süli, E., 2014. An implicit midpoint spectral approximation of nonlocal Cahn-Hilliard equations. SIAM J. Numer. Anal. 52 (3), 1466–1496.
- Bernardi, C., Maday, Y., 1997. Spectral methods. In: Handbook of Numerical Analysis, vol. 5. Elsevier, pp. 209-485.
- Bertozzi, A.L., Flenner, A., 2012. Diffuse interface models on graphs for classification of high dimensional data. Multiscale Model. Simul. 10 (3), 1090-1118.
- Biben, T., Kassner, K., Misbah, C., 2005. Phase-field approach to three-dimensional vesicle dynamics. Phys. Rev. E 72 (4), 041921.
- Biner, S.B., 2017. Programming Phase-Field Modeling. Springer.
- Binev, P., Dahmen, W., DeVore, R., 2004. Adaptive finite element methods with convergence rates. Numer. Math. 97 (2), 219-268. ISSN 0029-599X. https://doi.org/10.1007/s00211-003-0492-7.
- Boettinger, W.J., Warren, J.A., Beckermann, C., Karma, A., 2002. Phase-field simulation of solidification. Ann. Rev. Mater. Res. 32 (1), 163-194.
- Borden, M.J., Verhoosel, C.V., Scott, M.A., Hughes, T.J.R., Landis, C.M., 2012. A phase-field description of dynamic brittle fracture. Comput. Methods Appl. Mech. Eng. 217, 77-95.
- Borden, M.J., Hughes, T.J.R., Landis, C.M., Verhoosel, C.V., 2014. A higher-order phase-field model for brittle fracture: formulation and analysis within the isogeometric analysis framework. Comput. Methods Appl. Mech. Eng. 273, 100-118.
- Bourdin, B., Francfort, G.A., Marigo, J.-J., 2000. Numerical experiments in revisited brittle fracture. J. Mech. Phys. Solids 48 (4), 797-826.
- Boyd, J.P., 2001. Chebyshev and Fourier Spectral Methods. Courier Corporation.
- Brakke, K.A., 1978. The Motion of a Surface by its Mean Curvature. Mathematical Notes, vol. 20. Princeton University Press, Princeton, NJ. ISBN: 0-691-08204-9, p. i+252.
- Braun, R.J., Murray, B.T., 1997. Adaptive phase-field computations of dendritic crystal growth. J. Cryst. Growth 174 (1-4), 41-53.
- Bretin, E., Masnou, S., Oudet, E., 2014. Phase-field models for the approximation of the Willmore functional and flow, In: Congrès SMAI 2013, ESAIM Proc. Surveys, vol. 45. EDP Sci., Les Ulis, pp. 118–127.
- Bretin, E., Masnou, S., Oudet, É., 2015a. Phase-field approximations of the Willmore functional and flow. Numer. Math. 131 (1), 115-171.

- Bretin, E., Masnou, S., Oudet, E., 2015b. Phase-field approximations of the Willmore functional and flow. Numer. Math. 131 (1), 115-171. ISSN 0029-599X. https://doi.org/10.1007/s00211-014-0683-4.
- Brett, C., Elliott, C.M., Dedner, A.S., 2014. Phase field methods for binary recovery. In: Optimization With PDE Constraints, Springer, pp. 25–63.
- Bronsard, L., Kohn, R.V., 1991. Motion by mean curvature as the singular limit of Ginzburg-Landau dynamics. J. Differ. Equ. 90 (2), 211-237. ISSN 0022-0396. https://doi.org/ 10.1016/0022-0396(91)90147-2.
- Burke, S., Ortner, C., Süli, E., 2013. An adaptive finite element approximation of a generalized Ambrosio-Tortorelli functional. Math. Models Methods Appl. Sci. 23 (9), 1663-1697. ISSN 0218-2025. https://doi.org/10.1142/S021820251350019X.
- Caginalp, G., Chen, X., 1998. Convergence of the phase field model to its sharp interface limits. Eur. J. Appl. Math. 9 (4), 417-445. ISSN 0956-7925.
- Caginalp, G., Lin, J.T., 1987. A numerical analysis of an anisotropic phase field model. IMA J. Appl. Math. 39 (1), 51-66.
- Cahn, J.W., Hilliard, J.E., 1958. Free energy of a nonuniform system I. Interfacial free energy. J. Chem. Phys. 28, 258-267.
- Cahn, J.W., Elliott, C.M., Novick-Cohen, A., 1996. The Cahn-Hilliard equation with a concentration dependent mobility: motion by minus the Laplacian of the mean curvature. Eur. J. Appl. Math. 7 (3), 287-301. ISSN 0956-7925.
- Camley, B.A., Zhao, Y., Li, B., Levine, H., Rappel, W.-J., 2013. Periodic migration in a physical model of cells on micropatterns. Phys. Rev. Lett. 111 (15), 158102.
- Campelo, F., Hernandez-Machado, A., 2007. Shape instabilities in vesicles: a phase-field model. Eur. Phys. J. Spec. Top. 143 (1), 101-108.
- Canuto, C., Hussaini, M.Y., Quarteroni, A., Zang, T.A., 2007. Spectral Methods: Evolution to Complex Geometries and Applications to Fluid Dynamics. Springer Science & Business Media.
- Ceniceros, H.D., Fredrickson, G.H., 2004. Numerical solution of polymer self-consistent field theory. Multiscale Model. Simul. 2 (3), 452-474.
- Ceniceros, H.D., Roma, A.M., 2007. A nonstiff, adaptive mesh refinement-based method for the Cahn-Hilliard equation, J. Comput. Phys. 225 (2), 1849–1862.
- Ceniceros, H.D., Nós, R.L., Roma, A.M., 2010. Three-dimensional, fully adaptive simulations of phase-field fluid models. J. Comput. Phys. 229 (17), 6135-6155.
- Chan, T.F., Shen, J.J., 2005. Image Processing and Analysis: Variational, PDE, Wavelet, and Stochastic Methods, vol. 94, SIAM,
- Chaudhri, A., Bell, J.B., Garcia, A.L., Donev, A., 2014. Modeling multiphase flow using fluctuating hydrodynamics. Phys. Rev. E 90 (3), 033014.
- Chen, X., 1992. Generation and propagation of interfaces for reaction-diffusion equations. J. Differ. Equ. 96 (1), 116-141. ISSN 0022-0396.
- Chen, X., 1994. Spectrum for the Allen-Cahn, Cahn-Hilliard, and phase-field equations for generic interfaces. Comm. Partial Differ. Equ. 19 (7-8), 1371-1395. ISSN 0360-5302.
- Chen, X., 1996. Global asymptotic limit of solutions of the Cahn-Hilliard equation. J. Differ. Geom. 44 (2), 262-311. ISSN 0022-040X.
- Chen, L.-O., 2002. Phase-field models for microstructure evolution. Ann. Rev. Mater. Res. 32 (1), 113-140.
- Chen, L.O., Shen, J., 1998. Applications of semi-implicit Fourier-spectral method to phase field equations. Comput. Phys. Commun. 108 (2-3), 147-158.

- Chen, Y., Shen, J., 2016. Efficient, adaptive energy stable schemes for the incompressible Cahn-Hilliard-Navier-Stokes phase-field models. J. Comput. Phys. 308, 40-56.
- Chen, Y.G., Giga, Y., Goto, S., 1991. Uniqueness and existence of viscosity solutions of generalized mean curvature flow equations. J. Differ. Geom. 33 (3), 749-786. ISSN 0022-040X.
- Chen, X., Elliott, C.M., Gardiner, A., Zhao, J., 1998. Convergence of numerical solutions to the Allen-Cahn equation. Appl. Anal. 69 (1–2), 47–56.
- Chen, Z., Nochetto, R.H., Schmidt, A., 2000. Error control and adaptivity for a phase relaxation model. M2AN Math. Model. Numer. Anal. 34 (4), 775-797. ISSN 0764-583X.
- Chen, X., Wang, X., Xu, X., 2014. Analysis of the Cahn-Hilliard equation with a relaxation boundary condition modeling the contact angle dynamics. Arch. Ration. Mech. Anal. 213 (1), 1-24. ISSN 0003-9527. https://doi.org/10.1007/s00205-013-0713-x.
- Chen, R., Ji, G., Yang, X., Zhang, H., 2015. Decoupled energy stable schemes for phase-field vesicle membrane model. J. Comput. Phys. 302, 509-523.
- Chen, W., Han, D., Wang, X., 2017. Uniquely solvable and energy stable decoupled numerical schemes for the Cahn-Hilliard-Stokes-Darcy system for two-phase flows in karstic geometry. Numer. Math. 137 (1), 229-255. ISSN 0029-599X. https://doi.org/10.1007/ s00211-017-0870-1.
- Cheng, O., Shen, J., 2018. Multiple scalar auxiliary variable (MSAV) approach and its application to the phase-field vesicle membrane model. SIAM J. Sci. Comput. 40 (6), A3982-A4006. ISSN 1064-8275. https://doi.org/10.1137/18M1166961.
- Cheng, K., Wang, C., Wise, S.M., Yue, X., 2016. A second-order, weakly energy-stable pseudospectral scheme for the Cahn-Hilliard equation and its solution by the homogeneous linear iteration method. J. Sci. Comput. 69 (3), 1083-1114.
- Cheng, Q., Yang, X., Shen, J., 2017. Efficient and accurate numerical schemes for a hydrodynamically coupled phase field diblock copolymer model. J. Comput. Phys. 341, 44-60.
- Cheng, K., Feng, W., Wang, C., Wise, S.M., 2019. An energy stable fourth order finite difference scheme for the Cahn-Hlliard equation. J. Comput. Appl. Math. 362, 574–595.
- Choksi, R., Ren, X., 2003. On the derivation of a density functional theory for microphase separation of diblock copolymers. J. Statist. Phys. 113 (1-2), 151-176.
- Choksi, R., Peletier, M.A., Williams, J.F., 2009. On the phase diagram for microphase separation of diblock copolymers: an approach via a nonlocal Cahn-Hilliard functional. SIAM J. Appl. Math. 69 (6), 1712-1738.
- Choksi, R., Maras, M., Williams, J.F., 2011. 2D phase diagram for minimizers of a Cahn-Hilliard functional with long-range interactions. SIAM J. Appl. Dyn. Syst. 10 (4), 1344–1362.
- Chopp, D.L., Sethian, J.A., 1999. Motion by intrinsic Laplacian of curvature. Interfaces Free Bound. 1 (1), 107-123. ISSN 1463-9963.
- Chrysafinos, K., 2019. Stability analysis and best approximation error estimates of discontinuous time-stepping schemes for the Allen-Cahn equation. ESAIM M2AN 53, 551-583.
- Ciarlet, P.G., 1978. The Finite Element Method for Elliptic Problems. vol. 4. North-Holland Publishing Co., Amsterdam. ISBN: 0-444-85028-7, p. xix+530
- Cockburn, B., 2003. Continuous dependence and error estimation for viscosity methods. Acta Numer. 12, 127–180. ISSN 0962-4929. https://doi.org/10.1017/S0962492902000107.
- Cogswell, D.A., Szulczewski, M.L., 2017. Simulation of incompressible two-phase flow in porous media with large timesteps. J. Comput. Phys. 345, 856-865.
- Colli, P., Laurençot, P., 2011. A phase-field approximation of the Willmore flow with volume constraint. Interfaces Free Bound. 13 (3), 341-351. ISSN 1463-9963. https://doi.org/ 10.4171/IFB/261.

- Conti, S., Focardi, M., Iurlano, F., 2016. Phase field approximation of cohesive fracture models. Ann. Inst. Henri Poincar (C) Non-Linaire Analyze 33 (4), 1033–1067.
- Copetti, M.I.M., Elliott, C.M., 1992. Numerical analysis of the Cahn-Hilliard equation with a logarithmic free energy. Numer. Math. 63 (1), 39-65. ISSN 0029-599X.
- Cox, S.M., Matthews, P.C., 2002. Exponential time differencing for stiff systems. J. Comput. Phys. 176 (2), 430–455.
- Da Prato, G., Debussche, A., 1996. Stochastic Cahn-Hilliard equation. Nonlinear Anal. 26 (2), 241–263. ISSN 0362-546X. https://doi.org/10.1016/0362-546X(94)00277-O.
- Dai, S., Du, Q., 2012. Motion of interfaces governed by the Cahn-Hilliard equation with highly disparate diffusion mobility. SIAM J. Appl. Math. 72 (6), 1818-1841. ISSN 0036-1399. https://doi.org/10.1137/120862582.
- Dai, S., Du, Q., 2014. Coarsening mechanism for systems governed by the Cahn-Hilliard equation with degenerate diffusion mobility. Multiscale Model. Simul. 12 (4), 1870-1889. ISSN 1540-3459. https://doi.org/10.1137/140952387.
- Dai, S., Du, Q., 2016. Computational studies of coarsening rates for the Cahn-Hilliard equation with phase-dependent diffusion mobility. J. Comput. Phys. 310, 85-108.
- Dai, S., Promislow, K., 2013. Geometric evolution of bilayers under the functionalized Cahn-Hilliard equation. Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci. 469 (2153), 2012505. ISSN 1364-5021. https://doi.org/10.1098/rspa.2012.0505.
- Dai, S., Li, B., Lu, J., 2018. Convergence of phase-field free energy and boundary force for molecular solvation. Arch. Ration. Mech. Anal. 227 (1), 105-147. ISSN 0003-9527. https:// doi.org/10.1007/s00205-017-1158-4.
- Davis, T.A., Gartland Jr., E.C., 1998. Finite element analysis of the Landau-de Gennes minimization problem for liquid crystals. SIAM J. Numer. Anal. 35 (1), 336-362.
- De Giorgi, E., 1991. Some remarks on y-convergence and least squares method. In: Dal Maso, G., Dell'Antonio, G.F. (Eds.), Composite media and homogenization theory, Progress in Nonlinear Differential Equations and Their Applications, vol. 5. Birkhäuser, Boston, pp. 135-142.
- De Giorgi, E., 1994. Barriers, boundaries, motion of manifolds. Conference held at Department of Mathematics of Pavia, March 18, 1994.
- De Mottoni, P., Schatzman, M., 1995. Geometrical evolution of developed interfaces. Trans. Amer. Math. Soc. 347 (5), 1533-1589. ISSN 0002-9947.
- Dean, E.J., Glowinski, R., Trevas, D.A., 1996. An approximate factorization/least squares solution method for a mixed finite element approximation of the Cahn-Hilliard equation. Jpn J. Ind. Appl. Math. 13 (3), 495-517. ISSN 0916-7005. https://doi.org/10.1007/ BF03167260.
- Debussche, A., Goudenège, L., 2011. Stochastic Cahn-Hilliard equation with double singular nonlinearities and two reflections. SIAM J. Math. Anal. 43 (3), 1473-1494. ISSN 0036-1410. https://doi.org/10.1137/090769636.
- Debussche, A., Zambotti, L., 2007. Conservative stochastic Cahn-Hilliard equation with reflection. Ann. Probab. 35 (5), 1706-1739. ISSN 0091-1798. https://doi.org/10.1214/ 009117906000000773.
- Dedè, L., Quarteroni, A., 2018. Isogeometric analysis of a phase field model for Darcy flows with discontinuous data. Chin. Ann. Math. Ser. B 39 (3), 487-512. ISSN 0252-9599. https://doi. org/10.1007/s11401-018-0079-3.
- Dedè, L., Borden, M.J., Hughes, T.J.R., 2012. Isogeometric analysis for topology optimization with a phase field model. Arch. Comput. Methods Eng. 19 (3), 427-465. ISSN 1134-3060. https://doi.org/10.1007/s11831-012-9075-z.

- Di, Y., Li, R., Tang, T., 2008. A general moving mesh framework in 3D and its application for simulating the mixture of multi-phase flows. Commun. Comput. Phys. 3 (3), 582-602. ISSN 1815-2406.
- Diegel, A.E., Feng, X.H., Wise, S.M., 2015. Analysis of a mixed finite element method for a Cahn-Hilliard-Darcy-Stokes system. SIAM J. Numer. Anal. 53 (1), 127-152. ISSN 0036-1429. https://doi.org/10.1137/130950628.
- Diegel, A.E., Wang, C., Wang, X., Wise, S.M., 2017a. Convergence analysis and error estimates for a second order accurate finite element method for the Cahn-Hilliard-Navier-Stokes system. Numer. Math. 137 (3), 495-534.
- Dirr, N., Luckhaus, S., Novaga, M., 2001. A stochastic selection principle in case of fattening for curvature flow. Calc. Var. Partial Differ. Equ. 13 (4), 405-425. ISSN 0944-2669. https://doi. org/10.1007/s005260100080.
- Dondl, P.W., Lemenant, A., Wojtowytsch, S., 2017. Phase field models for thin elastic structures with topological constraint. Arch. Ration. Mech. Anal. 223 (2), 693-736. ISSN 0003-9527. https://doi.org/10.1007/s00205-016-1043-6.
- Du, Q., 2005. Numerical approximations of the Ginzburg-Landau models for superconductivity. J. Math. Phys. 46 (9), 095109.
- Du, Q., 2011. Phase field calculus, curvature-dependent energies, and vesicle membranes. Philos. Mag. 91 (1), 165–181.
- Du, Q., 2019. Nonlocal Modeling, Analysis and Computation. CBMS-NSF regional conference series in applied mathematics, vol. 94. SIAM.
- Du, Q., Li, M., 2011. Analysis of a stochastic implicit interface model for an immersed elastic surface in a fluctuating fluid. Arch. Ration. Mech. Anal. 199 (1), 329-352.
- Du, Q., Nicolaides, R.A., 1991. Numerical analysis of a continuum model of phase transition. SIAM J. Numer. Anal. 28 (5), 1310-1322. ISSN 0036-1429.
- Du, Q., Yang, J., 2016. Asymptotically compatible Fourier spectral approximations of nonlocal Allen-Cahn equations. SIAM J. Numer. Anal. 54 (3), 1899-1919.
- Du, Q., Yang, J., 2017. Fast and accurate implementation of Fourier spectral approximations of nonlocal diffusion operators and its applications. J. Comput. Phys. 332, 118-134. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2016.11.028.
- Du, Q., Zhang, J., 2008. Adaptive finite element method for a phase field bending elasticity model of vesicle membrane deformations. SIAM J. Sci. Comput. 30 (3), 1634–1657.
- Du, Q., Zhu, W., 2004. Stability analysis and applications of the exponential time differencing schemes and their contour integration modifications. J. Comput. Math. 22, 200-209.
- Du, Q., Zhu, W., 2005. Analysis and applications of the exponential time differencing schemes. BIT Numer. Math. 45, 307-328.
- Du, Q., Zhu, L., 2006. Analysis of a mixed finite element method for a phase field bending elasticity model of vesicle membrane deformation. J. Comput. Math. 24, 265-280.
- Du, Q., Gunzburger, M.D., Peterson, J.S., 1992. Analysis and approximation of the Ginzburg-Landau model of superconductivity. SIAM Rev. 34 (1), 54-81.
- Du, Q., Liu, C., Wang, X., 2004. A phase field approach in the numerical study of the elastic bending energy for vesicle membranes. J. Comput. Phys. 198 (2), 450-468.
- Du, Q., Liu, C., Wang, X., 2005. Retrieving topological information for phase field models. SIAM J. Appl. Math. 65 (6), 1913–1932.
- Du, O., Liu, C., Wang, X., 2006. Simulating the deformation of vesicle membranes under elastic bending energy in three dimensions. J. Comput. Phys. 212 (2), 757–777.
- Du, Q., Liu, C., Ryham, R., Wang, X., 2007. Diffuse interface energies capturing the Euler number: relaxation and renormalization. Commun. Math. Sci. 5 (1), 233-242. ISSN 1539-6746. http:/projecteuclid.org/euclid.cms/1175797629.

- Du, Q., Liu, C., Ryham, R., Wang, X., 2009. Energetic variational approaches in modeling vesicle and fluid interactions. Physica D 238 (9-10), 923-930.
- Du, Q., Ju, L., Tian, L., 2011. Finite element approximation of the Cahn-Hilliard equation on surfaces. Comput. Methods Appl. Mech. Eng. 200 (29-32), 2458-2470.
- Du, O., Ju, L., Li, X., Qiao, Z., 2018a. Maximum principles of semilinear parabolic equations and their exponential time differencing approximations. Preprint arXiv:1902.04998.
- Du, Q., Ju, L., Li, X., Qiao, Z., 2018b. Stabilized linear semi-implicit schemes for the nonlocal Cahn-Hilliard equation. J. Comput. Phys. 363, 39-54.
- Du, Q., Ju, L., Li, X., Qiao, Z., 2019. Maximum principle preserving exponential time differencing schemes for the nonlocal Allen-Cahn equation. SIAM J. Numer. Anal. 57, 875-898.
- Ecker, K., 2004. Regularity Theory for Mean Curvature Flow, Progress in Nonlinear Differential Equations and their Applications. 57, Birkhauser Boston, Inc., Boston, MA. ISBN: 0-8176-3243-3, p. xiv+165. https://doi.org/10.1007/978-0-8176-8210-1.
- Edwards, D.A., Brenner, H., Wasan, D.T., 1991. Interfacial Transport Processes and Rheology. Butterworth-Heinemann, Boston.
- Elliott, C.M., French, D.A., 1987. Numerical studies of the Cahn-Hilliard equation for phase separation. IMA J. Appl. Math. 38 (2), 97-128. ISSN 0272-4960.
- Elliott, C.M., French, D.A., 1989. A nonconforming finite-element method for the two-dimensional Cahn-Hilliard equation. SIAM J. Numer. Anal. 26 (4), 884-903. ISSN 0036-1429.
- Elliott, C.M., Ranner, T., 2015. Evolving surface finite element method for the Cahn-Hilliard equation. Numer. Math. 129 (3), 483-534.
- Elliott, C.M., Stinner, B., 2010a. A surface phase field model for two-phase biological membranes. SIAM J. Appl. Math. 70 (8), 2904-2928.
- Elliott, C.M., Stuart, A.M., 1993. The global dynamics of discrete semilinear parabolic equations. SIAM J. Numer. Anal. 30 (6), 1622-1663.
- Eriksson, K., Estep, D., Hansbo, P., Johnson, C., 1996. Computational Differential Equations. Cambridge University Press, Cambridge. ISBN 0-521-56312-7; 0-521-56738-6, p. xvi+538.
- Escher, J., Mayer, U.F., Simonett, G., 1998. The surface diffusion flow for immersed hypersurfaces. SIAM J. Math. Anal. 29 (6), 1419-1433. ISSN 1095-7154 (electronic).
- Esedoglu, S., Shen, J., 2002. Digital inpainting based on the Mumford-Shah-Euler image model. Eur. J. Appl. Math. 13 (4), 353–370.
- Esedoglu, S., Rätz, A., Röger, M., 2014. Colliding interfaces in old and new diffuse-interface approximations of Willmore-flow. Commun. Math. Sci. 12 (1), 125-147.
- Evans, L.C., Spruck, J., 1991. Motion of level sets by mean curvature. I. J. Differ. Geom. 33 (3), 635-681. ISSN 0022-040X.
- Evans, L.C., Soner, H.M., Souganidis, P.E., 1992. Phase transitions and generalized motion by mean curvature. Comm. Pure Appl. Math. 45 (9), 1097-1123. ISSN 0010-3640.
- Eyre, D.J., 1998. Unconditionally gradient stable time marching the Cahn-Hilliard equation. MRS Online Proc. Libr. Arch. 529.
- Feng, X., 2006. Fully discrete finite element approximations of the Navier-Stokes-Cahn-Hilliard diffuse interface model for two-phase fluid flows. SIAM J. Numer. Anal. 44 (3), 1049-1072. ISSN 0036-1429. https://doi.org/10.1137/050638333.
- Feng, X., Karakashian, O.A., 2007. Fully discrete dynamic mesh discontinuous Galerkin methods for the Cahn-Hilliard equation of phase transition, Math. Comp. 76 (259), 1093-1117. ISSN 0025-5718. https://doi.org/10.1090/S0025-5718-07-01985-0.
- Feng, X., Li, Y., 2015. Analysis of symmetric interior penalty discontinuous Galerkin methods for the Allen-Cahn equation and the mean curvature flow. IMA J. Numer. Anal. 35 (4), 1622-1651. ISSN 0272-4979. https://doi.org/10.1093/imanum/dru058.

- Feng, X., Prohl, A., 2003a. Numerical analysis of the Allen-Cahn equation and approximation for mean curvature flows. Numer. Math. 94 (1), 33-65. ISSN 0029-599X. https://doi.org/ 10.1007/s00211-002-0413-1.
- Feng, X., Prohl, A., 2004a. Analysis of a fully discrete finite element method for the phase field model and approximation of its sharp interface limits. Math. Comp. 73 (246), 541–567. ISSN 0025-5718. https://doi.org/10.1090/S0025-5718-03-01588-6.
- Feng, X., Prohl, A., 2004c. Error analysis of a mixed finite element method for the Cahn-Hilliard equation. Numer. Math. 99 (1), 47-84. ISSN 0029-599X. https://doi.org/10.1007/ s00211-004-0546-5.
- Feng, X., Prohl, A., 2005. Numerical analysis of the Cahn-Hilliard equation and approximation of the Hele-Shaw problem. Interfaces Free Bound. 7 (1), 1–28. ISSN 1463-9963. https://doi.org/ 10.4171/IFB/111.
- Feng, X., Wise, S., 2012. Analysis of a Darcy-Cahn-Hilliard diffuse interface model for the Hele-Shaw flow and its fully discrete finite element approximation. SIAM J. Numer. Anal. 50 (3), 1320–1343. ISSN 0036-1429. https://doi.org/10.1137/110827119.
- Feng, X., Wu, H., 2005. A posteriori error estimates and an adaptive finite element method for the Allen-Cahn equation and the mean curvature flow. J. Sci. Comput. 24 (2), 121-146. ISSN 0885-7474. https://doi.org/10.1007/s10915-004-4610-1.
- Feng, X., Wu, H., 2008. A posteriori error estimates for finite element approximations of the Cahn-Hilliard equation and the Hele-Shaw flow. J. Comput. Math. 26 (6), 767-796. ISSN 0254-9409.
- Feng, J.J., Liu, C., Shen, J., Yue, P., 2005a. An energetic variational formulation with phase field methods for interfacial dynamics of complex fluids: advantages and challenges. Modeling of Soft Matter, Springer, pp. 1-26.
- Feng, W.M., Yu, P., Hu, S.Y., Liu, Z.-K., Du, Q., Chen, L.-Q., 2006. Spectral implementation of an adaptive moving mesh method for phase-field equations. J. Comput. Phys. 220 (1),
- Feng, X., He, Y., Liu, C., 2007a. Analysis of finite element approximations of a phase field model for two-phase fluids. Math. Comp. 76 (258), 539-571. ISSN 0025-5718. https://doi.org/ 10.1090/S0025-5718-06-01915-6.
- Feng, W.M., Yu, P., Hu, S.Y., Liu, Z.-K., Du, Q., Chen, L.Q., 2009. A Fourier spectral moving mesh method for the Cahn-Hilliard equation with elasticity. Commun. Comput. Phys.
- Feng, X., Li, Y., Prohl, A., 2014. Finite element approximations of the stochastic mean curvature flow of planar curves of graphs. Stoch. Partial Differ. Equ. Anal. Comput. 2 (1), 54-83. ISSN 2194-0401. https://doi.org/10.1007/s40072-014-0025-7.
- Feng, X., Tang, T., Yang, J., 2015. Long time numerical simulations for phase-field problems using p-adaptive spectral deferred correction methods. SIAM J. Sci. Comput. 37 (1), A271-A294. ISSN 1064-8275. https://doi.org/10.1137/130928662.
- Feng, X., Li, Y., Xing, Y., 2016. Analysis of mixed interior penalty discontinuous Galerkin methods for the Cahn-Hilliard equation and the Hele-Shaw flow. SIAM J. Numer. Anal. 54 (2), 825-847. ISSN 0036-1429. https://doi.org/10.1137/15M1009962.
- Feng, X., Li, Y., Zhang, Y., 2017. Finite element methods for the stochastic Allen-Cahn equation with gradient-type multiplicative noise, SIAM J. Numer, Anal. 55 (1), 194-216. ISSN 0036-1429. https://doi.org/10.1137/15M1022124.
- Feng, X., Li, Y., Zhang, Y., 2018. Strong convergence of a fully discrete finite element method for a class of semilinear stochastic partial differential equations with multiplicative noise. arXiv:1811.05028 [math.NA] (under review).

- Feng, X., Li, Y., Zhang, Y., 2019. Fully discrete finite element methods for a stochastic Cahn-Hilliard equation with gradient-type multiplicative noise arXiv:1903.05146 [math. NA] (under review).
- Fife, P.C., 1988. Dynamics of internal layers and diffusive interfaces. CBMS-NSF Regional Conference Series in Applied Mathematics, vol. 53. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA. ISBN: 0-89871-225-4, p. vi+93.
- Fife, P., 2003. Some nonclassical trends in parabolic and parabolic-like evolutions. In: Trends in Nonlinear AnalysisSpringer, pp. 153–191.
- Fix, G., 1983. Phase field method for free boundary problems. In: Fasano, A., Primicerio, M. (Eds.), Free Boundary Problems. Pitman, London, pp. 580-589.
- Furihata, D., 2001. A stable and conservative finite difference scheme for the Cahn-Hilliard equation, Numer, Math. 87 (4), 675-699.
- Furihata, D., Kovács, M., Larsson, S., Lindgren, F., 2018. Strong convergence of a fully discrete finite element approximation of the stochastic Cahn-Hilliard equation. SIAM J. Numer. Anal. 56 (2), 708-731. ISSN 0036-1429. https://doi.org/10.1137/17M1121627.
- Gao, M., Wang, X.-P., 2014. An efficient scheme for a phase field model for the moving contact line problem with variable density and viscosity. J. Comput. Phys. 272, 704-718. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2014.04.054.
- Garcke, H., Novick-Cohen, A., 2000. A singular limit for a system of degenerate Cahn-Hilliard equations. Adv. Differ. Equ. 5 (4-6), 401-434. ISSN 1079-9389.
- Garcke, H., Nestler, B., Stoth, B., 2000. A multiphase field concept: numerical simulations of moving phase boundaries and multiple junctions. SIAM J. Appl. Math. 60 (1), 295-315. ISSN 0036-1399. https://doi.org/10.1137/S0036139998334895.
- Garcke, H., Preusser, T., Rumpf, M., Telea, A.C., Weikard, U., Van Wijk, J.J., 2001. A phase field model for continuous clustering on vector fields. IEEE Trans. Vis. Comput. Graph. 7 (3), 230-241.
- Garcke, H., Nestler, B., Stinner, B., 2004. A diffuse interface model for alloys with multiple components and phases. SIAM J. Appl. Math. 64 (3), 775-799. ISSN 0036-1399. https://doi.org/ 10.1137/S0036139902413143.
- Garcke, H., Lam, K.F., Nürnberg, R., Sitka, E., 2018. A multiphase Cahn-Hilliard-Darcy model for tumour growth with necrosis, Math. Models Methods Appl. Sci. 28 (03), 525–577.
- Georgoulis, E.H., Lakkis, O., Virtanen, J.M., 2011. A posteriori error control for discontinuous Galerkin methods for parabolic problems. SIAM J. Numer. Anal. 49 (2), 427–458. ISSN 0036-1429. https://doi.org/10.1137/080722461.
- Giga, Y., 2006. Surface Evolution Equations. Monographs in Mathematics, vol. 99. Birkhäuser Verlag, Basel. ISBN 978-3-7643-2430-8; 3-7643-2430-9, p. xii+264.
- Gomez, H., Hughes, T.J.R., 2011. Provably unconditionally stable, second-order time-accurate, mixed variational methods for phase-field models. J. Comput. Phys. 230 (13), 5310-5327.
- Gomez, H., 2018. Computational phase-field modeling. In: Encyclopedia of Computational Mechanics, second ed. Wiley Online Library, pp. 1-35.
- Gómez, H., Calo, V.M., Bazilevs, Y., Hughes, T.J.R., 2008. Isogeometric analysis of the Cahn-Hilliard phase-field model. Comput. Methods Appl. Mech. Eng. 197 (49–50), 4333–4352.
- Gottlieb, D., Orszag, S.A., 1983. Numerical Analysis of Spectral Methods: Theory and Applica-
- Grun, G., 2013. On convergent schemes for diffuse interface models for two-phase flow of incompressible fluids with general mass densities. SIAM J. Numer. Anal. 51 (6), 3036–3061.
- Gu, R., Wang, X., Gunzburger, M., 2016. A two phase field model for tracking vesicle-vesicle adhesion. J. Math. Biol. 73 (5), 1293-1319.

- Guan, Z., Lowengrub, J.S., Wang, C., Wise, S.M., 2014. Second order convex splitting schemes for periodic nonlocal Cahn-Hilliard and Allen-Cahn equations. J. Comput. Phys. 277, 48-71.
- Gui, C., Zhao, M., 2015. Traveling wave solutions of Allen-Cahn equation with a fractional Laplacian. Ann. Inst. Henri Poincar (C) Non-Linaire Analyze 32 (4), 785-812.
- Guillén-González, F., Tierra, G., 2013. On linear schemes for a Cahn-Hilliard diffuse interface model. J. Comput. Phys. 234, 140-171. ISSN 0021-9991. https://doi.org/10.1016/ j.jcp.2012.09.020.
- Guo, B.-Y., 1998. Spectral Methods and Their Applications. World Scientific.
- Guo, R., Xu, Y., 2018. A high order adaptive time-stepping strategy and local discontinuous Galerkin method for the modified phase field crystal equation. Commun. Comput. Phys. 24 (1), 123-151. ISSN 1815-2406.
- Guo, R., Ji, L., Xu, Y., 2016a. High order local discontinuous Galerkin methods for the Allen-Cahn equation: analysis and simulation. J. Comput. Math. 34 (2), 135-158. ISSN 0254-9409. https://doi.org/10.4208/jcm.1510-m2014-0002.
- Guo, R., Ji, L., Xu, Y., 2016b. High order local discontinuous Galerkin methods for the Allen-Cahn equation: analysis and simulation. J. Comput. Math. 34 (2), 135-158. ISSN 0254-9409. https://doi.org/10.4208/jcm.1510-m2014-0002.
- Hairer, M., Ryser, M.D., Weber, H., 2012. Triviality of the 2D stochastic Allen-Cahn equation. Electron. J. Probab. 17 (39), 14. ISSN 1083-6489. https://doi.org/10.1214/EJP.v17-1731.
- Han, D., Wang, X., 2018. A second order in time, decoupled, unconditionally stable numerical scheme for the Cahn-Hilliard-Darcy system. J. Sci. Comput. 77 (2), 1210-1233. ISSN 0885-7474. https://doi.org/10.1007/s10915-018-0748-0.
- He, Q., Glowinski, R., Wang, X.-P., 2011. A least-squares/finite element method for the numerical solution of the Navier-Stokes-Cahn-Hilliard system modeling the motion of the contact line. J. Comput. Phys. 230 (12), 4991-5009. ISSN 0021-9991. https://doi.org/10.1016/ j.jcp.2011.03.022.
- Heister, T., Wheeler, M.F., Wick, T., 2015. A primal-dual active set method and predictorcorrector mesh adaptivity for computing fracture propagation using a phase-field approach. Comput. Methods Appl. Mech. Eng. 290, 466-495.
- Hele-Shaw, H.S., 1898b. On the motion of a viscous fluid between two parallel plates. Trans. R. Inst. Nav. Archit. Lond. 40, 218.
- Heo, T.W., Zhang, L., Du, Q., Chen, L.-Q., 2010. Incorporating diffuse-interface nuclei in phasefield simulations. Scr. Mater. 63 (1), 8-11.
- Hintermüller, M., Wegner, D., 2012. Distributed optimal control of the Cahn-Hilliard system including the case of a double-obstacle homogeneous free energy density, SIAM J. Control. Optim. 50 (1), 388-418.
- Hintermüller, M., Hinze, M., Kahle, C., Keil, T., 2017. Fully adaptive and integrated numerical methods for the simulation and control of variable density multiphase flows governed by diffuse interface models. Transport Processes at Fluidic Interfaces, Springer, pp. 305-353.
- Hochbruck, M., Ostermann, A., 2010. Exponential integrators. Acta Numer. 19, 209-286.
- Hou, T., Tang, T., Yang, J., 2017. Numerical analysis of fully discretized Crank-Nicolson scheme for fractional-in-space Allen-Cahn equations. J. Sci. Comput. 72 (3), 1214-1231.
- Hu, S.Y., Chen, L.Q., 2001. A phase-field model for evolving microstructures with strong elastic inhomogeneity. Acta Mater. 49 (11), 1879-1890.
- Hu, J., Zhang, S., 2017. A canonical construction of  $H^m$ -nonconforming triangular finite elements. Ann. Appl. Math. 33 (3), 266–288. ISSN 2096-0174.
- Hu, X., Li, R., Tang, T., 2009. A multi-mesh adaptive finite element approximation to phase field models. Commun. Comput. Phys. 5 (5), 1012-1029. ISSN 1815-2406.

- Hua, J., Lin, P., Liu, C., Wang, Q., 2011. Energy law preserving C<sup>0</sup> finite element schemes for phase field models in two-phase flow computations. J. Comput. Phys. 230 (19), 7115-7131.
- Huang, W., Russell, R.D., 2011. Adaptive Moving Mesh Methods, Applied Mathematical Sciences. vol. 174, Springer, New York. ISBN: 978-1-4419-7915-5, p. xviii+432. https:// doi.org/10.1007/978-1-4419-7916-2.
- Huisken, G., Ilmanen, T., 2001. The inverse mean curvature flow and the Riemannian Penrose inequality. J. Differ. Geom. 59 (3), 353-437. ISSN 0022-040X.
- Humphries, A.R., Stuart, A.M., 1994. Runge-Kutta methods for dissipative and gradient dynamical systems. SIAM J. Numer. Anal. 31 (5), 1452-1485.
- Ilmanen, T., 1993. Convergence of the Allen-Cahn equation to Brakke's motion by mean curvature. J. Differ. Geom. 38 (2), 417-461. ISSN 0022-040X. http://projecteuclid.org/euclid.jdg/ 1214454300.
- Iserles, A., 2009. A First Course in the Numerical Analysis of Differential Equations. In: Cambridge Texts in Applied Mathematics, second ed. Cambridge University Press, Cambridge. ISBN: 978-0-521-73490-5, p. xx+459.
- Jacqmin, D., 1999. Calculation of two-phase Navier-Stokes flows using phasefield modeling. J. Comput. Phys. 155 (1), 96–127. ISSN 0021-9991.
- Jacqmin, D., 2000. Contact-line dynamics of a diffuse fluid interface. J. Fluid Mech. 402, 57–88. Jacqmin, D., 2004. Onset of wetting failure in liquid-liquid systems. J. Fluid Mech. 517, 209-228.
- Jamet, D., Misbah, C., 2007. Towards a thermodynamically consistent picture of the phase-field model of vesicles: local membrane incompressibility. Phys. Rev. E 76 (5), 051907.
- Jeong, D., Kim, J., 2015. Microphase separation patterns in diblock copolymers on curved surfaces using a nonlocal Cahn-Hilliard equation. Eur. Phys. J. E 38 (11), 117.
- Joshi, V., Jaiman, R.K., 2018. An adaptive variational procedure for the conservative and positivity preserving Allen-Cahn phase-field model. J. Comput. Phys. 366, 478-504. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2018.04.022.
- Ju, L., Zhang, J., Du, Q., 2015a. Fast and accurate algorithms for simulating coarsening dynamics of Cahn-Hilliard equations. Comput. Mater. Sci. 108, 272-282.
- Ju, L., Zhang, J., Zhu, L., Du, Q., 2015b. Fast explicit integration factor methods for semilinear parabolic equations. J. Sci. Comput. 62 (2), 431–455.
- Jung, Y.M., Kang, S.H., Shen, J., 2007. Multiphase image segmentation via Modica-Mortola phase transition. SIAM J. Appl. Math. 67 (5), 1213-1232.
- Kassam, A.-K., Trefethen, L.N., 2005. Fourth-order time-stepping for stiff PDEs. SIAM J. Sci. Comput. 26 (4), 1214-1233.
- Kästner, M., Metsch, P., de Borst, R., 2016. Isogeometric analysis of the Cahn-Hilliard equation a convergence study. J. Comput. Phys. 305, 360-371. ISSN 0021-9991. https://doi.org/ 10.1016/j.jcp.2015.10.047.
- Kay, D., Styles, V., Süli, E., 2009. Discontinuous Galerkin finite element approximation of the Cahn-Hilliard equation with convection. SIAM J. Numer. Anal. 47 (4), 2660-2685. ISSN 0036-1429. https://doi.org/10.1137/080726768.
- Kessler, D., Nochetto, R.H., Schmidt, A., 2004. A posteriori error control for the Allen-Cahn problem: circumventing Gronwall's inequality. M2AN Math. Model. Numer. Anal. 38 (1), 129-142. ISSN 0764-583X. https://doi.org/10.1051/m2an:2004006.
- Kim, J., Lowengrub, J., 2005. Phase field modeling and simulation of three-phase flows. Interfaces Free Bound. 7 (4), 435-466.
- Kim, J., Kang, K., Lowengrub, J., 2004. Conservative multigrid methods for Cahn-Hilliard fluids. J. Comput. Phys. 193 (2), 511-543.

- Kohn, R.V., Sternberg, P., 1989. Local minimisers and singular perturbations. Proc. R. Soc. Edinburgh Sect. A 111 (1-2), 69-84. ISSN 0308-2105. https://doi.org/10.1017/ S0308210500025026.
- Kovács, M., Larsson, S., Mesforush, A., 2011. Finite element approximation of the Cahn-Hilliard-Cook equation. SIAM J. Numer. Anal. 49 (6), 2407–2429.
- Kovács, M., Larsson, S., Mesforush, A., 2014. Erratum: Finite element approximation of the Cahn-Hilliard-Cook equation [mr2854602]. SIAM J. Numer. Anal. 52 (5), 2594-2597. ISSN 0036-1429. https://doi.org/10.1137/140968161.
- Kovács, M., Larsson, S., Lindgren, F., 2015. On the backward Euler approximation of the stochastic Allen-Cahn equation. J. Appl. Probab. 52 (2), 323-338. ISSN 0021-9002. https://doi.org/ 10.1239/jap/1437658601.
- Lakkis, O., Makridakis, C., 2006. Elliptic reconstruction and a posteriori error estimates for fully discrete linear parabolic problems. Math. Comp. 75 (256), 1627-1658. ISSN 0025-5718. https://doi.org/10.1090/S0025-5718-06-01858-8.
- Lakkis, O., Makridakis, C., Pryer, T., 2015. A comparison of duality and energy a posteriori estimates for  $l_{\infty}(0,t;l_2(\omega))$  in parabolic problems. Math. Comp. 84 (294), 1537–1569. ISSN 0025-5718. https://doi.org/10.1090/S0025-5718-2014-02912-8.
- Lambert, J.D., 1991. Numerical Methods for Ordinary Differential Systems. John Wiley & Sons, Ltd., Chichester. ISBN: 0-471-92990-5, p. x+293
- Landau, L.D., Lifshitz, E.M., 2013. Course of Theoretical Physics. Elsevier.
- Langer, J.S., 1986. Models of pattern formation in first-order phase transitions. In: Directions in Condensed Matter Physics, World Science Publishers, pp. 164–186.
- Lee, A.A., Münch, A., Süli, E., 2016. Sharp-interface limits of the Cahn-Hilliard equation with degenerate mobility. SIAM J. Appl. Math. 76 (2), 433-456. ISSN 0036-1399. https://doi. org/10.1137/140960189.
- Li, Y., 2019. Error analysis of a fully discrete Morley finite element approximation for the Cahn-Hilliard equation. J. Sci. Comput. 78, 1862–1892.
- Li, Y., Kim, J., 2011. Multiphase image segmentation using a phase-field model. Comput. Math. Appl. 62 (2), 737–745.
- Li, D., Qiao, Z., 2017. On second order semi-implicit Fourier spectral methods for 2D Cahn-Hilliard equations. J. Sci. Comput. 70 (1), 301-341.
- Li, B., Zhao, Y., 2013. Variational implicit solvation with solute molecular mechanics: from diffuse-interface to sharp-interface models. SIAM J. Appl. Math. 73 (1), 1-23. ISSN 0036-1399. https://doi.org/10.1137/120883426.
- Li, B., Lowengrub, J., Rätz, A., Voigt, A., 2009. Geometric evolution laws for thin crystalline films: modeling and numerics. Commun. Comput. Phys. 6 (3), 433-482. ISSN 1815-2406.
- Li, T., Zhang, P., Zhang, W., 2012. Numerical study for the nucleation of one-dimensional stochastic Cahn-Hilliard dynamics. Commun. Math. Sci. 10 (4), 1105-1132.
- Li, T., Zhang, P., Zhang, W., 2013. Nucleation rate calculation for the phase transition of diblock copolymers under stochastic Cahn-Hilliard dynamics. Multiscale Model. Simul. 11 (1), 385-409. ISSN 1540-3459. https://doi.org/10.1137/120876307.
- Li, D., Qiao, Z., Tang, T., 2016a. Characterizing the stabilization size for semi-implicit Fourier-spectral method to phase field equations, SIAM J. Numer. Anal. 54 (3), 1653-1681.
- Li, D., Qiao, Z., Tang, T., 2016b. Characterizing the stabilization size for semi-implicit Fourierspectral method to phase field equations. SIAM J. Numer. Anal. 54 (3), 1653-1681. ISSN 0036-1429. https://doi.org/10.1137/140993193.

- Li, Y., Lee, H.G., Xia, B., Kim, J., 2016c. A compact fourth-order finite difference scheme for the three-dimensional Cahn-Hilliard equation. Comput. Phys. Commun. 200, 108-116.
- Li, Y., Choi, Y., Kim, J., 2017. Computationally efficient adaptive time step method for the Cahn-Hilliard equation. Comput. Math. Appl. 73 (8), 1855-1864.
- Lima, E.A.B.F., Oden, J.T., Almeida, R.C., 2014. A hybrid ten-species phase-field model of tumor growth. Math. Models Methods Appl. Sci. 24 (13), 2569–2599.
- Lin, J.-T., 1988. The numerical analysis of a phase field model in moving boundary problems. SIAM J. Numer. Anal. 25 (5), 1015-1031.
- Liu, C., Shen, J., 2003a. A phase field model for the mixture of two incompressible fluids and its approximation by a Fourier-spectral method. Physica D 179, 211-228.
- Liu, C., Shen, J., 2003b. A phase field model for the mixture of two incompressible fluids and its approximation by a Fourier-spectral method. Physica D 179 (3-4), 211-228.
- Löber, J., Ziebert, F., Aranson, I.S., 2014. Modeling crawling cell movement on soft engineered substrates. Soft Matter 10 (9), 1365-1373.
- Lowengrub, J., Truskinovsky, L., 1998. Quasi-incompressible Cahn-Hilliard fluids and topological transitions. Proc. R. Soc. Lond. A Math. Phys. Eng. Sci. 454 (1978), 2617-2654.
- Lowengrub, J.S., Frieboes, H.B., Jin, F., Chuang, Y.-L., Li, X., Macklin, P., Wise, S.M., Cristini, V., 2009a. Nonlinear modelling of cancer: bridging the gap between cells and tumours. Nonlinearity 23 (1), R1.
- Lowengrub, J.S., Rätz, A., Voigt, A., 2009b. Phase-field modeling of the dynamics of multicomponent vesicles: spinodal decomposition, coarsening, budding, and fission. Phys. Rev. E 79(3), 0311926, 13, ISSN 1539-3755. https://doi.org/10.1103/PhysRevE.79.031926.
- Luo, F., Tang, T., Xie, H., 2016. Parameter-free time adaptivity based on energy evolution for the Cahn-Hilliard equation. Commun. Comput. Phys. 19 (5), 1542–1563.
- March, R., Dozio, M., 1997. A variational method for the recovery of smooth boundaries. Image Vis. Comput. 15 (9), 705-712.
- Marques, F.C., Neves, A., 2014. Min-max theory and the Willmore conjecture. Ann. Math. 179, 683-782.
- Marth, W., Voigt, A., 2014. Signaling networks and cell motility: a computational approach using a phase field description. J. Math. Biol. 69 (1), 91-112.
- McFadden, G.B., 2002. Phase field models of solidification. In: Feng, X., Schulze, T.P. (Eds.), Recent Advances in Numerical Methods for Partial Differential Equations and Applications, Contemporary Mathematics, vol. 306. American Mathematical Society, Providence, RI, pp. 107-145.
- Miehe, C., Hofacker, M., Welschinger, F., 2010a. A phase field model for rate-independent crack propagation: robust algorithmic implementation based on operator splits. Comput. Methods Appl. Mech. Eng. 199 (45-48), 2765-2778.
- Miehe, C., Welschinger, F., Hofacker, M., 2010b. Thermodynamically consistent phase-field models of fracture: variational principles and multi-field finite element implementations. Int. J. Numer. Methods Eng. 83 (10), 1273-1311.
- Miehe, C., Aldakheel, F., Raina, A., 2016. Phase field modeling of ductile fracture at finite strains: a variational gradient-extended plasticity-damage theory. Int. J. Plast. 84, 1–32.
- Milovanov, A.V., Rasmussen, J.J., 2005. Fractional generalization of the Ginzburg-Landau equation: an unconventional approach to critical phenomena in complex media. Phys. Lett. A 337 (1-2), 75-80.
- Morin, P., Nochetto, R.H., Siebert, K.G., 2002. Convergence of adaptive finite element methods. SIAM Rev. 44 (4), 631-658.
- Moure, A., Gomez, H., 2017. Phase-field model of cellular migration: three-dimensional simulations in fibrous networks. Comput. Methods Appl. Mech. Eng. 320, 162–197.

- Moure, A., Gomez, H., 2018. Three-dimensional simulation of obstacle-mediated chemotaxis. Biomech. Model. Mechanobiol. 17 (5), 12431268.
- Mugnai, L., 2013. Gamma-convergence results for phase-field approximations of the 2D-Euler elastica functional. ESAIM Control Optim. Calc. Var. 19 (3), 740-753. ISSN 1292-8119. https://doi.org/10.1051/cocv/2012031.
- Mullins, W.W., Sekerka, J., 1963. Morphological stability of a particle growing by diffusion or heat flow. J. Appl. Math. 34, 322-329.
- Najem, S., Grant, M., 2016. Phase-field model for collective cell migration. Phys. Rev. E 93 (5), 052405.
- Nochetto, R.H., Verdi, C., 1997. Convergence past singularities for a fully discrete approximation of curvature-driven interfaces. SIAM J. Numer. Anal. 34 (2), 490-512. ISSN 0036-1429.
- Nochetto, R.H., Paolini, M., Verdi, C., 1994. Optimal interface error estimates for the mean curvature flow. Ann. Scuola Norm. Sup. Pisa Cl. Sci. 21 (2), 193-212.
- Nochetto, R.H., Siebert, K.G., Veeser, A., 2009. Theory of adaptive finite element methods: an introduction. In: Multiscale, Nonlinear and Adaptive Approximation, Springer, Berlin, pp. 409–542.
- Novick-Cohen, A., 2000. Triple-junction motion for an Allen-Cahn/Cahn-Hilliard system. Physica D 137 (1-2), 1-24. ISSN 0167-2789. https://doi.org/10.1016/S0167-2789(99)00162-1.
- Ohta, T., Kawasaki, K., 1986. Equilibrium morphology of block copolymer melts. Macromolecules 19 (10), 2621-2632.
- Osher, S., Fedkiw, R., 2003. Level Set Methods and Dynamic Implicit Surfaces. Applied Mathematical Sciences, vol. 153. Springer-Verlag, New York. ISBN: 0-387-95482-1, p. xiv+273.
- Osher, S., Sethian, J.A., 1988. Fronts propagating with curvature-dependent speed: algorithms based on Hamilton-Jacobi formulations. J. Comput. Phys. 79 (1), 12-49. ISSN 0021-9991.
- Otto, F., Weber, H., Westdickenberg, M.G., 2014. Invariant measure of the stochastic Allen-Cahn equation: the regime of small noise and large system size. Electron. J. Probab. 19, no. 23, 76. ISSN 1083-6489. https://doi.org/10.1214/EJP.v19-2813.
- Pego, R.L., 1989. Front migration in the nonlinear Cahn-Hilliard equation. Proc. Roy. Soc. London Ser. A 422 (1863), 261-278. ISSN 0962-8444.
- Pismen, L.M., 2001. Nonlocal diffuse interface theory of thin films and the moving contact line. Phys. Rev. E 64 (2), 021603.
- Plapp, M., Karma, A., 2000. Multiscale finite-difference-diffusion-Monte-Carlo method for simulating dendritic solidification. J. Comput. Phys. 165 (2), 592-619. ISSN 0021-9991. https:// doi.org/10.1006/jcph.2000.6634.
- Pomeau, Y., 2002. Recent progress in the moving contact line problem: a review. C. R. Mecanique 330 (3), 207-222.
- Preusser, T., Droske, M., Garbe, C.S., Telea, A., Rumpf, M., 2007. A phase field method for joint denoising, edge detection, and motion estimation in image sequence processing. SIAM J. Appl. Math. 68 (3), 599-618.
- Priestly, E., 2012. Introduction to Liquid Crystals. Springer Science & Business Media.
- Printems, J., 2001. On the discretization in time of parabolic stochastic partial differential equations. ESAIM: Math. Model. Numer. Anal. 35 (6), 1055-1078.
- Prosperetti, A., Tryggvason, G., 2009. Computational Methods for Multiphase Flow. Cambridge University Press.
- Provatas, N., Elder, K., 2011. Phase-Field Methods in Materials Science and Engineering. John Wiley & Sons.
- Provatas, N., Goldenfeld, N., Dantzig, J., 1999. Adaptive mesh refinement computation of solidification microstructures using dynamic data structures. J. Comput. Phys. 148 (1), 265-290. ISSN 0021-9991. https://doi.org/10.1006/jcph.1998.6122.

- Provatas, N., Greenwood, M., Athreya, B., Goldenfeld, N., Dantzig, J., 2005. Multiscale modeling of solidification: phase-field methods to adaptive mesh refinement. Int. J. Mod. Phys. B 19 (31), 4525-4565.
- Qian, T., Wang, X.-P., Sheng, P., 2003. Molecular scale contact line hydrodynamics of immiscible flows. Phys. Rev. E 68 (1), 016306.
- Rayleigh, J.W., 1899. Scientific Papers (Vol. 1). University Press.
- Reddy, S.C., Trefethen, L.N., 1992. Stability of the method of lines. Numer. Math. 62, 235-267.
- Ren, X., Wei, J., 2007. Many droplet pattern in the cylindrical phase of diblock copolymer morphology. Rev. Math. Phys. 19 (08), 879-921.
- Rochery, M., Jermyn, I., Zerubia, J., 2005. Phase field models and higher-order active contours. In: Tenth IEEE International Conference on Computer Vision, 2005. ICCV 2005, vol. 2. IEEE, pp. 970-976.
- Röckner, M., Yang, H., Zhu, R., 2018. Conservative stochastic 2-dimensional Cahn-Hilliard equation. arXiv:1802.04141 [math.NA].
- Röger, M., Weber, H., 2013. Tightness for a stochastic Allen-Cahn equation. Stoch. Partial Differ. Equ. Anal. Comput. 1 (1), 175-203. ISSN 2194-0401. https://doi.org/10.1007/s40072-013-0004-4.
- Rogers, T.M., Elder, K.R., Desai, R.C., 1988. Numerical study of the late stages of spinodal decomposition. Phys. Rev. B 37 (16), 9638.
- Rosam, J., Jimack, P.K., Mullis, A.M., 2008. An adaptive, fully implicit multigrid phase-field model for the quantitative simulation of non-isothermal binary alloy solidification. Acta Mater. 56 (17), 4559-4569.
- Rowlinson, J.S., 1979. Translation of J. D. van der Waals' "The thermodynamic theory of capillarity under the hypothesis of a continuous variation of density" J. Statist. Phys. 20 (2), 197-244. ISSN 0022-4715. https://doi.org/10.1007/BF01011513.
- Rubinstein, J., Sternberg, P., Keller, J.B., 1989. Fast reaction, slow diffusion, and curve shortening. SIAM J. Appl. Math. 49 (1), 116-133. ISSN 0036-1399.
- Samson, C., Blanc-Féraud, L., Aubert, G., Zerubia, J., 2000. A variational model for image classification and restoration. IEEE Trans. Pattern Anal. Mach. Intell. 22 (5), 460-472.
- Schiesser, W.E., 1991. The Numerical Method of Lines: Integration of Partial Differential Equations. Academic Press, San Diego.
- Schillinger, D., Borden, M.J., Stolarski, H.K., 2015. Isogeometric collocation for phase-field fracture models. Comput. Methods Appl. Mech. Eng. 284, 583-610.
- Schwab, C., 1998. p- and hp-Finite Element Methods. Numerical Mathematics and Scientific Computation. The Clarendon Press, Oxford University Press, New York. ISBN: 0-19-850390-3, p. xii +374.
- Sethian, J.A., 1999. Level Set Methods and Fast Marching Methods. In: Cambridge Monographs on Applied and Computational Mathematics, second ed. vol. 3. Cambridge University Press. Cambridge, p. xx+378. ISBN 0-521-64204-3; 0-521-64557-3.
- Shao, D., Levine, H., Rappel, W.-J., 2012. Coupling actin flow, adhesion, and morphology in a computational cell motility model. Proc. Natl. Acad. Sci. 109 (18), 6851-6856.
- Shen, J., Xu, J., 2018. Convergence and error analysis for the scalar auxiliary variable (SAV) schemes to gradient flows. SIAM J. Numer, Anal. 56 (5), 2895-2912. ISSN 0036-1429. https://doi.org/10.1137/17M1159968.
- Shen, J., Yang, X., 2009a. An efficient moving mesh spectral method for the phase-field model of two-phase flows. J. Comput. Phys. 228 (8), 2978-2992.
- Shen, J., Yang, X., 2010. Numerical approximations of Allen-Cahn and Cahn-Hilliard equations. Discrete Contin. Dyn. Syst. 28 (4), 1669-1691.

- Shen, J., Yang, X., 2019. Decoupled, energy stable schemes for phase-field models of two-phase incompressible flows. SIAM Rev. 61, 474-506.
- Shen, J., Xu, J., Yang, J., 2018. A new class of efficient and robust energy stable schemes for gradient flows. Preprint (to appear in SIAM Rev).
- Shen, J., Tang, T., Wang, L.-L., 2011. Spectral Methods: Algorithms, Analysis and Applications. vol. 41. Springer Science & Business Media.
- Shen, J., Xu, J., Yang, J., 2018. The scalar auxiliary variable (SAV) approach for gradient flows. J. Comput. Phys. 353, 407–416. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2017.10.021.
- Shi, A.-C., 2019. Self-consistent field theory of inhomogeneous polymeric systems: a variational derivation. Adv. Theory Simul. 2, 1800188.
- Shi, Y., Bao, K., Wang, X.-P., 2013. 3D adaptive finite element method for a phase field model for the moving contact line problems. Inverse Probl. Imaging 7 (3), 947-959. ISSN 1930-8337. https://doi.org/10.3934/ipi.2013.7.947.
- Shin, J., Lee, H.G., Lee, J.-Y., 2017. Unconditionally stable methods for gradient flow using convex splitting Runge-Kutta scheme. J. Comput. Phys. 347, 367-381.
- Shu, Y., Ai, X., Li, B.Q., 2007. Phase-field modelling of solidification microstructure formation using the discontinuous finite element method. Int. J. Numer. Methods Eng. 69 (6), 1194–1211. ISSN 0029-5981. https://doi.org/10.1002/nme.1805.
- Simonett, G., 2001. The Willmore flow near spheres. Differ. Integral Equ. 14 (8), 1005–1014. ISSN 0893-4983.
- Singer-Loginova, I., Singer, H.M., 2008. The phase field technique for modeling multiphase materials. Rep. Prog. Phys. 71 (10), 106501.
- Song, H., Shu, C.-W., 2017. Unconditional energy stability analysis of a second order implicitexplicit local discontinuous Galerkin method for the Cahn-Hilliard equation. J. Sci. Comput. 73 (2-3), 1178–1203.
- Song, F., Xu, C., Karniadakis, G.E., 2016. A fractional phase-field model for two-phase flows with tunable sharpness: algorithms and simulations. Comput. Methods Appl. Mech. Eng. 305, 376–404.
- Souganidis, P.E., Yip, N.K., 2004. Uniqueness of motion by mean curvature perturbed by stochastic noise. Ann. Inst. Henri Poincar (C) Non-Linaire Analyze 21 (1), 1-23. ISSN 0294-1449. https://doi.org/10.1016/S0294-1449(03)00029-5.
- Steinbach, I., 2009. Topical review: phase-field models in material science. Model. Simul. Mater. Sci. Eng. 17, 073001.
- Stogner, R.H., Carey, G.F., Murray, B.T., 2008. Approximation of Cahn-Hilliard diffuse interface models using parallel adaptive mesh refinement and coarsening with  $C^1$  elements. Int. J. Numer. Methods Eng. 76 (5), 636-661.
- Sun, Z.Z., 1995. A second-order accurate linearized difference scheme for the two-dimensional Cahn-Hilliard equation. Math. Comput. 64 (212), 1463-1471.
- Sun, H., Wen, J., Zhao, Y., Li, B., McCammon, J.A., 2015. A self-consistent phase-field approach to implicit solvation of charged molecules with Poisson-Boltzmann electrostatics. J. Chem. Phys. 143 (24), 12B602\_1.
- Takashi, S., Takayuki, A., Tomohiro, T., Akinori, Y., Akira, N., Toshio, E., Naoya, M., Satoshi, M., Malcolm, C., 2011. Peta-scale phase-field simulation for dendritic solidification on the TSUBAME 2.0 supercomputer. In: Proceedings of 2011 International Conference for High Performance Computing, Networking, Storage and Analysis (SC '11).
- Tan, Z., Tang, T., Zhang, Z., 2006. A simple moving mesh method for one-and two-dimensional phase-field equations. J. Comput. Appl. Math. 190 (1-2), 252-269.
- Tarasov, V.E., Zaslavsky, G.M., 2005. Fractional Ginzburg-Landau equation for fractal media. Phys. A Stat. Mech. Appl. 354, 249-261.

- Tinkham, M., 2004. Introduction to Superconductivity. Courier Corporation.
- Valdinoci, E., 2013. A fractional framework for perimeters and phase transitions. Milan J. Math. 81 (1), 1–23.
- Van der Waals, J.D., 1894. Thermodynamic theory of capillarity assuming steady density change. J. Phys. Chem. 13 (1), 657–725.
- Van Gennip, Y., Guillen, N., Osting, B., Bertozzi, A.L., 2014. Mean curvature, threshold dynamics, and phase field theory on finite graphs. Milan J. Math. 82 (1), 3-65.
- Vilanova, G., Colominas, I., Gomez, H., 2017. A mathematical model of tumour angiogenesis: growth, regression and regrowth. J. R. Soc. Interface 14 (126), 20160918.
- Wang, X., Du, Q., 2008. Modelling and simulations of multi-component lipid membranes and open membranes via diffuse interface approaches. J. Math. Biol. 56 (3), 347-371.
- Wang, Y., Li, J., 2010. Phase field modeling of defects and deformation. Acta Mater. 58 (4), 1212-1235.
- Wang, X., Zhang, Z., 2013. Well-posedness of the Hele-Shaw-Cahn-Hilliard system. Ann. Inst. Henri Poincar (C) Non-Linaire Analyze 30 (3), 367-384. ISSN 0294-1449. https://doi.org/ 10.1016/j.anihpc.2012.06.003.
- Wang, X., Ju, L., Du, Q., 2016. Efficient and stable exponential time differencing Runge-Kutta methods for phase field elastic bending energy models. J. Comput. Phys. 316, 21–38.
- Wang, J., Zhai, Q., Zhang, R., Zhang, S., 2019. A weak Galerkin finite element scheme for the Cahn-Hilliard equation. Math. Comp. 88 (315), 211-235. ISSN 0025-5718. https://doi.org/ 10.1090/mcom/3369.
- Wanner, T., 2017. Computer-assisted equilibrium validation for the diblock copolymer model. Discrete Contin. Dyn. Syst. A 37 (2), 1075-1107.
- Weber, H., 2010a. On the short time asymptotic of the stochastic Allen-Cahn equation. Ann. Inst. Henri Poincaré Probab. Stat. 46 (4), 965-975. ISSN 0246-0203. https://doi.org/10.1214/09-AIHP333.
- Weber, H., 2010b. Sharp interface limit for invariant measures of a stochastic Allen-Cahn equation. Comm. Pure Appl. Math. 63 (8), 1071-1109. ISSN 0010-3640. https://doi.org/ 10.1002/cpa.20323.
- Weinan, E., Liu, D., 2002. Gibbsian dynamics and invariant measures for stochastic dissipative PDEs. J. Statist. Phys. 108 (5-6), 1125-1156. ISSN 0022-4715. https://doi.org/10.1023/ A:1019747716056.
- Wheeler, A.A., Murray, B.T., Schaefer, R.J., 1993. Computation of dendrites using a phase field model. Physica D 66 (1-2), 243-262.
- Wheeler, M.F., Wick, T., Wollner, W., 2014. An augmented-Lagrangian method for the phasefield approach for pressurized fractures. Comput. Methods Appl. Mech. Eng. 271, 69-85.
- Wise, S., Kim, J., Lowengrub, J., 2007. Solving the regularized, strongly anisotropic Cahn-Hilliard equation by an adaptive nonlinear multigrid method. J. Comput. Phys. 226 (1), 414-446.
- Wise, S.M., Wang, C., Lowengrub, J.S., 2009. An energy-stable and convergent finitedifference scheme for the phase field crystal equation. SIAM J. Numer. Anal. 47 (3), 2269-2288.
- Wodo, O., Ganapathysubramanian, B., 2011. Computationally efficient solution to the Cahn-Hilliard equation: Adaptive implicit time schemes, mesh sensitivity analysis and the 3D isoperimetric problem. J. Comput. Phys. 230 (15), 6037-6060.
- Wörner, M., 2012. Numerical modeling of multiphase flows in microfluidics and micro process engineering: a review of methods and applications. Microfluid. Nanofluid. 12 (6), 841-886.
- Wu, S., Li, Y., 2018. Analysis of the Morley element for the Cahn-Hilliard equation and the Hele-Shaw flow. arXiv:1808.08581 [math.NA].

- Xia, Y., Xu, Y., Shu, C.-W., 2007. Local discontinuous Galerkin methods for the Cahn-Hilliard type equations. J. Comput. Phys. 227 (1), 472-491.
- Xia, Y., Xu, Y., Shu, C.-W., 2009. Application of the local discontinuous Galerkin method for the Allen-Cahn/Cahn-Hilliard system. Commun. Comput. Phys. 51, 821-835.
- Xu, C., Tang, T., 2006. Stability analysis of large time-stepping methods for epitaxial growth models. SIAM J. Numer. Anal. 44 (4), 1759-1779. ISSN 0036-1429. https://doi.org/ 10.1137/050628143.
- Xu, J., Li, Y., Wu, S., Bousquet, A., 2019. On the stability and accuracy of partially and fully implicit schemes for phase field modeling. Comput. Methods Appl. Mech. Eng. 345, 826-853.
- Yan, Y., Chen, W., Wang, C., Wise, S.M., 2018. A second-order energy stable BDF numerical scheme for the Cahn-Hilliard equation. Commun. Comput. Phys. 23 (2), 572-602. ISSN 1815-2406.
- Yang, X., 2016. Linear, first and second-order, unconditionally energy stable numerical schemes for the phase field model of homopolymer blends. J. Comput. Phys. 327, 294-316. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2016.09.029.
- Yang, X., 2019. Efficient linear, stabilized, second-order time marching schemes for an anisotropic phase field dendritic crystal growth model. Comput. Methods Appl. Mech. Eng. 347, 316-339. ISSN 0045-7825. https://doi.org/10.1016/j.cma.2018.12.012.
- Yang, X., Ju, L., 2017a. Efficient linear schemes with unconditional energy stability for the phase field elastic bending energy model. Comput. Methods Appl. Mech. Eng. 315, 691–712.
- Yang, X., Ju, L., 2017b. Linear and unconditionally energy stable schemes for the binary fluidsurfactant phase field model. Comput. Methods Appl. Mech. Eng. 318, 1005-1029.
- Yang, X., Zhao, J., Wang, Q., Shen, J., 2017. Numerical approximations for a three-component Cahn-Hilliard phase-field model based on the invariant energy quadratization method. Math. Models Methods Appl. Sci. 27 (11), 1993-2030.
- Yang, J., Du, Q., Zhang, W., 2018. Uniform  $L_p$ -bound of the Allen-Cahn equation and its numerical discretization. Int. J. Numer. Anal. Model. 15 (1-2), 213-227. ISSN 1705-5105.
- Yip, N.K., 1998. Stochastic motion by mean curvature. Arch. Rational Mech. Anal. 144 (4), 313-355. ISSN 0003-9527. https://doi.org/10.1007/s002050050120.
- Yip, N.K., 2002. Stochastic curvature driven flows, In: Stochastic Partial Differential Equations and Applications (Trento, 2002). Lecture Notes in Pure and Appl. Math., vol. 227. Dekker, New York, pp. 443-460.
- Yu, P., Chen, L.Q., Du, Q., 2008. Applications of moving mesh methods to the fourier spectral approximations of phase-field equations. In: Recent Advances in Computational Sciences: Selected Papers from the International Workshop on Computational Sciences and Its Education, World Scientific, pp. 80-99.
- Yue, P., Feng, J.J., Liu, C., Shen, J., 2004. A diffuse-interface method for simulating two-phase flows of complex fluids. J. Fluid Mech. 515, 293-317.
- Zhai, S., Weng, Z., Feng, X., 2016. Fast explicit operator splitting method and time-step adaptivity for fractional non-local Allen-Cahn model. Appl. Math. Model. 40 (2), 1315-1324.
- Zhang, J., Du, Q., 2009. Numerical studies of discrete approximations to the Allen-Cahn equation in the sharp interface limit. SIAM J. Sci. Comput. 31 (4), 3042–3063.
- Zhang, J., Du, O., 2012. Shrinking dimer dynamics and its applications to saddle point search. SIAM J. Numer. Anal. 50 (4), 1899-1921.
- Zhang, Z., Qiao, Z., 2012. An adaptive time-stepping strategy for the Cahn-Hilliard equation. Commun. Comput. Phys. 11 (4), 1261-1278.
- Zhang, Z., Tang, H., 2007. An adaptive phase field method for the mixture of two incompressible fluids. Comput. Fluids 36 (8), 1307-1318.

- Zhang, S., Wang, M., 2010. A nonconforming finite element method for the Cahn-Hilliard equation. J. Comput. Phys. 229 (19), 7361-7372.
- Zhang, Q., Wang, X.-P., 2016. Phase field modeling and simulation of three-phase flow on solid surfaces. J. Comput. Phys. 319, 79-107. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2016.05.016.
- Zhang, L., Chen, L.-Q., Du, Q., 2007. Morphology of critical nuclei in solid-state phase transformations. Phys. Rev. Lett. 98 (26), 265703.
- Zhang, L., Chen, L.-Q., Du, Q., 2008. Diffuse-interface description of strain-dominated morphology of critical nuclei in phase transformations. Acta Mater. 56 (14), 3568-3576.
- Zhang, J., Das, S., Du, Q., 2009. A phase field model for vesicle-substrate adhesion. J. Comput. Phys. 228 (20), 7837-7849.
- Zhang, F., Xu, Y., Guo, R., 2015. Interior penalty discontinuous Galerkin based isogeometric analysis for Allen-Cahn equations on surfaces. Commun. Comput. Phys. 18 (5), 1380-1416. ISSN 1815-2406. https://doi.org/10.4208/cicp.010914.180315a.
- Zhang, J., Zhou, C., Wang, Y., Ju, L., Du, Q., Chi, X., Xu, D., Chen, D., Liu, Y., Liu, Z., 2016a. Extreme-scale phase field simulations of coarsening dynamics on the Sunway TaihuLight supercomputer. In: Proceedings of the International Conference for High Performance Computing, Networking, Storage and Analysis, p. 4.
- Zhang, L., Ren, W., Samanta, A., Du, O., 2016b. Recent developments in computational modelling of nucleation in phase transformations. NPJ Comput. Mater. 2, 16003.
- Zhang, F., Huang, W., Li, X., Zhang, S., 2018. A study on phase-field models for brittle fracture. arXiv preprint arXiv:1805.07357.
- Zhao, J., Yang, X., Shen, J., Wang, Q., 2016b. A decoupled energy stable scheme for a hydrodynamic phase-field model of mixtures of nematic liquid crystals and viscous fluids. J. Comput. Phys. 305, 539–556. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2015.09.044.
- Zhao, Y., Xu, B.-X., Stein, P., Gross, D., 2016c. Phase-field study of electrochemical reactions at exterior and interior interfaces in Li-ion battery electrode particles. Comput. Methods Appl. Mech. Eng. 312, 428–446. ISSN 0045-7825. https://doi.org/10.1016/j.cma.2016.04.033.
- Zhao, Y., Ma, Y., Sun, H., Li, B., Du, Q., 2018. A new phase-field approach to variational implicit solvation of charged molecules with the coulomb-field approximation. Comm. Math. Sci. 16, 1203-1223.
- Zheng, X., Yang, C., Cai, X.-C., Keyes, D., 2015. A parallel domain decomposition-based implicit method for the Cahn-Hilliard-Cook phase-field equation in 3D. J. Comput. Phys. 285, 55-70.
- Zhou, C., Yue, P., Feng, J.J., Ollivier-Gooch, C.F., Hu, H.H., 2010. 3D phase-field simulations of interfacial dynamics in Newtonian and viscoelastic fluids. J. Comput. Phys. 229 (2), 498-511.
- Zhu, X.-P., 2002. Lectures on Mean Curvature Flows. AMS/IP Studies in Advanced Mathematics, vol. 32. American Mathematical Society, Providence, RI; International Press, Somerville, MA. ISBN: 0-8218-3311-1, p. x+150.
- Zhu, L., Wang, Y., Ju, L., Wang, D., 2010. A variational phase field method for curve smoothing. J. Comput. Phys. 229 (6), 2390-2400.
- Ziebert, F., Aranson, I.S., 2016. Computational approaches to substrate-based cell motility. NPJ Comput. Mater. 2, 16019.

## **Further readings**

- Babuška, I., Rheinboldt, W.C., 1978. Error estimates for adaptive finite element computations. SIAM J. Numer. Anal. 15 (4), 736-754.
- Babuška, I., Feistauer, M., Šolín, P., 2001. On one approach to a posteriori error estimates for evolution problems solved by the method of lines. Numer. Math. 89 (2), 225-256. ISSN 0029-599X.

- Barrett, J.W., Garcke, H., Nürnberg, R., 2008. On sharp interface limits of Allen-Cahn/Cahn-Hilliard variational inequalities. Discrete Contin. Dyn. Syst. Ser. S 1 (1), 1-14. https://doi. org/10.3934/dcdss.2008.1.1.
- Bronsard, L., Kohn, R.V., 1990. On the slowness of phase boundary motion in one space dimension. Comm. Pure Appl. Math. 43 (8), 983-997. ISSN 0010-3640. https://doi.org/10.1002/ cpa.3160430804.
- Bronsard, L., Reitich, F., 1993. On three-phase boundary motion and the singular limit of a vector-valued Ginzburg-Landau equation. Arch. Ration. Mech. Anal. 124 (4), 355-379. ISSN 0003-9527. https://doi.org/10.1007/BF00375607.
- Bronsard, L., Wetton, B.T.R., 1995. A numerical method for tracking curve networks moving with curvature motion. J. Comput. Phys. 120 (1), 66-87. ISSN 0021-9991. https://doi.org/ 10.1006/jcph.1995.1149.
- Caffarelli, L.A., Muler, N.E., 1995. An  $L^{\infty}$  bound for solutions of the Cahn-Hilliard equation. Arch. Rational Mech. Anal. 133 (2), 129-144. ISSN 0945-8396.
- Cahn, J.W., Taylor, J.E., 1994. Surface motion by surface diffusion. Acta Met. Mat. 42, 1045-1063.
- Chen, Z., Hoffmann, K.H., 1994. An error estimate for a finite-element scheme for a phase field model. IMA J. Numer. Anal. 14 (2), 243-255.
- Coleman, B.D., Falk, R.S., Moakher, M., 1995. Stability of cylindrical bodies in the theory of surface diffusion. Physica D 89 (1-2), 123-135. ISSN 0167-2789.
- Coleman, B.D., Falk, R.S., Moakher, M., 1996. Space-time finite element methods for surface diffusion with applications to the theory of the stability of cylinders. SIAM J. Sci. Comput. 17 (6), 1434-1448. ISSN 1064-8275.
- Deckelnick, K., Dziuk, G., 2002. A fully discrete numerical scheme for weighted mean curvature flow. Numer. Math. 91 (3), 423-452. ISSN 0029-599X.
- Diegel, A.E., Wang, C., Wang, X., Wise, S.M., 2017b. Convergence analysis and error estimates for a second order accurate finite element method for the Cahn-Hilliard-Navier-Stokes system. Numer. Math. 137 (3), 495-534. ISSN 0029-599X. https://doi.org/10.1007/s00211-017-0887-5.
- Dziuk, G., Elliott, C.M., 2013. Finite element methods for surface PDEs. Acta Numer. 22, 289-396. ISSN 0962-4929. https://doi.org/10.1017/S0962492913000056.
- Elliott, C.M., Garcke, H., 1996. On the Cahn-Hilliard equation with degenerate mobility. SIAM J. Math. Anal. 27 (2), 404-423. ISSN 0036-1410.
- Elliott, C.M., Garcke, H., 1997. Existence results for diffusive surface motion laws. Adv. Math. Sci. Appl. 7 (1), 467-490. ISSN 1343-4373.
- Elliott, C.M., Larsson, S., 1992. Error estimates with smooth and nonsmooth data for a finite element method for the Cahn-Hilliard equation. Math. Comp. 58 (198), 603-630, S33-S36. ISSN 0025-5718.
- Elliott, C.M., Maier-Paape, S., 2001. Losing a graph with surface diffusion. Hokkaido Math. J. 30 (2), 297-305. ISSN 0385-4035.
- Elliott, C.M., Songmu, Z., 1986. On the Cahn-Hilliard equation. Arch. Rational Mech. Anal. 96 (4), 339-357. ISSN 0945-8396.
- Elliott, C.M., Stinner, B., 2010b. A surface phase field model for two-phase biological membranes, SIAM J. Appl. Math. 70 (8), 2904–2928. ISSN 0036-1399. https://doi.org/10.1137/ 090779917.
- Elliott, C.M., French, D.A., Milner, F.A., 1989. A second order splitting method for the Cahn-Hilliard equation. Numer. Math. 54 (5), 575-590. ISSN 0029-599X.
- Eriksson, K., Johnson, C., 1995. Adaptive finite element methods for parabolic problems. IV. Nonlinear problems. SIAM J. Numer. Anal. 32 (6), 1729-1749. ISSN 0036-1429.

- Feng, X., Prohl, A., 2003a. Analysis of total variation flow and its finite element approximations. M2AN Math. Model. Numer. Anal. 37 (3), 533-556. ISSN 0764-583X. https://doi.org/ 10.1051/m2an:2003041.
- Feng, X., Prohl, A., 2004b. Analysis of gradient flow of a regularized Mumford-Shah functional for image segmentation and image inpainting, M2AN Math, Model, Numer, Anal. 38 (2), 291–320. ISSN 0764-583X. https://doi.org/10.1051/m2an:2004014.
- Feng, X., Schulze, T.P., 2002. Recent Advances in Numerical Methods for Partial Differential Equations and Applications, Contemporary Mathematics. vol. 306American Mathematical Society, Providence, RI.
- Feng, X., von Oehsen, M., Prohl, A., 2005b. Rate of convergence of regularization procedures and finite element approximations for the total variation flow. Numer. Math. 100 (3), 441-456. ISSN 0029-599X. https://doi.org/10.1007/s00211-005-0585-6.
- Feng, X., Neilan, M., Prohl, A., 2007b. Error analysis of finite element approximations of the inverse mean curvature flow arising from the general relativity. Numer. Math. 108 (1), 93-119. ISSN 0029-599X. https://doi.org/10.1007/s00211-007-0111-0.
- Fife, P.C., 2000. Models for phase separation and their mathematics. Electronic J. of Diff. Eqns 2000 (48), 1-26.
- Garcke, H., Kwak, D. J. C., 2006. On asymptotic limits of Cahn-Hilliard systems with elastic misfit. In: Analysis, modeling and simulation of multiscale problemsSpringer, Berlin, pp. 87-111.
- Garcke, H., Stoth, B., Nestler, B., 1999. Anisotropy in multi-phase systems: a phase field approach. Interfaces Free Bound. 1 (2), 175-198. ISSN 1463-9963. https://doi.org/10.4171/ IFB/8.
- Garcke, H., Lam, K.F., Stinner, B., 2014. Diffuse interface modelling of soluble surfactants in two-phase flow. Commun. Math. Sci. 12 (8), 1475-1522. ISSN 1539-6746. https://doi.org/ 10.4310/CMS.2014.v12.n8.a6.
- Giga, Y., Ito, K., 1998. On pinching of curves moved by surface diffusion. Commun. Appl. Anal. 2 (3), 393-405. ISSN 1083-2564.
- Guillén-González, F., Tierra, G., 2013. On linear schemes for a Cahn-Hilliard diffuse interface model. J. Comput. Phys. 234, 140-171.
- Guo, Z., Lin, P., 2015. A thermodynamically consistent phase-field model for two-phase flows with thermocapillary effects. J. Fluid Mech. 766, 226-271. ISSN 0022-1120. https://doi.org/ 10.1017/jfm.2014.696.
- Guo, Z., Lin, P., Lowengrub, J.S., 2014. A numerical method for the quasi-incompressible Cahn-Hilliard-Navier-Stokes equations for variable density flows with a discrete energy law. J. Comput. Phys. 276, 486-507. ISSN 0021-9991. https://doi.org/10.1016/ j.jcp.2014.07.038.
- Guo, Z., Lin, P., Lowengrub, J., Wise, S.M., 2017. Mass conservative and energy stable finite difference methods for the quasi-incompressible Navier-Stokes-Cahn-Hilliard system: primitive variable and projection-type schemes. Comput. Methods Appl. Mech. Eng. 326, 144-174. ISSN 0045-7825. https://doi.org/10.1016/j.cma.2017.08.011.
- Han, D., Wang, X., 2015. A second order in time, uniquely solvable, unconditionally stable numerical scheme for Cahn-Hilliard-Navier-Stokes equation. J. Comput. Phys. 290, 139–156. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2015.02.046.
- Hele-Shaw, H.S., 1898a. The flow of water. Nature 58 (1489), 33-36.
- Ilmanen, T., 2003. Problems in mean curvature flow. Preprint, downloadable at www.math.ethz. ch/ ilmanen/papers/pub.html.

- Jiang, W., Bao, W., Thompson, C.V., Srolovitz, D.J., 2012. Phase field approach for simulating solid-state dewetting problems. Acta Mater. 60 (15), 5578-5592.
- Karma, A., Kessler, D.A., Levine, H., 2001. Phase-field model of mode III dynamic fracture. Phys. Rev. Lett. 87 (4), 045501.
- Kindermann, S., Osher, S., Jones, P.W., 2005. Deblurring and denoising of images by nonlocal functionals. Multiscale Model. Simul. 4 (4), 1091-1115.
- Mayer, U.F., Simonett, G., 2000. Self-intersections for the surface diffusion and the volume-preserving mean curvature flow. Differ. Integral Equ. 13 (7-9), 1189-1199. ISSN 0893-4983.
- Mullins, W.W., 1957. Theory of thermal grooving. J. Appl. Phys. 28, 333-339.
- Mullins, W.W., Sekerka, R.F., 1964. Stability of a planer interface during solidification of a dilute binary alloy. J. Appl. Phys. 35, 444-451.
- Nochetto, R.H., Savaré, G., Verdi, C., 2000. A posteriori error estimates for variable time-step discretizations of nonlinear evolution equations. Comm. Pure Appl. Math. 53 (5), 525-589. ISSN 0010-3640.
- Novick-Cohen, A., 1998. The Cahn-Hilliard equation: mathematical and modeling perspectives. Adv. Math. Sci. Appl. 8 (2), 965-985. ISSN 1343-4373.
- Nürnberg, R., 2009. Numerical simulations of immiscible fluid clusters. Appl. Numer. Math. 59 (7), 1612–1628. ISSN 0168-9274. https://doi.org/10.1016/j.apnum.2008.11.003.
- Rosolen, A., Peco, C., Arroyo, M., 2013. An adaptive meshfree method for phase-field models of biomembranes. Part I: Approximation with maximum-entropy basis functions. J. Comput. Phys. 249, 303-319. ISSN 0021-9991. https://doi.org/10.1016/j.jcp.2013.04.046.
- Schulze, T., Alexiades, V., Feng, X., 2008. Multi-scale modeling and simulation in materials science, preface. J. Sci. Comput. 37 (1), 1-2. ISSN 0885-7474. https://doi.org/10.1007/ s10915-008-9211-y.
- Sethian, J.A., 1990. Numerical algorithms for propagating interfaces: Hamilton-Jacobi equations and conservation laws. J. Differ. Geom. 31 (1), 131-161. ISSN 0022-040X.
- Shen, J., Yang, X., 2009b. An efficient moving mesh spectral method for the phase-field model of two-phase flows. J. Comput. Phys. 228 (8), 2978-2992.
- Shen, J., Yang, X., Wang, Q., 2013. Mass and volume conservation in phase field models for binary fluids. Commun. Comput. Phys. 13 (4), 1045-1065. ISSN 1815-2406. https://doi. org/10.4208/cicp.300711.160212a.
- Shen, J., Tang, T., Yang, J., 2016. On the maximum principle preserving schemes for the generalized Allen-Cahn equation. Commun. Math. Sci. 14 (6), 1517-1534. ISSN 1539-6746. https://doi.org/10.4310/CMS.2016.v14.n6.a3.
- Stoth, B.E.E., 1996. Convergence of the Cahn-Hilliard equation to the Mullins-Sekerka problem in spherical symmetry. J. Differ. Equ. 125 (1), 154-183. ISSN 0022-0396.
- Tang, T., Yang, J., 2016. Implicit-explicit scheme for the Allen-Cahn equation preserves the maximum principle. J. Comput. Math. 34 (5), 471-481. ISSN 0254-9409. https://doi.org/10.4208/ jcm.1603-m2014-0017.
- Taylor, J.E., 2003. Some mathematical challenges in materials science. Bull. Amer. Math. Soc. (N.S.) 40 (1), 69-87 (electronic). ISSN 0273-0979 Mathematical challenges of the 21st century (Los Angeles, CA, 2000).
- Walkington, N.J., 1996. Algorithms for computing motion by mean curvature. SIAM J. Numer. Anal. 33 (6), 2215-2238. ISSN 0036-1429.
- Wang, H., Li, R., Tang, T., 2008. Efficient computation of dendritic growth with r-adaptive finite element methods. J. Comput. Phys. 227 (12), 5984-6000. ISSN 0021-9991. https://doi.org/ 10.1016/j.jcp.2008.02.016.

- Xie, Z., Song, L., Feng, X., 2008. A moving boundary problem derived from heat and water transfer processes in frozen and thawed soils and its numerical simulation. Sci. China Ser. A 51 (8), 1510–1521. ISSN 1006-9283. https://doi.org/10.1007/s11425-008-0096-x.
- Zhao, J., Wang, Q., Yang, X., 2016a. Numerical approximations to a new phase field model for two phase flows of complex fluids. Comput. Methods Appl. Mech. Eng. 310, 77–97. ISSN 0045-7825. https://doi.org/10.1016/j.cma.2016.06.008.